



Strategic Environmental Assessment and Sustainability Appraisal

Proposed Submission Site Allocations and Policies (Incorporating IP-One Area Action Plan) Development Plan Document

SA Report Addendum – Pre-Submission Main Modifications

Arcadis Consulting (UK) Limited 2212959 Manning House 22 Carlisle Place London SW1P 1JA United Kingdom Tel: +44 (0)20 3014 9000 Fax: +44 (0)20 7828 8428 www.arcadis.com



Ipswich Borough Council

Strategic Environmental Assessment and Sustainability Appraisal

Proposed Submission Site Allocations and Policies (Incorporating IP-One Area Action Plan) Development Plan Document

SA Report Addendum – Pre-Submission Main Modifications

Author	Mwale Mutale	M. Mutale
Checker	David Hourd	
Approver	David Hourd	- Han
Report No	004-UA006314-UE31-01	-F

Date 06 October 2015

This report has been prepared for Ipswich Borough Council in accordance with the terms and conditions of appointment for the Site Allocations and Policies (Incorporating IP-One Area Action Plan) Development Plan Document dated 28 August 2013. Arcadis Consulting (UK) Limited (2212959) cannot accept any responsibility for any use of or reliance on the contents of this report by any third party.

Strategic Environmental Assessment and Sustainability Appraisal – Proposed submission Site Allocations and Policies Plan

Arcadis Consulting (UK) Limited-2212959

Strategic Environmental Assessment and Sustainability Appraisal – Proposed submission Site Allocations and Policies Plan Arcadis Consulting (UK) Limited-2212959



CONTENTS

1	INTRODUCTION	
2	APPROACH	
3	PRE-SUBMISSION MODIFICATIONS	5
	3.1 Pre-submission Main Modifications for Public Consulta	tion5
	3.2 Pre-submission Additional Modifications	
4	SA OBJECTIVE COMPATIBILITY	10
5	BASELINE AND REVIEW OF RELEVANT PLANS,	
	PROGRAMMES AND ENVIRONMENTAL PROTECTIO	N
	OBJECTIVES	10
6	HABITATS REGULATIONS ASSESSMENT	10
7	CUMULATIVE EFFECTS	

Appendices

Appendix A

Sustainability Appraisal Matrices - Site Allocations

Appendix B

Pre-Submission Additional Modifications Table

Appendix C

Consultation Comments on the Proposed Submission SA Report

1 INTRODUCTION

This Sustainability Appraisal (SA) addendum provides an update to the published Strategic Environmental Assessment and Sustainability Appraisal – Proposed Submission Site Allocations and Policies (Incorporating IP-One Area Action Plan) DPD SA Report (December 2014) (known hereinafter as the Proposed Submission SA Report). The Proposed Submission Site Allocations and Policies (Incorporating IP-One Area Action Plan) DPD was consulted upon between 12th December 2014 to 5th March 2015. Following this consultation, a series of proposed modifications to the DPD have been identified. These have been categorised as either Pre-Submission Main Modifications or Pre-Submission Additional Modifications. Consultation is taking place on the Pre-Submission Main Modifications between 9th October 2015 and 23rd November 2015. This addendum to the Proposed Submission SA Report considers whether the proposed modifications alter the findings of the SA Report.

This report also seeks to provide an update on the consideration of consultation comments on the Proposed Submission SA Report received during the December 2014 - March 2015 consultation. The consultation comments are provided within Appendix C. Once this current consultation is complete, the Proposed Submission DPD and the Pre-Submission Main Modifications will be submitted to the Planning Inspectorate in December 2015 for independent examination. The Pre-Submission Additional Modifications will also be submitted. For submission, the Proposed Submission SA report and this Addendum will be combined and updated to form one SA report, for submission in accordance with Regulation 22(1)(a) of the Town and Country Planning (Local Planning) (England) Regulations 2012.

Advice has been received from the Planning Inspectorate that it is appropriate to consult on main modifications prior to the submission of development plan documents to the Government for examination. The Pre-Submission Main Modifications are substantive changes which alter the meaning of a policy or strategy (e.g. rewording policies to change their meaning, adding new sites or deleting existing ones). The Pre-Submission Additional Modifications are minor textual and grammar corrections; re-phrasing or limited new text to add clarity; or updates to figures and references, which are necessary due to alterations which have been made elsewhere or for which new information has come to light. These do not need to be consulted upon. This Sustainability Appraisal addendum considers whether any of the Main Modifications would affect the results of the Sustainability Appraisal and, for completeness, checks whether the Additional Modifications would result in any changes to the assessment.

This addendum should be read alongside the Proposed Submission SA Report and its Non-Technical Summary. Representations should be submitted in relation to the Sustainability Appraisal of the Pre-Submission Main Modifications only however in doing so reference can be made to the Proposed Submission SA Report insofar as its contents relate to this addendum.

Note that the Proposed Submission SA Report was produced by Hyder Consulting (UK) Limited who now operates as Arcadis Consulting (UK) Limited following a buy-out in 2014.

2 APPROACH

The focus of this addendum is on the assessment of the Pre-Submission Main Modifications. These are being consulted upon prior to submission of the Site Allocations and Policies (Incorporating IP-One Area Action Plan) development plan document.

Section 3 of this report presents a table of the Pre-Submission Main Modifications and identifies if these would result in a change to the existing SA Findings and if so, how and whether any mitigation is required. These tables should be read in conjunction with the relevant revised appraisal matrices which are presented in Appendices A and B. The scope and methodology,

including sustainability objectives, are contained within the December 2014 Proposed Submission SA Report – please refer to this when reading this addendum. Whilst not the main focus of this SA, the table in Appendix B considers whether any of the Council's proposed Pre-Submission Additional Modifications would have any implications for the sustainability appraisal conclusions, for example through highlighting factual amendments. Please note the Pre-Submission Additional Modifications are not subject to consultation.

Section 4 of this report provides an update to the internal compatibility of the SA Objectives presented in the SA Report.

Section 5 of this report provides an update in relation to the SA baseline data and review of plans, programmes and relevant environmental protection objectives.

Section 6 of this report provides further information on links between the SA and the ongoing Habitats Regulations Assessment (HRA).

Section 7 of this report provides an update to the cumulative effects assessment taking account of development sites outside of Ipswich but adjacent or close to its boundary.

3 PRE-SUBMISSION MODIFICATIONS

3.1 Pre-submission Main Modifications for Public Consultation

The Pre-Submission Main Modifications presented in Table 3-1 below are the more substantive changes which alter the meaning of a policy or strategy (e.g. rewording policies to change their meaning, adding new sites or deleting existing ones). New text added is shown underlined, deleted text is shown struck through. The changes to the SA assessment matrices have been evaluated and explained within the 'significance to the SA' column. Revised matrices showing the assessments of the proposed modifications to the SP policies and the site allocations are contained in Appendices A and B.

These changes do no result in any changes to the significant effects identified in section 5 of the Proposed Submission SA Report and no further mitigation measures are considered necessary. The changes also do not result in any changes to the proposed monitoring framework which was set out in section 6.4 of the Proposed Submission SA Report.

Policy / paragraph	Main Modification	Reason	Objector(s)	Significance to the SA
SP2 LAND ALLOCATED FOR HOUSING and policies map	 Additional site allocations of sites previously listed under policy SP3 – to be added to Table 1: IP011a Lower Orwell Street (former Gym & Trim and formerly listed as Smart Street/Foundation Street) 0.15ha. 14 dwellings (90dph on 100% of site) (permission for student accommodation granted 25/10/11 - lapsed Oct 2014) IP047 Land at Commercial Rd 2.86ha. 103 dwellings (90dph on 40%) as part of a mixed use scheme of 40% housing, 20% public open space and enhanced river path, 40% office, leisure, hotel (development including supermarket approved 7/2/11 – lapsed Feb 2015) IP090 Europa Way 1.43ha, 18 dwellings (indicative masterplan), as part of a mixed use district centre (approved 11/07/11 – lapsed July 2014) IP132 Former St Peters Warehouse Site, 4 Bridge Street, 0.18ha, 73 dwellings as part of a mixed use scheme (with B1a office use, leisure or small scale retail) (approved 14/4/11 – lapsed April 2014) IP135 112-116 Bramford Road 0.17ha, 14 dwellings (pending application withdrawn October 2014) 	The plan baseline has been updated to 1st April 2015. The planning permissions for some sites have now lapsed or applications been withdrawn and therefore it is appropriate to allocate the sites for development. Reference to the district centre added to IP090 to ensure the allocation through SP2 aligns with its allocation as a District Centre through policy DM21. Capacity of IP132 Former St Peter's Warehouse increased from 48 to 73 to reflect the successful bid for the site.	Editorial update. ALDI	The sites have been removed from Policy SP3 and placed into Policy SP2 within the SA. Similarly, one site has been moved from SP2 to SP3 (IP116 – see below). The number of dwellings per site allocation has not changed therefore the significant findings and the overall assessment of the sites within the SA, in terms of the housing elements of the allocations, has not changed. The mixed uses of IP047, IP090 and IP132 sites have been assessed. A positive score has been given against for IP132 against SA Objective ER2 as the provision of B1a office use, leisure or small scale would contribute towards providing employment opportunities. A positive score has been given for the assessment of IP047 against SA Objective ER2, ER4 and ER7 as the provision of new office, leisure and hotel development will contribute towards providing employment opportunities and supporting economic

Table 3-1 Pre-submission Modifications for Public Consultation

Strategic Environmental Assessment and Sustainability Appraisal — Site Allocations and Policies DPD Arcadis Consulting (UK) Limited-2212959

Policy /	Main Modification	Reason	Objector(s)	Significance to the SA
paragraph	Site sheets have been prepared for these allocations for inclusion in Appendix 3A. Delete site which has a resolution to grant permission subject to a Section 106 Agreement and which needs to be safeguarded through policy SP3: IP116 St Clement's Hospital Grounds, 227 dwellings. Add 222 dwellings to total capacity figure of SP2 Table 1 but delete 227 dwellings: Total 1,934 1,929			growth. A positive score was recorded for the assessment of IP090 against SA Objective ER2 as the provision of a new District Centre would create employment opportunities.
SP3 LAND WITH PLANNING PERMISSION and policies map	 Delete table entries for sites identified through SP3, for which planning permission has now lapsed or which have been withdrawn. The sites are to be allocated through policy SP2 (see above) and are listed below. IP11a Smart Street Foundation Street (former Gym & Trim) (planning permission for student accommodation granted 25/10/11 - lapsed Oct 2014) IP047 Land at Commercial Rd (approved 7/2/11 – lapsed Feb 2015) IP090 Europa Way (approved 11/07/11 – lapsed July 2014) IP132 Former St Peters Warehouse Site 4 Bridge Street (approved 14/4/11 – lapsed April 2014) IP135 112 116 Bramford Road 	The sites have been moved into policy SP2 as allocations.	Editorial update	 Further to the action above, the Policy removes sites for which planning permission has lapsed. IP178 has been removed from the assessment. It is recognised that these sites may potentially come forward later than originally planned. However, at the scale of the assessment, this does not changed the overall SA scores.

Strategic Environmental Assessment and Sustainability Appraisal — Site Allocations and Policies DPD Arcadis Consulting (UK) Limited-2212959

Policy /	Main Modification	Reason	Objector(s)	Significance to the SA
paragraph				
	(pending application withdrawn			
	October 2014)			
	Delete table entry now being used as			
	a music venue under planning			
	permission 10/00277:			
	• IP178 Island House, Duke Street			
	(student accommodation approved			
	Feb 2012 – lapsed Feb 2015)			
	Add sites granted planning			
	permission or with a resolution to			
	grant subject to a Section 106			
	Agreement, which need to be			
	safeguarded through policy SP3:			
	IP116 St Clement's Hospital			
	Grounds, 227 dwellings and public			
	open space provision.			
	IP272 72 Foundation Street, 10			
	dwellings (reference 14/00939/FUL			
	granted 19/02/15)			
	Delete 232 dwellings from capacity			
	figure of SP3, Table 2, but add 237.			
	Total <u>1,8921,897</u>			
SP5 Land	Add to second part of Table 3 (sites	To update the plan baseline to 1st April	Editorial Update	The modification updates
allocated for	allocated for employment uses with a	2015.		IP047 and IP132 with the
employment use	mix of other uses):			inclusion of employment and
	IP047 Land at Commercial Road,			a mix of other uses at the
	2.86ha (40%) 1.14ha, as part of a			sites. These updates have
	mixed use scheme with 40% housing			been reflected within the
	and 20% public open Space and			assessment of the sites and
	enhanced river path. The B1a			a positive score has been
	element could also include hotel and			recorded for both sites
	leisure uses.			against SA Objectives ER2
				and for IP047 against ER4 a
	IP132 Former St Peters Warehouse			the provision of new, office,
	Site, 4 Bridge Street, 0.18ha /			leisure and hotel
	0.05ha, B1a office, leisure, small			development will contribute
	scale retail as part of a mixed use			towards providing

Strategic Environmental Assessment and Sustainability Appraisal — Site Allocations and Policies DPD Arcadis Consulting (UK) Limited-2212959

Policy / paragraph	Main Modification	Reason	Objector(s)	Significance to the SA
	<u>scheme of 73 dwellings</u> (approved 14/4/11 – lapsed April 2014)			employment opportunities and supporting economic growth. There were no changes to SA scores for IP132.
SP6 LAND ALLOCATED AND PROTECTED AS OPEN SPACE	Delete: Site IP263 West of Bridge Street, north of the River Orwell is allocated for public open space. Add to table: <u>IP047 Commercial Road (UC048)</u> <u>20% Open Space and enhanced</u> <u>river path; 50% housing. 30% hotel,</u> <u>leisure.</u> Paragraph 4.37: for this use at site IP263 <u>IP047</u> and within	The planning permission on the site IP047 has now lapsed and therefore it is appropriate to allocate the whole site, including the IP236 element, for residential-led mixed use development including public open space.	Editorial Update	The modification combines the open space element of IP263 into IP047 for which planning permission has lapsed. The SA of IP263 has been deleted. The SA of IP047 has been slightly amended to reflect the inclusion of 0.17ha of open space. Whilst this may have some minor benefits regarding health, landscape and climate change adaptation, the scale of open space is relatively small and has not changed the SA scores for IP047.

3.2 Pre-submission Additional Modifications

Whilst the focus of this SA Addendum is on the assessment of the Pre-Submission Main Modifications, the table in Appendix B checks whether the Council's proposed Additional Modifications would change the SA. The Additional Modifications presented in Appendix B are textual and grammar corrections; re-phrasing or limited new text to add clarity; or updates to figures and references, which are necessary due to alterations which have been made elsewhere or for which new information has come to light. The consideration of the Additional Modifications has resulted in some factual updates to the SA but has not resulted in any change to the significant effects identified in Section 5 of the Proposed Submission SA Report and no further mitigation measures are considered necessary. No changes to the monitoring framework, contained in Section 6 of the Proposed Submission SA Report are considered necessary. Any resultant changes to the matrices are shown in Appendix A.

4 SA OBJECTIVE COMPATIBILITY

The Proposed Submission SA Report includes an internal compatibility assessment of the 21 SA Objectives in order to identify any potential areas of internal incompatibility (see section 2.3.24 – 2.3.28 of the Proposed Submission SA Report).

Comments submitted in relation to the Proposed Submission SA Report for the Core Strategy and Policies DPD Review suggested that this assessment did not sufficiently identify negative effects. Generally the SA Objectives were either compatible or no clear impacts between the objectives could be established. However, some uncertainties were identified. It should be emphasised that this is an assessment of the SA objectives against each other, without considering any potential effects of the plan. Therefore it is not correct to amend the assessment in light of the subsequent assessment of specific policies as the same objectives need to be applied equally to all aspects of the DPD and the proposals within it. For example, whilst meeting the housing requirements for the whole community promoted in SA Objective ER3 may result in conflicts with other objectives relating to, for example traffic growth for some developments it is not inevitable that this will be the case for all and therefore the assessment of uncertain against this objective is still applicable. Similarly, some of these uncertainties could be addressed through mitigation measures proposed in other policies within the DPD.

BASELINE AND REVIEW OF RELEVANT PLANS, PROGRAMMES AND ENVIRONMENTAL PROTECTION OBJECTIVES

5

6

Since the SA Report was produced at the end of 2014, some aspects of the baseline information used in the SA have changed. Similarly, there are a small number of additional or revised relevant plans, programmes or environmental protection objectives that could also be considered. It is considered that the evolution of this baseline is limited, minor and would not result in changes to the SA objectives or findings. However, where changes have occurred these will be updated in the revised SA Report for December 2015.

HABITATS REGULATIONS ASSESSMENT

Comments from Natural England on the Proposed Submission SA report stated that further cross-reference is needed between the Habitats Regulations Assessment and the Sustainability Appraisal. The Proposed Submission SA Report provides cross-references to the HRA of the

Site Allocations DPD which was undertaken in parallel, '*Habitats Regulations Assessment for Ipswich Borough Council Proposed Submission Site Allocations and Policies (incorporating IP-One Area Action Plan) DPD December 2014*'. This assessment identified that the DPD would not be sound if an Appropriate Assessment could not show that there was no adverse effect upon the integrity of nature conservation sites of European importance as recognised by their designation as Special Areas of Conservation, Special Protection Areas and/or Ramsar sites. The Assessment showed that Policy SP8 (allocation IP149 for a new Country Park at Pond Hall Farm), had been revised from the previous draft to give greater clarity that further studies were needed on SPA impacts before this allocation could be implemented, giving reassurance that the DPD would have no likely significant effect on a European site. The Assessment also concluded that none of the site allocations under SP2 would individually have an adverse effect on the integrity of a European site, although it should be noted that cumulative effects from housing growth across Ipswich and surrounding areas have been considered through an Appropriate Assessment of the Proposed Submission Core Strategy and Policies DPD Review.

The screening conclusion was that the plan was not likely to have a significant effect on European protected sites and that an appropriate assessment of the plan was not necessary. An addendum to the Habitats Regulations Assessment has been produced to consider the effects of the Pre-Submission Main Modifications and Pre-Submission Additional Modifications. This concludes that the conclusions of the Proposed Submission HRA Report remain unchanged. Further reference to the conclusions of the HRA have been added to the assessment matrices in Appendix A in relation to site IP149. As the SP2 conclusions relate to all housing site allocations it has not been practical to include reference within each matrix but the conclusions above apply.

The Proposed Submission Site Allocations and Policies (incorporating IP-One Area Action Plan) DPD was also assessed in combination with the Proposed Submission Core Strategy and Policies DPD Review and the conclusions are reliant on the conclusions and mitigation measures of the Appropriate Assessment for the Core Strategy review.

A greater degree of cross-referencing will be provided in the revised SA Report for submission in December 2015 in order to ensure the consistencies between the findings of the SA and HRA are clear.

7 CUMULATIVE EFFECTS

The SEA Directive requires that the assessment includes identification of cumulative and synergistic effects (where the combined effects are greater than the sum of their component parts). This assessment is presented in Section 4 the Proposed Submission SA Report. In reflection of representations submitted in respect of the Core Strategy Proposed Submission SA Report potential cumulative effects have been considered in relation to sites located on the edge of the town but outside the Borough boundary. These sites are:

- Babergh / Ipswich fringe allocation (approximately 350 homes at south west Ipswich) (Babergh Core Strategy and Policies, February 2014)
- Sproughton Strategic Employment Site allocation (former Sugar Beet Factory site) (Babergh Core Strategy and Policies, February 2014)
- Adastral Park (2,000 homes at Martlesham) (Suffolk Coastal Core Strategy and Development Management Policies, 2013)
- Trinity Park (Suffolk Coastal planning permission)

Table 6-1 summarises the cumulative and synergistic impacts of the plan. This has been updated to provide reference to the above sites where relevant. Section 4 of the Proposed Submission SA Report also stated that the cumulative effects were considered to be the

significant effects of the plan. Whilst the cumulative effects table has been updated to reflect development outside of the Borough, no amendments to the proposed monitoring framework are considered necessary.

Table 6-1 Cumulative and Synergistic Effects

Topic/SA Objective	Cumulative / Synergistic Effect (Positive, Negative, Neutral)	Commentary and Causes
Education provision and educational attainment (CL1 To maintain and improve access to education and skills for both young people and adults)	Positive	Educational attainment in the borough is generally lower than regional and national levels. The policies/site allocations are likely to generate positive effects upon educational attainment through the allocation of land for employment use, primary and secondary schools and where necessary for students accommodation. New and diverse employment provision is likely to increase training opportunities, and encourage local partnerships and initiatives. All the residential site allocations in the IP-One area may benefit from the close location of University Campus Suffolk and Suffolk New College.
Crime and Fear of Crime (CD1 To minimise potential opportunities for crime and anti-social activity)	Positive/Negative	Crime levels are generally higher across the borough when compared to national average figures. The projected housing and population growth may result in a significant increase of new residents. This increase in population may have a negative effect on crime within the Borough in the short term, however it can be mitigated cumulatively through improved quality of life. The level of certainty of prediction is low as many factors could influence the crime levels.
Access to services (ET5 To improve access to key services for all sectors of the population)	Positive/Negative	Cumulatively, access will be improved where sites are located in the town centre near existing facilities and employment hubs. However, if all the sites are taken forward negative effects associated with access could occur on key roads in the town centre or the approach roads due to increased traffic and congestion. Mitigation measures will be required to reduce the negative effects, e.g. improved transport links, pedestrian infrastructure, etc.
Health and Well-Being (HW1 To improve the health of those most in need HW2 To improve the quality of life where people live and encourage community participation)	Positive	Whilst health and well-being is affected by a number of factors, there is the potential to improve well-being by delivery of new housing and employment, areas of open/green space and the provision of new schools. There could be indirect benefits for health and well-being within areas where sites are allocated for new parks, river paths, and playing grounds. Many of the policies seek enhancements for walking and cycling which would potentially improve health.
Housing (ER3 To help meet the housing requirements for the whole community)	Positive	Cumulatively, the site allocations for residential use should ensure that new housing development occurs in the most appropriate locations and meets the needs of a wide range of people. The residential site allocations should be planned in a way that it does not lead to oversupply of flats particularly in the town centre. The provision of 350 homes at the Babergh / Ipswich Fringe, the residential development at Trinity Park to the east of Ipswich and the large 2000 home development at Adastral Park (also outside

Topic/SA Objective Cumulative / Synergistic Effect (Positive, Negative Neutral)		Commentary and Causes			
		the Ipswich boundary) will cumulatively support meeting community housing requirements.			
Sustainable Economic Growth (ER4 To achieve sustainable levels of prosperity and economic growth throughout the plan area)	Positive	Cumulatively, the site allocations for employment/retail/office use will have a positive effect on the sustainable growth of the borough as employment opportunities will increase both in the short-term and the long-term. The sites will contribute to the regeneration / redevelopment of central town areas, with some infrastructure improvements (car parking, the replacement of an electricity substation, etc.) which may increase the business attractiveness of the area. The Sproughton Strategic Employment site allocation (in the Babergh Core Strategy) would cumulatively support economic growth in the area, helping to attract inward investment to Ipswich as a whole.			
Biodiversity – protection of designated sites (ET8 To conserve and enhance biodiversity and geodiversity, including favourable conditions on SSSIs, SPAs and SACs)	Negative/Positive	Cumulatively, the site allocations will have both negative and positive effects on biodiversity. Negative effects are identified where TPOs and some larger greenfield sites could potentially be affected by development. Some greenfield sites outside but on the edge of Ipswich could also contribute to this. In addition, sites allocations near the River Orwell may result in disturbance of species during construction activities. The redevelopment of brownfield sites would provide opportunities for enhancements, particularly where this relates to improving connections. Some positive effects may occur as a result of the delivery of open space but cumulatively the amount of land allocated for open space does not suggest a significant positive effect. The remediation of contaminated land may have positive indirect effects in the long term.			
Landscape/Townscapes (ET10 To conserve and enhance the quality and local distinctiveness of landscapes and townscapes)	Negative/Positive	Cumulatively, there is potential for urban expansion and projected growth to have a negative effect on landscape through the loss of greenfield sites, when considered with the development of the Garden Suburb. Similarly some of the edge of town developments outside of Ipswich could contribute to changes in the urban fringe. Cumulative impacts on townscape are due to significant changes in townscape character and the concentration of a great number of listed buildings in the town centre. However, there are proposals for the improvement of public realm, which would contribute to enhance the townscape through redevelopment of brownfield sites and overall regeneration of the town centre.			
Climate Change Air Quality Energy Efficiency Natural Resources Sustainable Transport Transboundary (ET1 To improve air quality	Positive/Negative	The policies in the Site Allocations DPD have a strong focus upon ensuring new development is accessible by public transport, walking and cycling links and that new housing, services and employment opportunities are appropriately sited. All of these measures should contribute in the long-term to enabling sustainable patterns of living and travel to be developed which could have a positive effect upon reducing carbon emissions from transportation sources and upon improving local air quality. However at the same time, new development and population growth along with the			

Topic/SA Objective	Cumulative / Synergistic Effect (Positive, Negative, Neutral)	Commentary and Causes
ET4 To reduce the effects of traffic upon the environment ET6 To limit and adapt to climate change ER6 To encourage efficient patterns of movement in support of		number of cars in the borough would have adverse effects on sustainable travel, air quality and climate change. The development of sites overall and combined with Ipswich Garden Suburb may lead to increases in traffic and air quality issues; however it is uncertain at this stage whether this could significantly affect the AQMAs. There is a degree of uncertainty about these cumulative effects being realised as this is reliant upon travel choices of individual residents and workers.
ET3 To reduce waste)		Transboundary issues relate to people travelling to and from Ipswich for shopping or work. In addition, housing development on the edge but outside of Ipswich Borough may also contribute to increased movements into Ipswich. Whilst there may be increase in retail and employment uses, these are not considered significant and any traffic impacts are not considered significant.
		New development across the borough has the potential to result in a cumulative increase in the use of natural resources and waste generation. Mitigation measures should be implemented to minimise the potential cumulative negative effects.
Water Resources (ET7 To protect and enhance the quality of water features and resources and reduce the risk of flooding)	Negative /Positive	New development, including that just outside the borough boundary is likely to place pressure on water resources and increase consumption of water resources. However, there is mitigation suggested within the site allocations assessment to ensure sustainable design, appropriate flood risk management and sustainable (urban) drainage systems incorporated into new development. The enhancement of the green infrastructure network across the borough will also provide benefits for infiltration and water management. Mitigation should also be implemented to encourage measures to reduce potable demand, use of rain water harvesting and grey water recycling systems to reduce domestic water use.

Appendix A

Sustainability Appraisal Matrices – Site Allocations

Ipswich North West

Whitton

IP140: Land north of Whitton Lane

IP032:King George V Field, Old Norwich Road, 99 homesIP005:Former Tooks Bakery, Old Norwich Rd, 101 homes and a health centre

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET1 To improve air quality	IP140 +/- IP032 - IP005 -	Short, medium and long term Direct and indirect Reversible Medium	North West Central Ipswich transboundary, Mid Suffolk	IP140 would help contribute to improving air quality in central Ipswich by potentially supporting the future use of the former adjacent park and ride facility. IP140 has potential for positive and negative effects, depending on whether there is public transport, walking and cycling provision to mitigate growth in cars. IP032, IP005 are sites where large scale residential development is proposed, which cumulatively could contribute to higher levels of greenhouse gas emissions through increased private car use. If car journeys were generated towards the town centre, they could adversely affect the AQMA at Norwich Rd/Chevallier Street. However, the sites are on at least one frequent bus route (No. 8 service runs every 10 minutes and evenings and weekends) and there is cycle route provision linking the sites to the town centre and other destinations. Therefore there would be every opportunity for sustainable travel choices to be made by residents. <i>It is recommended that pedestrian and cycling infrastructure is enhanced in this</i> <i>area.</i>

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET2 To conserve soil resources and quality	IP140 - IP032 - IP005 +/ -	Medium and long term Direct and indirect Irreversible Medium	North West	IP140 would result in the loss of greenfield land and its associated soil resource. IP032 would see 80% of land used for housing provision which would result in loss of land currently used as playing fields. There is also the potential for contaminated land to be encountered as at IP005, the Former Tooks Bakery though remediation of this would help to improve the soil resource. Land to the north of the site would be retained for use as open space/playing fields which would offer some minor benefits to the SA Objective. <i>Remediation of land at IP005 would help to improve soil quality.</i> <i>Provision of wildlife and recreational green corridor functions associated with the</i> <i>'green rim' (Core Strategy Policy CS16,</i> <i>Site Allocations and Policies DPD draft policies DM33 and DM31) should be</i> <i>encouraged as this could provide partial</i> <i>benefits to the SA Objective.</i>
ET3 To reduce waste	IP140 - IP032 - IP005 -	Medium and long term Direct and indirect Reversible Medium	Site	IP140 is allocated for employment use and in the long term would lead to an increase in waste during operation. Similarly the provision of homes with an indicative capacity of 99 for IP032 and 101 with IP005 would increase waste in the long term. The provision of adequate waste facilities and where possible recycling facilities would encourage recycling. It is recommended that adequate waste facilities should be provided with development at IP140, IP032 and IP005 and where possible provision of recycling facilities. Where possible, waste reduction initiatives should also be encouraged.
ET4 To reduce the effects of traffic upon the environment	IP140 +/- IP032 +/- IP005 +/-	Medium and long term Direct and indirect	North West transboundary, Mid Suffolk	IP140 would help contribute to achieving the SA objective by potentially supporting the future use of the former adjacent park and ride facility. This would provide the

Objective	Performance	Temporal Scale	Geographical	Commentary/Mitigation
	of Policy	Nature of Impact (Direct/Indirect)	Extent	
		Reversibility		
		Certainty		
		Reversible Medium		opportunity for the local community and those travelling from further afield to travel locally in a more sustainable manner which would contribute towards reducing the effects of traffic upon the environment and easing congestion within central Ipswich though ultimately it would lead to a minor localised increase in vehicles. The provision of business park with IP140 and homes with IP032 and IP005 would lead to an increase of traffic within the local area. However the sites allocated for housing would provide homes in close proximity to
				a large existing employment area. There are also bus services running in the north west area which would provide the opportunity for convenient and sustainable travel and help reduce the effects of traffic upon the environment. A transport assessment and travel plan is required for each of the sites. The traffic impact of access from Bury Road in relation to IP032 will also be considered. This will help to reduce any potential negative effects on traffic.
ET5 To improve access to key services for all sectors of the population	IP140 + IP032 +/- IP005 +	Medium and long term Direct and indirect Reversible Medium	North West Transboundary Central Ipswich	A park and ride facility adjacent to IP140 would contribute to improving access to key services within the north west and would have positive transboundary effects. The provision of homes with IP032 and IP005 close to an existing District Centre (3) and the provision of a new health centre with IP005 will improve access to key services in the long term. IP032 would also see the loss of open space though some of this would be
				retained and the allocation expressly requires replacement provision in a suitable location.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET6 To limit and adapt to climate change	IP140 +/ - IP032 +/ - IP005 +/ -	Medium and long term Direct and indirect Reversible Medium	North West and central Ipswich	A park and ride facility adjacent to IP140 would encourage the use of public transport for trips into central Ipswich which would contribute to decreasing vehicle emissions. The provision of business park at IP140 and residential units at IP032 and IP005 would result in potential localised increase in vehicle emissions. If car journeys were generated towards the town centre, they could adversely affect the AQMA at Norwich Rd/Chevallier Street. However, the sites are on at least one frequent bus route (No. 8 service runs every 10 minutes and evenings and weekends) and there is cycle route provision linking the sites to the town centre and other destinations. Therefore there would be every opportunity for sustainable travel choices to be made by residents. The location of this proposed development is within close proximity of existing and proposed employment land and near a District Centre which may contribute to reducing reliance on private cars. These sites are not located in flood risk areas. <i>It is recommended that housing development at IP032 and IP005 should meet Code for Sustainable Homes standards as required by DM1 of the Core Strategy.</i> <i>It is recommended that the business park at IP140 should meet BREEAM standards and where possible SuDS features should be incorporated into design as required by DM1 and DM4 of the Core Strategy.</i>

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET7 To protect and enhance the quality of water features and resources and reduce the risk of flooding	IP140 - IP032 - IP005 -	Medium and long term Direct and indirect Reversible Medium	Site	The provision of residential units at IP032 and IP005 would lead to an increase in water use. IP140 would replace a greenfield site and this could increase surface run-off, but could be offset by adequate drainage system and gardens. In respect of water supply and waste water treatment, infrastructure and / or treatment upgrades will be required to serve the proposed growth, or diversion of assets may be required. Surface water flooding local to site – will be considered at planning application stage for each site. See Appendix 1 of the Ipswich SFRA. It is recommended that where possible SuDS features should be incorporated into design of these sites as required by DM4 of the Core Strategy.
ET8 To conserve and enhance biodiversity and geodiversity, including favourable conditions on SSSIs, SPAs and SACs	IP140 - IP032 0 IP005 0	Medium and long term Indirect Reversible Low	Site	IP140 contains hedgerows and mature trees which have great wildlife value. This is a valuable habitat for nesting birds and wildlife which would be lost as a result of development. Further surveys will be needed prior to any vegetation clearance to establish the habitat value and presence/absence of species, including birds and reptiles. There would be a minor loss of green space at IP032 though this would not significantly affect biodiversity. IP140, IP032, and IP005 could offer benefits to biodiversity through incorporating soft landscaping and tree planting. Where possible the site allocations should include provision of soft landscaping thus providing some benefits to biodiversity. It is recommended that where possible the wildlife and recreational green corridor functions associated with the 'green rim' Core Strategy Policy CS16, DM33 and DM34 should be supported particularly at IP140. The hedgerows and mature trees have the greatest wildlife value, but further

Performance of Policy	Temporal Scale Nature of Impact	Geographical Extent	Commentary/Mitigation
	(Direct/Indirect)		
	-		
			surveys will be needed prior to any vegetation clearance to establish the full wildlife interest, including for birds and reptiles. Compensation measures should be implemented where appropriate.
IP140 0 IP032 0 IP005 0	N/A	N/A	IP140 is located adjacent to Whitton Conservation Area and IP032 is adjacent to Whitton Conservation Area. Any cumulative impacts on the conservation area with the development of IP005 adjacent to IP032 will need to be taken into account. IP005 has been evaluated and features of prehistoric and Roman date were identified over much of the site in the form of pits and boundary ditches. There is a need for archaeological excavation. There is no objection in principle to development but <i>any permission will require a condition relating to archaeological investigation.</i> The Council has published a development brief for this site and the adjacent former
			Tooks bakery site (reference IP005). Thus there are potential archaeology and conservation area issues but they can all be addressed through the planning process. On the whole it is considered that the effects will be negligible.
IP140 - IP032 + IP005 +	Medium and long term Direct and indirect Reversible Medium	North West	Development could contribute towards enhancing quality and local distinctiveness though IP140 would see the loss of greenfield land. Ipswich Borough Council is in a process of preparing an Urban Character Supplementary Planning Document (SPD) to provide additional guidance to support policies set out in their development plan documents. Phase 2 of the work (adoption expected 2016) will cover this area of the town. Once adopted, this document should be consulted upon at design stage. The Council also currently has in place an adopted planning brief for IP005 and IP032.
	of Policy	of Policy Nature of Impact (Direct/Indirect) Reversibility Certainty IP140 0 IP032 0 IP005 0 IP005 0 IP140 - IP140 - IP140 - IP032 + IP005 + Medium and long term Direct and indirect Reversible	of Policy Nature of Impact (Direct/Indirect) Reversibility Certainty Extent IP140 0 N/A N/A IP032 0 N/A N/A IP005 0 Medium and long term North West IP140 - IP005 + Medium and long term North West

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
				should complement the existing character of the area opportunities should be sought to enhance townscape. Where possible soft landscaping should be incorporated into design.
HW1 To improve the health of those most in need	IP140 + IP032 + IP005 ++	Medium and long term Direct and indirect Reversible Medium	Localised	The provision of a new health centre at IP005 would directly support the SA Objective as it would provide a health facility to serve existing and future residents in the local area. Residential developments would provide high quality housing to local areas which would help to support the SA Objective. Each of the sites is located close to playing fields and cycle routes, which would contribute to improving health by encouraging healthier lifestyles.
HW2 To improve the quality of life where people live and encourage community participation	IP140 + IP032 + IP005 +	Medium and long term Direct and indirect Reversible Medium	Localised	The provision of a new health centre with IP005 would contribute to improving the quality of life of where people live. Residential developments would provide high quality housing and alongside the creation of an employment area (IP140) that would help to support the SA Objective. Each site is located near playing fields and cycle routes which would help to encourage healthier lifestyles and encourage community participation.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ER1 To reduce poverty and social exclusion	IP140 + IP032 + IP005 +	Medium and long term Direct and indirect Reversible Medium	Localised	The provision of a new health centre with IP005 would make a small contribution towards reducing social exclusion. Housing stock will be improved through IP032 and IP005, indirectly supporting the SA objective in a localised area. IP140 would generate job opportunities which could help to address poverty.
ER2 To offer everybody the opportunity for rewarding and satisfying employment	IP140 ++ IP032 + IP005 +	Short, medium and long term Direct and indirect Reversible High	North West, transboundary, Mid Suffolk	Employment opportunities will be created during the construction phase of development. IP140 directly supports the SA Objective as during operation it would provide an opportunity for rewarding and satisfying employment.
ER3 To help meet the housing requirements for the whole community	IP140 0 IP032 ++ IP005 ++	Medium and long term Direct Reversible Medium	North West Ipswich	IP032 seeks to provide homes with an indicative capacity of 99 and 101 for IP005 which directly supports the SA Objective. Ensuring homes are designed to minimise energy use and maximise energy efficiency may also offer some benefits to reducing fuel poverty. IP140 would not contribute to the achievement of the SA objective as this site is allocated for employment use. Affordable housing will be secured through the implementation of policy CS12 in the Core Strategy.
ER4 To achieve sustainable levels of prosperity and economic growth throughout the plan area	IP140 ++ IP032 0 IP005 0	Medium and long term Indirect Reversible Medium	North West Ipswich	The provision of land for employment use would support and encourage employment and investment which would support the SA Objective.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ER5 To support vital and viable town, district and local centres	IP140 + IP032 + IP005 +	Medium and long term Indirect Reversible Medium	North West and central Ipswich	The provision of homes within 600m of an existing District Centre with IP032 and IP005 will result in an increase of residents which in the long term will serve to support the vitality and viability of the District centre. If the park and ride facility is brought back to service this will improve access to local centres which serves to support the SA Objective.
ER6 To encourage efficient patterns of movement in support of economic growth	IP140 ++ IP032 + IP005 +	Medium and long term Indirect Reversible Medium	North West Ipswich	The employment area at IP140 and adjacent park and ride facility (if brought back to use) will serve to support the SA Objective. The provision of homes with IP032 and IP005 close to existing employment areas and land allocated for employment will also encourage efficient patterns of movement and in the long term this could help to support economic growth. <i>Frequent transport provision from the park and ride facility will help to support the SA Objective.</i>
ER7 To encourage and accommodate both indigenous and inward investment	IP140 + IP032 + IP005 +	Medium and long term Indirect Reversible Medium	North West Ipswich	Development at each of these sites would encourage investment and in the long term. The increase of local residents associated with housing and employment would support economic growth which would contribute to encouraging and accommodating indigenous and inward investment.
CL1 To maintain and improve access to education and skills for both young people and adults	IP140 0 IP032 0 IP005 0	Medium and long term Direct and indirect Reversible Medium	North West Ipswich	The provision of homes within 1km of more than 10 educational facilities for IP032 and IP005 could contribute to maintaining access to education and skills for both young people and adults in the long term though it is considered that there will not be a significant change from the baseline conditions.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
CD1 To minimise potential opportunities for crime and anti-social activity	IP140 +/ - IP032 + IP005 +	Medium and long term Direct and indirect Reversible Medium	Localised	The provision of homes with IP032 and IP005 would increase natural surveillance which may help to minimise potential opportunities for crime and anti-social activity. Development at greenfield sites could increase crime from a previously crime free piece of land though design measures could mitigate any negative effects. Secured by design should be considered during the design phase in order to deter crime.

Whitehouse

IP221 Flying Horse PH, 12 homes

IP029 Land opposite 674-734 Bramford Rd, 71 homes and open space IP165 Eastway Business Park, Europa Way, 94 dwellings IP033 Land at Bramford Rd (Stocks Site), 46 homes and open space IP090 Europa Way, 18 dwellings and district centre

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET1 To improve air quality	IP221 0 IP029 - IP165 - IP033 - IP090 -	Short, medium and long term Direct and indirect Reversible Medium	Localised	70% of land at IP029 is proposed to accommodate 71 homes. In the medium to long term the localised increase in population could have a minor contribution to increasing vehicular emissions which may negatively affect local air quality in conjunction with the A14 to the east of the site. IP165, IP033, and IP090 cumulatively in the long term would result in localised population increase which could have a negative effect on air quality. There is currently limited access at IP165 and development has the potential to put pressure on Bramford Road/B1067 to the north and Europa Way to the south of the site. Bramford Road passes through an AQMA area at Chevallier Street on its way into town. It is recommended that pedestrian and cycling infrastructure is enhanced in this area.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET2 To conserve soil resources and quality	IP221 - IP029 - IP165 - IP033 - IP090 -	Medium and long term Direct and indirect Reversible Medium	Site	IP221 seeks to develop 6 dwellings and 6 flats, IP029 seeks to develop 71 homes, IP165 seeks to develop 94 homes and IP090 seeks to develop 18 homes and a district centre. Each of these sites would result in the loss of grassland and scrubland. Development at IP033 would result in the loss of grass and scrubland. There is the potential for contaminated land to be encountered at this site as land here was previously used for landfill purposes. It is recommended that site investigation is undertaken at IP033 and IP261 and where appropriate remediation should take place.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET3 To reduce waste	IP221 0 IP029 - IP165 - IP033 - IP090 -	Medium and long term Direct and indirect Reversible Medium	Site	The proposed development of 12 units at IP221 would not result in significant waste production. The provision of a significant number of homes with IP029, IP165 and IP033 and IP090 would result in a localised increase in residents and increase of construction and household waste. Whilst the provision of 18 homes at IP090 alone would not result in significant waste production combined with the new district centre there would be an increase waste production. <i>It is recommended that adequate waste</i> <i>facilities should be provided with</i> <i>development at IP029, IP165, IP033 and</i> <i>IP090. Where possible recycling facilities</i> <i>should be provided. Waste reduction</i> <i>initiatives should also be encouraged.</i>

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET4 To reduce the effects of traffic upon the environment	IP221 0 IP029 + /- IP165 - IP033 - IP090 -	Medium and long term Direct and indirect Reversible Medium	North West and North Ipswich	Due to the number of units proposed with IP221, it is not expected this site would have a significant impact on the SA Objective. The provision of 71 homes with IP029 would result in an increase of people and private vehicles which could potentially increase traffic within the local area and the A14. This site would also experience traffic noise. However the site is served by two local bus routes. It is also located in close proximity to two Local Centres (40 and 42), three existing employment areas and one allocated employment site and this would provide the opportunity for convenient and sustainable travel which could help to reduce the effects of traffic upon the environment in the long term. For these reasons, it is considered that the effects will be both positive and negative. The increase of vehicles associated with IP165 could potentially put pressure on Bramford Road/B1067 to the north and Europa Way to the south of the site particularly at peak times which could exacerbate congestion and its effects to the environment. The increase of vehicles associated with the provision of a new district centre at IP090 could potentially put pressure on Sproughton Road to the south east and cumulatively with IP165 this could also put pressure on Bramford Road/B1067. The provision of homes at IP033 would result in the increase of vehicles to and from the site which would put pressure on access routes. <i>It is recommended that green buffers are used around IP029 to account for the close proximity of the A14 and the main railway line to the south of the site. It is recommended that adequate access at IP165 and IP033 is provided where possible to avoid exacerbating peak time congestion issues. A transport assessment and travel plan is</i>

Objective	Performance	Temporal Scale	Geographical	Commentary/Mitigation
	of Policy	Nature of Impact	Extent	
		(Direct/Indirect)		
		Reversibility		
		Certainty		
				required for IP029, IP033 and IP165. Access visibility and junction spacing along Bramford Road in relation to IP033 will also be considered. This will help to reduce any potential negative effects on traffic.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET5 To improve access to key services for all sectors of the population	IP221 + IP029 + IP165 + IP033 + IP090 ++	Medium and long term Direct and indirect Reversible Medium	North Ipswich	The proposed units with IP221 would be well served with two existing Local Centres and proposed District Centre in close proximity. The site is directly adjacent to playing fields and is also in close proximity to open spaces and an allotment. IP029 would provide homes located in close proximity to local bus routes which in the medium to long term would improve access to key services around the site including open space and two Local Centres. The site will also provide a link road between Bramford Road and Europa Way subject to impacts being assessed. IP165 would provide homes located within 400m of an existing Local Centre and in close proximity to two bus routes and a proposed District Centre. IP033 would provide homes close to a Local Centre and a proposed District Centre. IP090 would provide 18 homes and a District Centre which in the long term would directly support the SA Objective.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET6 To limit and adapt to climate change	IP221 0 IP029 +/ - IP165 +/ - IP033 +/ - IP090 +/ -	Medium and long term Direct and indirect Reversible Medium	North Ipswich	IP029, IP165 and IP033 each seek to provide a significant number of homes and IP090 seeks to provide 18 homes and a District Centre which would result in a localised population increase and private cars. The increase in vehicular emissions could have a negative effect on local air quality however these sites are located in close proximity to local bus routes, Local Centres, a proposed District Centre and employment areas which may offer some mitigation. IP221 has an indicative capacity of 12; therefore it is not considered the site will have a significant impact on climate change. The replacement of derelict buildings with more energy and water efficient buildings could support the SA Objective. <i>It is recommended that where possible</i> <i>design should incorporate sustainability</i> <i>measures (e.g. reduce water use, use of</i> <i>grey water), SuDS features and ensure</i> <i>that Code for Sustainable Homes</i> <i>standards are achieved.</i>
ET7 To protect and enhance the quality of water features and resources and reduce the risk of flooding	IP221 0 IP029 0 IP165 0 IP033 0 IP090 0	Medium and long term Direct and indirect Reversible Medium	Localised	None of these site allocations are located within areas at risk of flooding. There would be no significant effects on water resources with development at grassland sites due to the size of development. However replacement of this land with development could increase the surface run-off (e.g. sites IP029 and IP033), but this could be partially offset by adequate drainage and gardens. Surface water flooding local to site – will be considered at planning application stage for IP033 and IP165. See Appendix 1 of the Ipswich SFRA. <i>It is recommended that where possible design should incorporate sustainability measures (e.g. reduce water use, use of grey water) and SuDS features.</i>
Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
--	---	---	------------------------	--
ET8 To conserve and enhance biodiversity and geodiversity, including favourable conditions on SSSIs, SPAs and SACs	IP221 ? IP029 - IP165 - IP033 - IP090 -	Medium and long term Direct and indirect Reversible Medium	Localised	 IP221 is located close to a TPO. This site currently contains trees and vegetation which would require clearance and there is the potential for some trees to provide habitat for birds. The majority of the perimeter at IP029 is surrounded by trees and areas within IP165 and IP090 contain trees and vegetation some of which may require removal for access during the construction period. There is the potential for the trees and vegetation to contain birds. IP033 is a local wildlife site which would be lost to development. The site contains trees, vegetation and a pond which are considered suitable habitats for birds and reptiles. <i>Bird and reptile surveys should be undertaken prior to construction at IP221, IP029 and IP165. Should any nesting birds or reptiles be found, appropriate mitigation measures should be put in place.</i> <i>Where possible IP221, IP029 IP033, and IP090 should include provision of soft landscaping and should seek to retain existing trees and the pond at IP033 as this may offer some benefits to biodiversity.</i>
ET9 To conserve and enhance the historic environment, heritage assets and their settings	IP221 0 IP029 0 IP165 0 IP033 0 IP090 0	N/A	N/A	The site allocations are not located close to any Listed Buildings. IP029 lies in the vicinity of Roman and Prehistoric sites. There were gravel pits across part of IP033. Bronze Age and Neolithic finds were recovered and Saxon remains were recorded to the south. Evaluation is needed to identify the impact of past land use. There are Saxon sites between this one and the river. IP090 lies in an area of Prehistoric archaeology. Part of the site is likely to have been destroyed by extraction of clay for brick-making. There is no objection in principle to development of these sites but any permission will require a condition relating to archaeological investigation

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
				attached to any planning consent. Early evaluation is advisable. Therefore it is considered the effects will be negligible on conserving or enhancing the historic environment, heritage assets and their settings.
ET10 To conserve and enhance the quality and local distinctiveness of landscapes and townscapes	IP221 + IP029 + IP165 + IP033 + IP090 +	Medium and long term Direct and indirect Reversible Medium	Localised	Development with IP221, IP029, IP165, IP033 and IP090 could contribute towards enhancing townscape quality and local distinctiveness. Ipswich Borough Council is in a process of preparing an Urban Character Supplementary Planning Document (SPD) to provide additional guidance to support policies set out in their development plan documents. Phase 1 is due for adopted in January 2015 and provides guidance about the Gipping Valley area. Phase 2 of the work (adoption expected 2016) will cover more outer areas of the town. Once adopted, this document should be consulted upon at design stage. <i>It is recommended that development at IP221, IP029, IP165, IP033 and IP090</i> <i>should be undertaken sensitively and</i> <i>should complement the existing character</i> <i>of the area which could also serve to</i> <i>enhance townscape. Where possible soft</i> <i>landscaping should be incorporated into</i> <i>design.</i>

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
HW1 To improve the health of those most in need	IP221 + IP029 + IP165 + IP033 + IP090 +	Medium and long term Direct and indirect Reversible Medium	Localised	IP221, IP029, IP165, IP033 and IP090 are located close to areas of open space. The provision of quality development could make partial contributions towards improving health in the long term through encouraging people to go outdoors and promoting healthier lifestyles.
HW2 To improve the quality of life where people live and encourage community participation	IP221 + IP029 + IP165 + IP033 + IP090 +	Medium and long term Direct and indirect Reversible Medium	Localised	IP221 is close to Local and District Centres, open space and an allotment. This could make partial contributions towards improving the quality of life by promoting healthier lifestyles encouraging community participation. IP165 seeks to provide homes in close proximity to open spaces and close to Local Centres and a proposed District Centre which collectively could make some contributions towards improving the quality of life where people live. Development at IP033 and IP090 will provide homes close to and immediately adjacent to the new District Centre which could make partial contributions towards achieving the SA Objective.
ER1 To reduce poverty and social exclusion	IP221 + IP029 + IP165 + IP033 + IP090 +	Medium and long term Indirect Reversible Low	Localised	Each of the site allocations are located close to an existing Local centre or proposed District Centre. The Local and District centres would serve each of the site allocations and contribute towards reducing social exclusion. Housing provision would improve the quality of housing stock and could contribute towards increasing quality of life for residents.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ER2 To offer everybody the opportunity for rewarding and satisfying employment	IP221 0 IP029 0 IP165 0 IP033 0 IP090 +	Short term Indirect Reversible High	North Ipswich	The new District Centre at IP090 would create employment opportunities and this would support the SA Objective. During the construction phase of development, there would be some opportunity for construction related employment, though this would be temporary and in the short term.
ER3 To help meet the housing requirements for the whole community	IP221 + IP029 ++ IP165 ++ IP033 ++ IP090 ++	Medium and long term Direct Reversible High	North Ipswich	IP221, IP029 and IP165 each support the SA Objective and seek to provide 12 units, 71 new homes and 94 new homes respectively which would help to meet the housing requirements for the whole community. IP033 and IP090 seek to provide 46 and 18 homes respectively which would directly support the SA Objective. The provision of affordable housing is addressed in Policy CS12 in the Core Strategy. Housing type and tenure are addressed in Policy CS8 in the Core Strategy.
ER4 To achieve sustainable levels of prosperity and economic growth throughout the plan area	IP221 0 IP029 0 IP165 0 IP033 0 IP090 +	Medium and long term Indirect Reversible Medium	North Ipswich	The site allocations are not considered to have a significant effect on achieving sustainable levels of prosperity and economic growth throughout the plan area. The provision of the District Centre at IP090 would be supported by the development of 18 homes and the localised increase of residents may encourage economic growth in the area.
ER5 To support vital and viable town, district and local centres	IP221 0 IP029 + IP165 + IP033 + IP090 ++	Medium and Long term Indirect and direct Reversible Medium	North Ipswich	The provision of homes with IP221 in close proximity to a Local Centre may offer some benefits to the SA Objective. However due to the small number of units proposed, benefits would be limited. The provision of a significant number of homes IP029, IP165 and IP033 in close proximity to Local Centres and a proposed District Centre would help to support the

Performance	Temporal Scale	Geographical	Commentary/Mitigation
of Policy	Nature of Impact (Direct/Indirect) Reversibility Certainty	Extent	
			vitality and viability of these centres in the long term. IP090 would directly support the vitality and viability of the proposed District Centre as it would result in increase of local residents and would be conveniently located to meet their needs.
IP221 0 IP029 + IP165 + IP033 + IP090 +	Medium to long term Indirect Reversible Low	North Ipswich	The provision of the District Centre at IP090 would be supported by the development of 18 homes. The cumulative localised population increase from development at IP033, IP029, and IP165 near community facilities would encourage efficient patterns of movement as well as sustainable and convenient travel. All of the above would encourage economic growth in the area. The sites are located close to main roads and the A14 which could support efficient patterns of movement into town centre areas. IP221 is not considered to have a significant effect on encouraging efficient patterns of movement in support of economic growth. <i>Public transport provision would support</i> <i>the SA Objective.</i>
IP221 0 IP029 + IP165 + IP033 + IP090 +	Long term Indirect Reversible Low	North Ipswich	The provision of the District Centre at IP090 would be supported by the development of 18 homes. The localised increase in population cumulatively from the sites would encourage indigenous and inward investment. IP221 is not considered to have significant effects on the SA Objective due to the small scale of development proposed.
	of Policy	of Policy Nature of Impact (Direct/Indirect) Reversibility Certainty IP221 0 IP221 0 IP029 + IP165 + IP033 + IP090 + IP221 0 IP221 0 I	of PolicyNature of Impact (Direct/Indirect) Reversibility CertaintyExtentIP221 0Medium to long term Indirect IP033 + IP090 +Medium to long term Indirect Reversible LowNorth IpswichIP221 0Medium to long term Indirect Reversible LowNorth Ipswich

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
CL1 To maintain and improve access to education and skills for both young people and adults	IP221 0 IP029 0 IP165 0 IP033 0 IP090 0	N/A	N/A	Development at these sites is not considered to have a significant effect on maintaining and improving access to education and skills for both young people and adults.
CD1 To minimise potential opportunities for crime and anti-social activity	IP221 0 IP029 + IP165 + IP033 + IP090 +	Medium and Long term Indirect Reversible Low	Localised	The provision of homes at IP029, IP165 IP033 and IP090 would increase natural surveillance which may help to minimise potential opportunities for crime and anti- social behaviour. IP221is not considered to have a significant effect on the SA Objective.

Ipswich North East

Bixley and St Johns

IP109 R/O Jupiter Road & Reading Road, 13 homes IP131 Milton Street, 13 homes IP088 79 Cauldwell Hall Road, 16 homes

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET1 To improve air quality	IP109 0 IP131 0 IP088 0	N/A	N/A	IP109 and IP131 seek to provide 13 new homes each and IP088 seeks to provide 16 homes and are unlikely to have a significant effect on air quality due to the small scale of proposed development.
ET2 To conserve soil resources and quality	IP109 + IP131 + IP088 +	Medium and long term Indirect Irreversible Medium	Site	Each of the sites will be developed on brownfield land and will not involve land take of greenfield sites which will offer some benefits to the SA Objective. It is possible that contaminated land is encountered at IP109 and IP131 and remediation would help to improve the soil resource. It is recommended that remediation works are undertaken where appropriate.
ET3 To reduce waste	IP109 - IP131 - IP088 -	Medium and long term Indirect Reversible Medium	Site	Housing development at IP109, IP131 and IP088 would result in increasing waste production at these sites. It is recommended that adequate waste facilities should be provided with new development. Where possible recycling facilities should be provided. Waste reduction initiatives should also be encouraged.
ET4 To reduce the effects of traffic upon the environment	IP109 + IP131 + IP088 +	Medium and long term Indirect Reversible Medium	North East Ipswich	Each of the sites is located close to a District Centre, two Local Centres and areas of open space. This may encourage sustainable travel locally.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET5 To improve access to key services for all sectors of the population	IP109 + IP131 + IP088 +	Medium and long term Indirect Reversible Medium	North East Ipswich	IP109, IP131 and IP088 each have good access and are located close to a District Centre, two Local Centres and areas of open space which would allow easy access to these services.
ET6 To limit and adapt to climate change	IP109 0 IP131 0 IP088 0	Medium and long term Indirect Reversible Medium	North East Ipswich	The close proximity of Local Centres, District Centres and bus routes would promote the use of public transport and sustainable travel modes. IP109, IP131 and IP088 are not thought to have significant impacts upon the SA Objective due to the number of proposed homes. It is recommended that where possible design should incorporate sustainability measures (e.g. reduce water use, use of grey water), SuDS features and ensure that Code for Sustainable Homes standards are achieved.
ET7 To protect and enhance the quality of water features and resources and reduce the risk of flooding	IP109 0 IP131 0 IP088 0	Medium and long term Indirect Reversible Medium	Localised	None of these site allocations are located within areas at risk of flooding. Surface water flooding local to site - will be considered at planning application stage for IP131. See Appendix 1 of the Ipswich SFRA. It is recommended that where possible design should incorporate sustainability measures (e.g. reduce water use, use of grey water) and SuDS features.
ET8 To conserve and enhance biodiversity and geodiversity, including favourable conditions on SSSIs, SPAs and SACs	IP109 - IP131 0 IP088 0	Medium and long term Indirect Reversible Medium	Localised	There are areas of vegetation and trees at IP109 which have the potential to contain nesting birds. IP131 and IP088 are not considered to have a significant effect on biodiversity as they are not located close to any designated sites and do not comprise any valuable habitats. It is recommended that vegetation is not removed during bird nesting season at IP109. Soft landscaping and vegetation planting should be encouraged where appropriate.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET9 To conserve and enhance the historic environment, heritage assets and their settings	IP109 0 IP131 0 IP088 0	N/A	N/A	St Johns Baptist Church Grade II Listed Building is within 400m of IP131 and IP088. However at such distance it is not anticipated that there would be a direct negative effect on the setting of the Listed Building. There are no heritage assets located near or within IP109.
ET10 To conserve and enhance the quality and local distinctiveness of landscapes and townscapes	IP109 + IP131 + IP088 +	Medium and long term Indirect Reversible Medium	Localised	Development at each of these site allocations could contribute towards enhancing quality and local distinctiveness through modern design and the replacement of existing structures. Ipswich Borough Council is in a process of preparing an Urban Character Supplementary Planning Document (SPD) (Phase 1 to be adopted in January 2015, Phase 2 for outer Ipswich in 2016) to provide additional guidance to support policies set out in their development plan documents. Once adopted, this document should be consulted upon at design stage. <i>It is recommended that development at each of the sites should be undertaken sensitively and should complement the existing character of the area which could also serve to enhance townscape. Where possible soft landscaping should be incorporated into design.</i>
HW1 To improve the health of those most in need	IP109 + IP131 + IP088 +	Medium and long term Indirect Reversible Low	Localised	Each of these sites is located close to protected playing fields which could make partial contributions towards improving health in the long term through encouraging people to go outdoors and promoting healthier lifestyles. High quality housing would also help to support the SA Objective.
HW2 To improve the quality of life where people live and encourage community participation	IP109 + IP131 + IP088 +	Medium and long term Indirect Reversible Low	Localised	Development at each of these sites would be close to District Centres, Local Centres and protected playing fields which collectively could make contributions towards improving the quality of life by promoting healthier lifestyles encouraging

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
				community participation. High quality housing would also help to support the SA Objective.
ER1 To reduce poverty and social exclusion	IP109 + IP131 + IP088 +	Medium and long term Indirect Reversible Low	North East Ipswich	IP109, IP131 and IP088 will be located close to an existing District Centre which provides community facilities and could contribute to reducing social exclusion.
ER2 To offer everybody the opportunity for rewarding and satisfying employment	IP109 0 IP131 0 IP088 0	Short term Indirect Reversible High	North East Ipswich	There would be the opportunity for construction related employment. Though this would be opportunities would be limited to the construction phase.
ER3 To help meet the housing requirements for the whole community	IP109 + IP131 + IP088 +	Medium and long term Direct Reversible High	North East Ipswich	IP109 and IP131 each seek to provide 13 homes and IP088 seeks to provide 16 homes. Each of these site allocations directly supports the SA Objective.
ER4 To achieve sustainable levels of prosperity and economic growth throughout the plan area	IP109 0 IP131 0 IP088 0	N/A	N/A	Housing provision would not significantly contribute to achieving sustainable levels of prosperity and economic growth.
ER5 To support vital and viable town, district and local centres	IP109 + IP131 + IP088 +	Medium and long term Indirect Reversible Medium	North East Ipswich	The provision of homes with each of these site allocations close to existing District Centre 18 and Local Centres 19 and 36 would encourage residents to use these facilities which would support the vitality and viability of these centres.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ER6 To encourage efficient patterns of movement in support of economic growth	IP109 + IP131 + IP088 +	Medium and long term Indirect Reversible Low	North East Ipswich	Efficient patterns of movement will be encouraged as the sites are located near existing District and Local Centres. The area is well served by public transport.
ER7 To encourage and accommodate both indigenous and inward investment	IP109 + IP131 + IP088 +	Medium and long term Indirect Reversible Medium	North East Ipswich	The cumulative development at these sites would encourage investment and in the long term. The increase of people would cumulatively serve to support local business which would contribute to encouraging and accommodating indigenous and inward investment.
CL1 To maintain and improve access to education and skills for both young people and adults	IP109 0 IP131 0 IP088 0	Medium and long term Indirect Reversible Low	North East Ipswich	There is one high school within 500m of IP109, IP131 and IP088. The provision of homes at these sites would make some contributions towards maintaining access to education skills for young people and adults though contributions are considered to have a negligible effect.
CD1 To minimise potential opportunities for crime and anti-social activity	IP109 + IP131 + IP088 +	Medium and long term Indirect Reversible Low	Localised	The provision of homes may result in increase of natural surveillance which may in turn help to minimise potential opportunities for crime and anti-social behaviour.

Ipswich Central

Westgate

IP105 Depot, Beaconsfield Road, 15 homes IP246 158-160 London Road, 13 homes IP135 112-116 Bramford Road, 24 homes IP130 South of South Street, 11 homes

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET1 To improve air quality	IP105 0 IP246 0 IP135 0 IP130 0	Medium and long term Indirect Reversible Low	Central	IP246 and IP105 are sites currently in use therefore due to the proposed number of housing at these sites it is not considered that the increase on private cars as a result of development would significantly increase vehicular emissions. Cumulatively the housing proposed at IP135, IP130 and IP105 would potentially have a negative effect on local air quality, however the level of certainty is low. The provision of a cycle path to the south west boundary of IP105 linking with the existing riverside cycle paths to the north and south of the site would make partial contributions to promoting cycling though it would not contribute significantly to achieving the SA Objective. It is recommended that where possible public transport links should be provided within 400m of development.
ET2 To conserve soil resources and quality	IP105 + IP246 + IP135 + IP130 +	Medium and long term Indirect Irreversible Medium	Site	Each of these sites would be developed on previously developed land. Contaminated land may be encountered at IP105 and IP135. Remediation works would benefit the soil resource. <i>Contaminated land should be</i> <i>appropriately remediated at IP105 and</i> <i>IP135.</i>

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET3 To reduce waste	IP105 - IP246 - IP135 - IP130 -	Medium and long term Indirect Reversible Medium	Site	Cumulatively the site allocations will result in increase of construction and household waste. Adequate waste facilities should be provided with new development. Where possible recycling facilities should be provided. Waste reduction initiatives should also be encouraged.
ET4 To reduce the effects of traffic upon the environment	IP105 0 IP246 0 IP135 0 IP130 0	Medium and long term Indirect Reversible Medium	Localised	IP105 and IP246 propose 15 and 13 homes respectively. Effects upon traffic during operation would not increase significantly due to the number of new homes proposed though there would be a temporary increase in construction traffic. IP130 proposes 11 new homes which would not have a significant impact on air quality. IP135 proposed 24 homes and <i>similarly</i> will only have a negligible effect on traffic the area.
ET5 To improve access to key services for all sectors of the population	IP105 + IP246 + IP135 + IP130 +	Medium and long term Indirect Reversible Medium	Localised	IP105 is relatively close to the existing Local Centre 35 which would contribute to the achievement of the SA objective. The provision of a cycle path along the south west boundary of the site linking with existing riverside cycle paths to the north and south would also contribute to improving access to key services. Similarly IP246, IP135 and IP130 are located relatively close to District Centre 7, and Local Centre 35 which would provide easy access to these services in the long term.
ET6 To limit and adapt to climate change	IP105 0 IP246 0 IP135 0	Medium and long term Indirect	Central	Housing provision at these sites is considered to have a very minor effect on vehicular emissions due to the number of

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
	IP130 0	Reversible Medium		 houses proposed. Therefore no significant effects to the SA Objective are likely to occur. Where possible design should incorporate sustainability measures (e.g. reduce water use, use of grey water), SuDS features and ensure that Code for Sustainable Homes standards are achieved. These sites should ensure that homes are designed to maximise energy efficiency which may also contribute to reducing fuel poverty.
ET7 To protect and enhance the quality of water features and resources and reduce the risk of flooding	IP105 0 IP246 0 IP135 0 IP130 0	Medium and long term Indirect Reversible Medium	Localised	Provision of houses would increase water use at each of these sites though not to a significant amount. With sites currently in use, no significant drainage problems would be envisaged. None of these site allocations are located within areas at risk of flooding. It is recommended that where possible design should incorporate sustainability measures (e.g. reduce water use, use of grey water) and SuDS features.
ET8 To conserve and enhance biodiversity and geodiversity, including favourable conditions on SSSIs, SPAs and SACs	IP105 0 IP246 0 IP135 - IP130 0	Medium and long term Indirect Reversible Medium	Localised	There are TPOs at IP135 which may be affected by new development. It is recommended that where possible TPOs should be incorporated into design at IP135. Each of the site allocations should maximise the use of soft landscaping and vegetation planting which would offer some benefits to the SA Objective.
ET9 To conserve and enhance the historic environment, heritage assets and their settings	IP105 0 IP246 0 IP135 0 IP130 0	N/A	N/A	There are no heritage assets within 100m of the sites therefore no impacts are recorded for this set of sites.
ET10 To conserve and enhance the quality and local	IP105 + IP246 + IP135 + IP130 +	Medium and long term Indirect Reversible	Central	Development with each of these site allocations could contribute towards enhancing townscape quality by implementing careful design techniques

Performance	Temporal Scale	Geographical	Commentary/Mitigation
of Policy	Nature of Impact (Direct/Indirect)	Extent	
	Reversibility		
	Certainty		
	Medium		that integrate with the local character of the area. Ipswich Borough Council is in a process of preparing an Urban Character Supplementary Planning Document (SPD) (Phase 1 to be adopted in January 2015, Phase 2 in 2016) to provide additional guidance to support policies set out in their development plan documents. Once adopted, this document should be consulted upon at design stage. <i>It is recommended that development at each of the sites should complement the existing character of the area which could also serve to enhance townscape. Where possible soft landscaping should be</i>
IP105 + IP246 + IP135 0 IP130 0	Medium and long term Indirect Reversible Low	Central	<i>incorporated into design.</i> Development at IP105 is located close to a protected playing field and would provide a cycle path to the south west boundary of the site connecting with existing riverside cycle paths.IP246 is located close to an area of woodland. Each of these factors could make partial contributions towards improving health in the long term through encouraging people to cycle and go
			outdoors and promoting healthier lifestyles. No significant effects with IP135 and IP130 are considered likely.
IP105 + IP246 + IP135 + IP130 +	Medium and long term Indirect Reversible Medium	Central	The provision of housing with these site allocations would help to improve quality and availability of homes cumulatively within the local area. Housing would also be provided close to Local and District Centres which would contribute to healthier lifestyles and encouraging community participation.
	of Policy	of Policy Nature of Impact (Direct/Indirect) Reversibility Certainty Medium	of Policy Nature of Impact (Direct/Indirect) Reversibility Certainty Extent IP105 + IP135 0 IP130 0 Medium Image: Central stress of the stress

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ER1 To reduce poverty and social exclusion	IP105 + IP246 + IP135 + IP130 +	Medium and long term Indirect Reversible Low	Central	The provision of housing close to Local and District Centres would contribute to improving access to community facilities and reducing social exclusion.
ER2 To offer everybody the opportunity for rewarding and satisfying employment	IP105 0 IP246 0 IP135 0 IP130 0	Short term Indirect Reversible High	Central	There would be the opportunity for construction related employment though this would be limited to the construction phase.
ER3 To help meet the housing requirements for the whole community	IP105 + IP246 + IP135 + IP130 +	Medium and long term Direct Reversible High	Central	Each of these site allocations seeks to provide housing within the local area which directly supports the SA Objective.
ER4 To achieve sustainable levels of prosperity and economic growth throughout the plan area	IP105 0 IP246 0 IP135 0 IP130 0	N/A	N/A	The sites do not have a significant effect on encouraging employment and investment.
ER5 To support vital and viable town, district and local centres	IP105 + IP246 + IP135 + IP130 +	Medium and long term Indirect Reversible Medium	Central	The provision of homes with each of these site allocations close to existing District and Local Centres would help to encourage residents to use these facilities in the long term which would partially support the vitality and viability of these centres.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ER6 To encourage efficient patterns of movement in support of economic growth	IP105 + IP246 + IP135 + IP130 +	Medium and long term Indirect Reversible Medium	Central	The cumulative minor increase in population close to existing centres would help to encourage efficient patterns of movement. There are bus links close to each of these sites which would encourage sustainable and convenient travel. All of the above would encourage economic growth.
ER7 To encourage and accommodate both indigenous and inward investment	IP105 + IP246 + IP135 + IP130 +	Medium and long term Indirect Reversible Medium	Central	Development at these sites may help to encourage investment and in the long term. The cumulative population increase would support local business which would contribute to encouraging and accommodating indigenous and inward investment. Providing the right type of housing in the right locations could also help attract or retain skills to/in lpswich which are key to attracting businesses.
CL1 To maintain and improve access to education and skills for both young people and adults	IP105 0 IP246 0 IP135 0 IP130 0	N/A	N/A	The site allocations are not considered to have an effect on maintaining and improving access to education and skills for both young people and adults.
CD1 To minimise potential opportunities for crime and anti-social activity	IP105 + IP246 + IP135 + IP130 +	Medium and long term Indirect Reversible Low	Central	The provision of homes with each of the site allocations may help to increase natural surveillance which may in turn help to minimise potential opportunities for crime and anti-social behaviour.

Castle Hill and St Margarets - a

IP009 Victoria Nurseries, Westerfield Road, 12 homes IP256 Artificial hockey pitch, Ipswich Sports Club, subject to the requirements of policy DM28 being met, 18 homes

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET1 To improve air quality	IP009 0 IP256 0	Medium and long term Indirect Reversible Medium	Central	The provision of 12 homes at IP009 and 18 homes IP256 would cumulatively result in a minor increase in private cars which in the medium to long term may have a very small contribution to an increase in vehicular emissions locally. There is however one local bus route in close proximity to each of these sites which may encourage more sustainable travel. Both are located within one mile of the town centre where there are job opportunities and services available. For these reasons effects on air quality are considered to be negligible.
ET2 To conserve soil resources and quality	IP009 + IP256 +	Medium and long term Indirect Irreversible Medium	Central	Each of these sites would be developed on previously developed land. It is possible that contaminated land may be encountered at IP256. Remediation works would help to improve the soil resource. It is recommended that remediation works take place at IP256.
ET3 To reduce waste	IP009 - IP256 -	Medium and long term Indirect Reversible Medium	Central	The proposed development with each of the site allocations would increase waste production. It is recommended that adequate waste facilities should be provided with development at these sites. Where possible recycling facilities should be provided. Waste reduction initiatives should also be encouraged.
ET4 To reduce the effects of traffic upon the environment	IP009 0 IP256 0	Medium and long term Indirect Reversible Medium	Central	Housing provision will have a potential negative effect on emissions and their effect on the environment though on the whole impacts would be negligible. There is one local bus route which serves each of these site allocations which may help to reduce the effects of traffic upon the environment the long term.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET5 To improve access to key services for all sectors of the population	IP009 + IP256 +	Medium and long term Indirect Reversible Low	Central	IP256 is located close to Local Centre 4 and both of the site allocations are located close to a large open space area which partially supports the SA Objective.
ET6 To limit and adapt to climate change	IP009 - IP256 -	Medium and long term Indirect Reversible Medium	Central	Housing provision at these sites would have a potential negative effect on vehicular emissions however each of the sites are located close to a local bus route and the town centre, which could encourage people to use public transport, cycle or walk. <i>It is recommended that where possible public transport links should be provided within 400m of development. Where possible design should incorporate sustainability measures (e.g. reduce water use, use of grey water), SuDS features and ensure that Code for Sustainable Homes standards are achieved.</i> <i>These sites should ensure that homes are designed to maximise energy efficiency which may also contribute to reducing fuel poverty.</i>
ET7 To protect and enhance the quality of water features and resources and reduce the risk of flooding	IP009 0 IP256 -	Medium and long term Indirect Reversible Medium	Central	There may be a drainage constraint at IP256 due to the previous use of the site as a sports pitch. None of the site allocations are located within areas at risk of flooding. It is recommended that where possible design should incorporate sustainability measures (e.g. reduce water use, use of grey water) and SuDS features. Surface water flooding local to site - will be considered at planning application stage. See Appendix 1 of the Ipswich SFRA.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET8 To conserve and enhance biodiversity and geodiversity, including favourable conditions on SSSIs, SPAs and SACs	IP009 - IP256 -	N/A	N/A	Each of these sites contains TPOs which could be affected by development. Where possible TPOs should be incorporated into design and soft landscaping and tree planting should be encouraged to support biodiversity.
ET9 To conserve and enhance the historic environment, heritage assets and their settings	IP009 0 IP256 0	N/A	N/A	The Spinney including Car Port and Log Store Grade II Listed Building is located within 100m of IP009. However as development would not affect the Listed Building setting no negative effects are considered likely.
ET10 To conserve and enhance the quality and local distinctiveness of landscapes and townscapes	IP009 + IP256 +	Medium and long term Indirect Reversible Medium	Central	Development of these site allocations could contribute towards enhancing townscape quality and local distinctiveness through careful design. Ipswich Borough Council is in a process of preparing an Urban Character Supplementary Planning Document (SPD) (Phase 1 to be adopted in January 2015, Phase 2 in 2016) to provide additional guidance to support policies set out in their development plan documents. Once adopted, this document should be consulted upon at design stage. It is recommended that development should complement the existing character of the area which could also serve to enhance townscape. Where possible soft landscaping should be incorporated into design.
HW1 To improve the health of those most in need	IP009 + IP256 +	Medium and long term Indirect Reversible Low	Central	Each of these proposed developments are located close to a large area of open space which could make minor contributions towards improving health in the long term through encouraging people to go outdoors and promoting healthier lifestyles.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
HW2 To improve the quality of life where people live and encourage community participation	IP009 + IP256 +	Medium and long term Indirect Reversible Medium	Central	The provision of housing with these site allocations would improve quality and availability of homes within the local area. Housing would also be provided close to a Local Centre at IP256 which would contribute to healthier lifestyles and encouraging community participation. IP256 is currently an all-weather sports pitch which will be lost to a housing development however mitigated through Core Strategy policy DM28.
ER1 To reduce poverty and social exclusion	IP009 0 IP256 +	Medium and long term Indirect Reversible Low	Central	The provision of housing close to a Local Centre at IP256 would contribute to improving access to community facilities and reducing social exclusion. IP009 is not considered to have a significant effect on the SA Objective.
ER2 To offer everybody the opportunity for rewarding and satisfying employment	IP009 0 IP256 0	Short term Indirect Reversible High	Central	Construction jobs may become available in the short term; however it is considered that on the whole the effect will be negligible.
ER3 To help meet the housing requirements for the whole community	IP009 + IP256 +	Medium and long term Direct Reversible High	Central	Each of these site allocations seeks to provide housing within the local area which directly supports the SA Objective.
ER4 To achieve sustainable levels of prosperity and economic growth throughout the plan area	IP009 0 IP256 0	N/A	N/A	The sites would not support the SA Objective.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ER5 To support vital and viable town, district and local centres	IP009 0 IP256 +	Medium and long term Indirect Reversible Medium	Central	The provision of homes at IP256 close to an existing Local Centre would encourage residents to use these facilities which would support the vitality and viability of these centres. IP009 is not considered to have a significant effect on the SA Objective as it is not located close to any Local/District Centres.
ER6 To encourage efficient patterns of movement in support of economic growth	IP009 0 IP256 +	Medium and long term Indirect Reversible Medium	Central	The population increase associated with IP256 close to the existing Local Centre would encourage efficient patterns of movement. There are bus links close to this site which would encourage sustainable and convenient travel all of which would encourage economic growth. IP009 is not considered to have a significant effect on the SA Objective due it its location and distance from a Local or a District Centre.
ER7 To encourage and accommodate both indigenous and inward investment	IP256 +	Medium and long term Indirect Reversible Medium	Central	Development at IP256 would encourage investment and in the long term. The increase in numbers of local residents would support local business which would contribute to encouraging and accommodating indigenous and inward investment. IP009 is not considered to have a significant effect on the SA Objective due to its location.
CL1 To maintain and improve access to education and skills for both young people and adults	IP256 0	N/A	N/A	The site allocations are not considered to have an effect on maintaining and improving access to education and skills for both young people and adults.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
CD1 To minimise potential opportunities for crime and anti-social activity	IP009 + IP256 +	Medium and long term Indirect Reversible Low	Central	The provision of homes with each of the site allocations may help to increase natural surveillance which may in turn help to minimise potential opportunities for crime and anti-social behaviour.

Ipswich South East

Holywells

IP066 - JJ Wilson, White Elm Street – 18 dwellings IP010a - Co-op Depot, Felixstowe Road – 66 dwellings IP010b - Felixstowe Road – 75 dwellings

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET1 To improve air quality	IP066: - IP010a: - IP010b: -	Long term Indirect Reversible Medium	Localised	Provision of extra housing at IP010a and IP010b could result in minor traffic and emissions increase. IP066 is within an AQMA and any additional traffic associated with housing provision may result in further deterioration of air quality. However, the site is located close to the town centre which could facilitate walking and cycling, and it is currently in industrial use so there is some traffic generation already. <i>Use of sustainable modes of transport</i> <i>should be encouraged.</i>
ET2 To conserve soil resources and quality	IP066: + IP010a: + IP010b: +	Long term Indirect Irreversible Low	Site	Housing within all of these areas will increase the green space through the provision of gardens and open spaces (all are primarily hard surfaced or built on at present). The existing land use associated with commercial and industrial properties at IP010a and IP010b will be removed and replaced. Potential for contaminated land has been identified along the railway track. <i>Appropriate remediation techniques and survey for contaminated land should be conducted during design of any scheme.</i>
ET3 To reduce waste	IP066: - IP010a: - IP010b: -	Long term Direct Reversible High	Site	The provision of housing will potentially increase the volume of household waste produced within these areas. <i>Recycling</i> <i>household waste initiatives would mitigate</i> <i>any potential increase</i> . Industrial and commercial waste will be reduced at IP010a and IP101b through removal of these land uses. However, there will be associated removal of construction and demolition waste.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET4 To reduce the effects of traffic upon the environment	IP066: - IP010a: 0 IP010b: 0	Long term Indirect Reversible High	Localised	All units will realise an increase in residential traffic. Construction traffic will increase temporarily. IP066 is already in AQMA and even a slight increase in air pollution would be considered significant. IP010a and IP010b will realise a reduction in commercial and industrial traffic, through elimination of this land use. This may broadly offset the increase in residential traffic. A transport assessment and travel plan will be required for IP010a and IP010b which will help to reduce any potential negative effects on traffic. <i>Provision of public transport will enable the increase in traffic volumes to be minimised.</i>
ET5 To improve access to key services for all sectors of the population	IP066: + IP010a: + IP010b: +	Long term Indirect Reversible High	Localised	IP010a and IP010b are in close proximity to an existing District Centre 23 (along Felixstowe Road) and development within these areas would not hinder access to key services. IP066 is in close proximity to District Centre 21 (along Foxhall Road) and central Ipswich, therefore development would allow access to key services. As part of the development of IP010a, Rose Hill Primary School will be extended and provide an increased level of service to the immediate area. <i>Any development should include provision to maintain and improve accessibility and connectivity to the green spaces within the district.</i>
ET6 To limit and adapt to climate change	IP066: + IP010a: + IP010b: +	Long term Indirect Reversible High	Ipswich – South East	Residential development is expected at IP066, IP010a, and IP010b. This will replace existing buildings with energy efficient new buildings including insulation and heating; therefore replacing existing inefficient practices with modern standards. These would be in line with the Code for Sustainable Homes.
ET7 To protect and	IP066: + IP010a: +	Long term Indirect	Sites	Removal of existing hard standing and car parks and replacement with open space

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
enhance the quality of water features and resources and reduce the risk of flooding	IP010b: +	Irreversible Low		and landscaping will improve run off and aid ground absorption. The implementation of SuDS will offer benefits with regard to drainage and may reduce flood risk within the area. These methods should be studied and applied to any proposals where applicable. Surface water flooding local to site - will be considered at planning application stage for IP066. See Appendix 1 of the Ipswich SFRA.
ET8 To conserve and enhance biodiversity and geodiversity, including favourable conditions on SSSIs, SPAs and SACs	IP066: +/- IP010a: +/- IP010b: +/-	Long term Indirect Irreversible Low	Localised	All of the sites are presently of low biodiversity value. However, IP066 is located within close proximity to a County Wildlife Site. To the south of the site is an area of land designated as open space with geologically important features. IP010a and IP010b border the railway line, where a wildlife corridor exists. <i>Proposals that may have a negative impact on wildlife habitats should not be accepted. Opportunities for enhancement should be sought.</i> Tree Preservation Orders (TPOs) are present within (or nearby) IP010a and IP010b; <i>any development would be required to adhere to any conditions associated with these TPOs.</i>
ET9 To conserve and enhance the historic environment, heritage assets and their settings	IP066: 0 IP010a: 0 IP010b: 0	Long term Indirect Irreversible Low	Localised	There are three listed buildings within 150m of the development areas. The Church of St Bartholomew (Grade II), Uplands (Grade II) and Boundary Wall, Gate piers and Gate at Uplands (Grade II). None of the areas will directly impact on any known features of historical interest nor will the proposals affect the setting of these listed buildings.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET10 To conserve and enhance the quality and local distinctiveness of landscapes and townscapes	IP066: 0 IP010a: 0 IP010b: 0	Long term Indirect Irreversible Low	Localised	New housing developments at all sites should be designed to be sensitive and in- keeping with local character. It is likely that the construction of new housing at the sites in place of existing developments will have a slight positive impact upon the townscape. However, the impact is considered to be negligible. There are no listed buildings or conservation areas within proximity to the sites. IP010a and b fall within the California Urban Character Area in the Urban Character SPD (to be adopted January 2015) and IP066 is adjacent to the Parks area. The SPD provides design guidance.
HW1 To improve the health of those most in need	IP066: + IP010a: + IP010b: +	Long term Direct Irreversible Medium	Site	New development at IP066, IP010a and IP010b will add improved quality housing to the area. Associated recreational areas and gardens should improve the health and amenity for local residents. IP010a is designated to include some community usage through extension to the local primary school.
HW2 To improve the quality of life where people live and encourage community participation	IP066: + IP010a: + IP010b: +	Long term Direct Reversible High	Site	Provision of new housing at the sites will increase the quality and availability of housing stock in eastern Ipswich. New housing will be constructed to a higher standard and the council will ensure that affordable housing will be provided. The provision of affordable housing is addressed in Policy CS12 in the Core Strategy.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ER1 To reduce poverty and social exclusion	IP066: 0 IP010a: + IP010b: 0	Long term Indirect Reversible Low	Localised	Provision of new housing at these sites will increase the quality and availability of housing stock in eastern Ipswich. IP010a and IP010b are situated in close proximity to the District centre 23 (along Felixstowe Road). IP010a is allocated for community usage, which is likely to include the extension to a local school. Employment will be created within the construction sector for the new housing provision, the extension of Rose Hill School and associated amenity improvements.
ER2 To offer everybody the opportunity for rewarding and satisfying employment	IP066: + IP010a: - IP010b: +	Short term Indirect Reversible Low	Localised	Employment will be created across all sites during the construction period of the new housing developments. However, some IP010a existing employment site will be lost. IP066 and IP010b will displace existing economic activities so they will need to be relocated within Ipswich if the jobs benefit is to be retained. Increased job opportunities will be available for the District Centre 23 (along Foxhall Road).
ER3 To help meet the housing requirements for the whole community	IP066: + IP010a: + IP010b: +	Long term Indirect Reversible High	Localised	Provision of new housing at IP066, IP010a and IP010b will increase the quality and availability of housing stock in eastern Ipswich. It is envisaged that an element of affordable housing will be included in any future development. The provision of affordable housing is addressed in Policy CS12 in the Core Strategy.
ER4 To achieve sustainable levels of prosperity and economic growth throughout the plan area	IP066: 0 IP010a: 0 IP010b: 0	Long term Indirect Reversible High	Site	Employment and investment would be encouraged across IP066, IP010a and IP010b during the design and construction of the proposed development. However, the short term nature of this employment results in a negligible impact. An existing employment area is located to the north of IP066. An increase in the number of local residents could provide a

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation workforce and expansion opportunity for the businesses within this area. <i>Improved</i> <i>transport, connectivity and access routes</i> <i>should be encouraged to link the new</i> <i>developments to the employment area</i>
ER5 To support vital and viable town, district and local centres	IP066: + IP010a: + IP010b: +	Long term Indirect Reversible High	Localised	The developments at IP066, IP010a, IP010b all offer new housing stock and are in close proximity to two local district centres 21 & 23 (along Foxhall Road and Felixstowe Road) containing shops and retail services. Although none of the development sites are primarily designated for commercial uses, their residents will use local services and add to the economy of the area and also that of Ipswich town centre.
ER6 To encourage efficient patterns of movement in support of economic growth	IP066: + IP010a: + IP010b: +	Long term Indirect Reversible High	Ipswich – South East	New housing created at IP066, IP010a and IP010b will provide an available workforce to the surrounding district centres, adjacent town centre and designated employment area. There are presently bus routes in close proximity to the sites along Felixstowe Road to central lpswich.
ER7 To encourage and accommodate both indigenous and inward investment	IP066: + IP010a: + IP010b: +	Long term Indirect Reversible High	Localised	IP066, IP010a and IP010b are all housing developments. There will be investment into the areas for the design and construction of these sites. There are limited opportunities for new business to develop on these sites; however, the improvement in housing stock and accessibility will benefit surrounding business and attract investment. IP066 and IP010b will displace existing economic activities so they will need to be relocated within Ipswich if the jobs benefit is to be retained.
CL1 To maintain and improve access to education and skills for both young people and	IP066: 0 IP010a: + IP010b: 0	Long term Direct Reversible High	Localised	The developments will provide new housing and some amenity in open space. IP010a will ensure that community facilities are maintained and extend a valuable resource in Rose Hill Primary

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
adults				School. The extension to Rose Hill Primary School in IP010a could contribute to achieving higher educational attainment and increase living standards within the local area.
CD1 To minimise potential opportunities for crime and anti-social activity	IP066: + IP010a: + IP010b: +	Medium term Indirect Reversible High	Site	IP066, IP010a and IP010b would provide new housing to the local area; this will likely be an improvement on existing stock and increase community pride. Employment during construction will attract investment to the area and could have beneficial effects on crime levels thorough improved design and accessibility. The extension to Rose Hill Primary School in IP010a could contribute to achieving higher educational attainment and increase living standards within the local area. <i>Proposals should incorporate safety in design principles and increase natural surveillance thereby minimising opportunities for crime and anti-social behaviour.</i>

Gainsborough

IP067 – Former British Energy Site – employment allocation IP058 - Former Volvo site, Raeburn Rd South – employment allocation IP099 - Part of former Volvo Site, Raeburn Road South – employment allocation

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET1 To improve air quality	IP067: - IP058: - IP099: -	Long term Direct Reversible High	Localised	The area around IP067, IP099 and IP058 is presently affected by odour from the nearby sewage works. This has been identified as an existing barrier to development.
				The proposals create employment areas at IP067, IP099 and IP058. It is considered likely that traffic volumes and emissions will increase. Public transport provision would mitigate some of this increase. The closest existing bus routes are to the north and east of the sites along Landseer Road. The sites are also next to a large residential area which could supply a labour force for future employment uses <i>Bus improvements should be sought to</i> <i>reduce impacts on air quality if the</i> <i>proposals are completed.</i>
ET2 To conserve soil resources and quality	IP067: + IP058: + IP099: +	Long term Direct Irreversible High	Site	 IP067 is formerly the British Energy site; there may be some contaminants in this area. Site investigation may be required prior to any development. Remediation could be necessary and would improve the quality of soil resource. IP099 and IP058 are former Volvo sites and are presently consisting predominantly of hard standing car parking/depot and structures. <i>Remediation works to be undertaken where appropriate.</i>
ET3 To reduce waste	IP067: - IP058: - IP099: -	Long term Direct Reversible High	Site	Creation of employment areas at IP067, IP099 and IP058 will create industrial/commercial waste. This can be offset by the implementation of recycling schemes. Some demolition work and removal of contaminants may be required and this waste should be disposed of appropriately off-site.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation Any contaminated excavated ground should be disposed of appropriately.
ET4 To reduce the effects of traffic upon the environment	IP067: - IP058: - IP099: -	Long term Indirect Reversible High	Localised	The proposals create employment areas at IP067, IP099 and IP058 with likely increase in traffic volumes and emissions. Public transport provision would mitigate some of this increase. The closest existing bus routes are to the north and east of the sites along Landseer Road. A transport assessment and travel plan will be required for the sites which will help to reduce any potential negative effects on traffic. An improvement to the public transport network and green travel plans should be developed to include access to these sites.
ET5 To improve access to key services for all sectors of the population	IP067: + IP058: + IP099: +	Long term Indirect Reversible Medium	Localised	IP058 and IP067 contain areas of open space which should be developed to provide an amenity or recreational asset for the local area. IP099 should contribute to the development of open spaces within these sites.
ET6 To limit and adapt to climate change	IP067: +/- IP058: +/- IP099: +/-	Long term Indirect Reversible Low	Ipswich – South East	New employment areas and open space at IP067, IP058 and IP099 will replace derelict and former use buildings with more energy and water efficient structures. However, overall emissions would still increase.
ET7 To protect and enhance the quality of water features and resources and reduce the risk of flooding	IP067: + IP058: + IP099: +	Long term Indirect Irreversible High	Localised	None of the areas designated for employment use are within the floodplain. It is likely that the removal of existing depot/car parking structures and replacement with landscaped developments will decrease the potential run off by increasing ground absorption. Surface water flooding local to site - will be considered at planning application stage for IP067 and IP099. See Appendix 1 of the Ipswich SFRA. <i>Suitable drainage systems should be</i>

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
				minimised.
ET8 To conserve and enhance biodiversity and geodiversity, including favourable conditions on SSSIs, SPAs and SACs	IP067: - IP058: - IP099: -	Long term Direct Irreversible High	Localised	IP058 is presently a County Wildlife Site. <i>It</i> would be beneficial for any development on this site to promote the wildlife where possible associated with the CWS as per the recommendations in the 2012/2013 Wildlife Audit. IP067 and IP099 also have potential to
				create areas with rich biodiversity and IP067 contains part of a Local Wildlife Site. IP099 and IP058 are close to the Pipers Vale Local Nature Reserve (LNR) to the south. <i>Opportunities should be sought to</i> <i>enhance wildlife corridors by linking the</i> <i>areas of open space.</i>
				Consultation with Natural England would be beneficial to determine potential enhancement of habitats.
				It is important that any development is in keeping and does not deteriorate the sensitive nature of the protected sites.
ET9 To conserve and enhance the historic environment, heritage assets and their settings	IP067: 0 IP058: 0 IP099: 0	Long term Direct Irreversible Low	Localised	There are no heritage assets within close proximity of IP067, IP058, and IP099.
ET10 To conserve and enhance the quality and local distinctiveness of landscapes and townscapes	IP067: +/- IP058: +/- IP099: +/-	Long term Direct Irreversible Low	Localised	Development at the disused sites IP058, IP099 and IP067 should incorporate green spaces to add to the wildlife sites and corridors in the locality. Development should encourage an improvement to the natural and man-made landscape. It is likely that negative impacts will be realised on the landscape of the CWS, however positive impacts could be realised through the development of derelict sites.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation Design of this development should be sensitive and encourage a continuation of the features contained within the protected sites.
HW1 To improve the health of those most in need	IP067: 0 IP058: 0 IP099: 0	Long term Direct Reversible Low	Localised	The creation of employment areas at IP067, IP058 and IP099 should attract investment into the area, potentially resulting in a slight increase in wages and standard of living for the local population, however this impact is assessed to be negligible. The improvement of the local nature reserves and green space should be encouraged to allow access and recreational use of these sites. Removal and appropriate disposal of potentially contaminated land within all areas could bring added health benefits.
HW2 To improve the quality of life where people live and encourage community participation	IP067: + IP058: + IP099: +	Long term Direct Irreversible Low	Localised	The creation of employment areas at IP067, IP058 and IP099 would create jobs and an increase in wages and standard of living for the local population. The improvement of the County Wildlife Site and green space should be encouraged to provide a wildlife corridor and if possible allow access and recreational use within IP058 and IP067.
ER1 To reduce poverty and social exclusion	IP067: + IP058: + IP099: +	Long term Indirect Irreversible Low	Ipswich – South East	Creation of employment areas at IP058, IP067 and IP099 will attract investment and create jobs. Nearby residents north of Sandyhill Road should benefit from close proximity to employment areas. The Index of Multiple Deprivation highlights that the area containing these development sites is toward the most deprived. Creation of employment and associated improvement in amenities, transport linkages and townscape will benefit the area.

Objective ER2 To offer everybody the opportunity for rewarding and satisfying employment	Performance of Policy IP067: + IP058: + IP099: +	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty Long term Direct Reversible High	Geographical Extent	Commentary/Mitigation Employment will be created within IP067, IP058, and IP099. This will be permanent employment during operation and temporary employment during the construction period.
ER3 To help meet the housing requirements for the whole community	IP067: 0 IP058: 0 IP099: 0	Long term Indirect Irreversible Low	Localised	There are no housing proposals associated with the development policies for IP058, IP067, and IP099.
ER4 To achieve sustainable levels of prosperity and economic growth throughout the plan area	IP067: + IP058: + IP099: +	Long term Direct Reversible High	Ipswich – South East	Creation of employment areas at IP058, IP067 and IP099 will attract investment and create jobs. Nearby residents north of Sandyhill Road should benefit from close proximity to employment areas.
ER5 To support vital and viable town, district and local centres	IP067: + IP058: + IP099: +	Long term Indirect Reversible High	Localised	Creation of employment areas at IP058, IP067 and IP099 will bring investment to an area where there are several disused industrial units. The proposals should support existing employment areas and alongside appropriate investment in public transport and connectivity should increase usage of the area and the nearest local centres 25 and 26.
ER6 To encourage efficient patterns of movement in support of economic growth	IP067: + IP058: + IP099: +	Long term Direct Reversible High	Ipswich – South East	Creation of employment areas at IP058, IP067 and IP099 should provide economic growth to the area. Improvement of public transport links and green space connectivity should encourage efficient movement of a workforce. Access from the local residential area north of Sandyhill Lane should be encouraged to reduce traffic, commuting distances and provide an available local workforce to enable business investment.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ER7 To encourage and accommodate both indigenous and inward investment	IP067: + IP058: + IP099: +	Long term Direct Reversible High	Ipswich – South East	The primary aim of the policies at IP067, IP058, and IP099 is to encourage and accommodate investment. To encourage inward investment it would be beneficial to encourage existing businesses and employers within the nearby employment area and district centre 26 (along Landseer Road) by improving access and connectivity between the sites.
CL1 To maintain and improve access to education and skills for both young people and adults	IP067: 0 IP058: 0 IP099: 0	Long term Indirect Reversible High	Localised	Pipers Vale School is situated on Sandyhill Lane and increased investment in the surrounding area could bring benefits. Creation of employment areas at IP058, IP067 and IP099 should provide economic growth to the area and potentially raise the standard of living within the local community. An increase in public transport and green space connectivity could benefit the students of the school. However, these effects are more likely to be cumulative in combination with other proposals, therefore the performance of the each site is assessed to be negligible.
CD1 To minimise potential opportunities for crime and anti-social activity	IP067: + IP058: + IP099: +	Long term Indirect Reversible High	Localised	Creation of employment areas at IP058, IP067 and IP099 should attract investment and job creation. The living standards within the local area could also rise. The development of disused sites at IP058, IP067, and IP099 will improve the local amenity and vibrancy. Disused sites can be associated with crime and anti- social behaviour. By bringing these back into acceptable use, then the community should benefit. Secured by design principles and an increase in natural surveillance may reduce crime levels within the area.
Priory Heath B - Ravenswood

IP149 - Pond Hall Carr and Farm; extension to Orwell Country Park

IP150a - Ravenswood T (adjacent Fen Bight Circle) Ravenswood U, V, W 138 new dwellings

IP150b - Land south of Ravenswood; sports facilities

IP150c - Land south of Ravenswood; employment allocation

IP152 - Airport Farm Kennels, north of the A14; employment area

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET1 To improve air quality	IP149: 0 IP150a: - IP150b: - IP150c: - IP152: -	Long term Indirect Reversible Medium	Localised	 Extension to Orwell Country Park at IP149 is unlikely to have a significant effect on air quality. There could be a slight increase in vehicle numbers visiting the site, but this is negligible. New housing provision at IP150a will likely lead to an increase in traffic movements due to population increase. <i>To mitigate these effects, proposals to improve the public transport system in south east Ipswich (frequency, bus station provision, passenger information) should be considered.</i> IP150b includes the development of sporting facilities which may increase visitors and traffic to the area. <i>Cycling and walking should be encouraged through appropriate infrastructure and public realm improvements.</i> IP152 proposes the establishment of an employment area, which will likely increase traffic to the area. <i>This could be mitigated by increasing public transport provision.</i> It should be noted that cumulatively the allocation of homes, employment and recreation facilities within a small area could encourage efficient patterns of movement in the long term.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET2 To conserve soil resources and quality	IP149: + IP150a: - IP150b: - IP150c: - IP152: -	Long term Indirect Reversible Low	Site	Proposal to extend the country park at IP149 will extend the level of protection of soils within this area. New housing developments, sporting facilities and employment areas at IP150a, IP150b, IP150c and IP152 are likely to cause some disturbance to the soils. Contaminated areas may be uncovered and will require specialist disposal.
ET3 To reduce waste	IP149: 0 IP150a: - IP150b: 0 IP150c: - IP152: -	Long term Direct Reversible High	Site	New housing provision at IP150a will result in an increase in household waste generated within the area. <i>This could be</i> offset by the promotion of recycling schemes throughout the area. Creation of a new employment zone at IP152 and IP150c will lead to an increase in waste during construction and operation. <i>This could be offset by recycling</i> schemes. It is considered unlikely that the extension of the country park at IP149 will lead to any waste increase.
ET4 To reduce the effects of traffic upon the environment	IP149: 0 IP150a: - IP150b: - IP150c: - IP152: -	Long term Indirect Reversible High	Localised	Extension to the country park in IP149 could realise a slight increase in vehicle numbers visiting the site, but this is negligible. New housing provision at IP150a may lead to an increase in traffic movements due to population increase. IP152 and IP150c proposes the establishment of an employment area, which may increase traffic to the area. <i>To mitigate these</i> <i>effects, proposals to improve the public</i> <i>transport system in south east lpswich</i> (<i>frequency, bus station provision,</i> <i>passenger information</i>) <i>should be</i> <i>considered.</i> IP150b includes the development of sport facilities which may increase visitors and traffic to the area. A transport assessment and travel plan will be required for IP150c and IP152 which will help to reduce any potential negative effects on traffic. <i>Cycling and walking should be encouraged</i> <i>through appropriate infrastructure and</i>

Objective	Performance	Temporal Scale	Geographical	Commentary/Mitigation
	of Policy	Nature of Impact (Direct/Indirect)	Extent	
		Reversibility		
		Certainty		
				public realm improvements.
ET5	IP149: + IP150a: 0	Long term	Localised	Extension to the country park in IP149 will
To improve access to key services for all	IP150b: +	Direct Reversible		provide additional recreational opportunity for the local population.
sectors of the population	IP150c: 0 IP152: 0	Low		Housing development at IP150a will not provide key services directly, however, increases in population should encourage key service provision and this should be considered during the planning and design stage. It is also considered that the development could lead to an increase in
				the viability of Ravenswood District Centre. IP150b adds sport facilities and will directly
				benefit the key service provision within the area.
				IP152 and IP150c do not add any key services to the area but workers could access Ravenswood District Centre.
ET6 To limit and adapt to climate change	IP149: 0 IP150a: - IP150b: - IP150c: - IP152: -	Long term Indirect Irreversible Low	Ipswich – South East	Proposals to extend the country park at IP149 will not have an impact upon climate change. New housing development, sport facilities and employment areas at IP150a, IP150b,
				IP150c and IP152 should be constructed using energy efficient techniques and materials. The increase in traffic associated with these developments could have an adverse impact; <i>this could be</i> <i>offset by improvements to the public</i>
				transport system.
ET7	IP149: 0 IP150a: -	Long term	Sites	None of the sites are presently within the floodplain.
To protect and enhance the quality of water features and resources and reduce the risk of flooding	IP150b: - IP150c: - IP152: -	Direct Irreversible High		Housing developments, sporting facilities and location of businesses within IP150a, IP150b, IP150c and IP152 could replace areas of open space with housing,
				structures, associated drainage and necessary road infrastructure. This could increase the surface run-off. <i>Proposals</i> <i>should investigate whether SuDS would be</i>
				appropriate within this area. Surface water flooding local to site – will be considered at planning application stage for IP150b and IP150c. See Appendix 1 of the Ipswich SFRA.
				Extension to the country park in IP149 is

Objective	Performance	Temporal Scale	Geographical	Commentary/Mitigation
	of Policy	Nature of Impact (Direct/Indirect)	Extent	
		Reversibility		
		-		
		Certainty		
				unlikely to have any impact upon flooding or water resources.
ET8 To conserve and enhance biodiversity and geodiversity, including favourable conditions on SSSIs, SPAs and SACs	IP149: • IP150a: • IP150b: • IP150c: • IP152: •	Long term Direct / Indirect Irreversible High	Localised	The extension of the country park at Pond Hall Farm (IP149) could enhance the biodiversity within the area. The site is bordered by the internationally designated SPA/Ramsar and nationally designated SSSI. Visitors' management measures will be implemented to ensure the Orwell Estuary SPA is protected and any disturbance of birds is minimised. The Habitats Regulations Assessment (2014) has concluded that Policy SP8, regarding Site IP149, will not result in a likely significant effect upon any European site. The addendum to the Habitats Regulations Assessment (September 2015) reaches the same conclusions taking into account pre-submission additional modifications to the proposed allocation. Housing development at IP150a could benefit biodiversity if adequate planting and landscaping is included within any proposals. IP150b has been highlighted as an area where reptiles and invertebrate species are potentially present. <i>Development of any proposals in this area will be required to be sensitive to any protected species.</i> <i>Appropriate surveys should be conducted prior to any development.</i> Creation of the employment areas at IP152 and IP150c may have an adverse impact upon biodiversity. At present approximately one third of the site IP152 is designated as Area of Outstanding Natural Beauty (AONB), which could be lost if development proceeds. <i>Mitigation measures to offset the effects on species</i>
				and habitats should be included within the proposals.
				For the area as a whole, the increase in connectivity through retention of a network
				of green space will ensure further impacts
				upon biodiversity are minimised.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET9 To conserve and enhance the historic environment, heritage assets and their settings	IP149: 0 IP150a: 0 IP150b: 0 IP150c: 0 IP152: 0	Long term Indirect Irreversible Low	Localised	Pond Hall Farm is a listed building (Grade II) however the country park proposal is not expected to have a significant effect on it. Sites IP150a, b and c lie in the vicinity of Prehistoric remains and cropmarks, and as such any necessary mitigation measures should be addressed at an appropriate stage in the planning process. IP152 includes a known Bronze Age barrow (IPS 027). This indicates that there are other prehistoric archaeological remains of high importance in the vicinity. As such any necessary mitigation measures should be addressed at an appropriate stage in the planning process.
ET10 To conserve and enhance the quality and local distinctiveness of landscapes and townscapes	IP149: + IP150a: 0 IP150b: 0 IP150c: 0 IP152: 0	Long term Indirect Irreversible Low	Localised	There is the possibility that these developments may result in adverse impact upon views and open space. Therefore, the new housing developments, sport facilities and employment areas at sites IP150a, IP150b, IP150c and IP152 should be designed to be sensitive and in- keeping with local character. There are no listed buildings or conservation areas within proximity to the sites. New landscaping will be required at IP152 to maintain the view from the A14 across the site. IP149 will develop the existing country park and would potentially add benefits to the landscape character. Proposed facilities should be designed to blend in with the surroundings.
				Where possible linkages between the green spaces should be encouraged to improve the landscape characteristics. Existing countryside features, such as hedgerows, should be maintained at IP152 wherever possible'.
HW1 To improve the health of those most in need	IP149: + IP150a: + IP150b: + IP150c: + IP152: +	Long term Direct Reversible High	Site	New development at IP150a will provide high quality housing to the area. Associated recreational areas and gardens should improve the amenity for local residents. IP150b will develop sport facilities,
of those most in need	IP150c: +			should improve the amening residents.

Objective	Performance	Temporal Scale	Geographical	Commentary/Mitigation
	of Policy	Nature of Impact (Direct/Indirect)	Extent	
		Reversibility		
		Certainty		
				IP149 will extend the existing countryside park and increase the recreational area for use by the public.
				Employment opportunities at IP152 and IP150c would offer benefits with regard to mental health.
				For these reasons, it is considered that the sites will offer some benefits for the local residents with regard to both physical and mental health.
HW2 To improve the quality of life where people live and encourage community participation	IP149: + IP150a: + IP150b: + IP150c: + IP152: +	Long term Direct Reversible High	Site	New housing development at IP150a will improve the quality of the housing stock within the area. Associated recreational areas would improve both physical and mental health and may increase community participation. Any proposals should include provision for open spaces for community use.
				IP152 and IP150c will offer some additional employment offering indirect benefits to mental health and well-being.
				IP149 and IP150b will provide additional recreational facilities for the local population. IP149 will provide health benefits to the area through preservation and extension of the country park.
ER1 To reduce poverty and social exclusion	IP149: + IP150a: + IP150b: + IP150c: +	Long term Indirect Reversible	Localised	The area containing IP149, IP150a, IP150b, IP150c and IP52 is ranked as third most deprived on the national scale. (1 most deprived – 10 least deprived).
	IP152: +	High		The creation of employment at IP152 and IP150c could have a positive impact on the quality of life and standard of living within the area.
				New housing provision will improve the quality of the stock and increase the quality of life for residents.
				Community facilities, such as the sport facilities at IP150b and the country park extension at IP149 will encourage community participation. <i>The proposals</i> <i>should include plans to maintain in good</i> <i>condition these community assets in the</i> <i>long term and provide linkages and</i> <i>connectivity where possible.</i>

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ER2 To offer everybody the opportunity for rewarding and satisfying employment	IP149: 0 IP150a: 0 IP150b: 0 IP150c: + IP152: +	Long term Direct Reversible High	Localised	Short term employment will be created with the construction of new housing and improvements to sport and country park facilities at IP149, IP150a and IP150b. It is unlikely that these provisions would provide long term opportunities for employment although the country park extension could if a visitors' centre is provided on site. Development at IP152 and IP150c has the potential to offer long term employment to the local population.
ER3 To help meet the housing requirements for the whole community	IP149: 0 IP150a: + IP150b: 0 IP150c: 0 IP152: 0	Long term Direct Reversible High	Localised	IP149, IP150b and c, and IP152 will not create further housing provision for the community. IP150a will create 138 (when complete) new homes. The provision of affordable housing is addressed in Policy CS12 in the Core Strategy.
ER4 To achieve sustainable levels of prosperity and economic growth throughout the plan area	IP149: + IP150a: + IP150b: + IP150c: + IP152: +	Long term Direct Reversible High	Site	Creation of employment areas at IP152 and IP150c will attract investment and create jobs. Nearby residents along Nacton Road and Landseer Road should benefit from close proximity to the employment areas. Linkages should be sought with the existing employment area to the east of Nacton Road to extend the scope of investment in the area. Improved housing and community facilities at IP149, IP150a and IP150b will assist in regeneration of the area and potentially attract further investment.
ER5 To support vital and viable town, district and local centres	IP149: + IP150a: + IP150b: + IP150c: + IP152: +	Long term Direct Reversible High	Localised	Creation of an employment area at IP152 and IP150c will bring investment to the area. The proposals should support the existing employment areas to the east of Nacton Road and alongside appropriate investment in public transport and connectivity should increase usage of the area. Improved housing and community facilities at IP149, IP150a and IP150b will assist in regeneration of the area and potentially attract further investment. This will benefit

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation the existing District Centre 47 (Ravenswood).
ER6 To encourage efficient patterns of movement in support of economic growth	IP149: + IP150a: + IP150b: + IP150c: + IP152: +	Long term Direct Reversible High	Ipswich – South East	Creation of employment areas at IP152 and IP150c should provide economic growth to the area. New housing and sport facilities will provide temporary employment during the construction phase. The location of the sites at the periphery of lpswich, mix of uses and improvement of public transport links and green space connectivity should encourage efficient movement of the workforce. The use of sustainable modes of transport should be encouraged to reduce traffic, commuting distances and provide easy access to
ER7 To encourage and accommodate both indigenous and inward investment	IP149: + IP150a: + IP150b: + IP150c: + IP152: +	Long term Direct Reversible High	Localised	 employment areas. The primary aim of the allocations at IP152 and IP150c is to encourage and accommodate investment. An increase in high quality housing and an improvement in the landscape at IP149, IP150a and IP150b could make the area more attractive to investors. To encourage inward investment it would be beneficial to improve access and connectivity between the existing and planned developments.
CL1 To maintain and improve access to education and skills for both young people and adults	IP149: + IP150a: 0 IP150b: + IP150c: 0 IP152: +	Long term Direct Irreversible Medium	Localised	A visitor centre at IP149 would raise awareness of the potential impacts of disturbance on birds on the estuary. Ravenswood School is situated on Ravenswood Ave, just adjacent to IP150a and increased investment in the surrounding area could bring benefits. Creation of employment areas at IP152 and IP150c should encourage economic growth locally and potentially raise the standard of living within the local community.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
CD1 To minimise potential opportunities for crime and anti-social activity	IP149: 0 IP150a: + IP150b: + IP150c: 0 IP152: 0	Long term Indirect Reversible High	Site	Creation of employment areas at IP152 and IP150c should attract investment and job creation. The living standards within the local area could also rise as a result. The development of IP150a and IP150b should help regenerate the area and raise living standards. <i>Proposals should include</i> <i>open/green spaces for community use.</i> <i>Secured by design principles should be</i> <i>considered during the design stage.</i> It is unlikely that the extension of the county park at IP149 will have an impact upon crime or anti-social behaviour.

Priory Heath C – The Havens

IP146 - Ransomes Europark East; three employment areas

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET1 To improve air quality	IP146: -	Long term Direct Reversible High	Localised	The proposals create employment areas at three locations at IP146. These areas are likely to generate traffic in peak hours and have a negative effect on air quality. It is likely that these traffic effects will be in combination with the effect of the other employment areas nearby. There are bus services running through the Ransomes Europark, with connection close to the Makro store. Sustainable modes of transport should be encouraged through the implementation of Travel plans.
ET2 To conserve soil resources and quality	IP146: -	Long term Direct Irreversible High	Site	IP146 is presently an area of unused fields. Development on greenfield land could have a negative effect on soil resource. Compensation measures may be required if the soil is of good agricultural quality.
ET3 To reduce waste	IP146: -	Long term Direct Reversible High	Site	The proposed employment area at IP146 will be constructed on open space/farmland. There will be an inevitable increase in waste from the site once operational. <i>However, this can be partly</i> <i>mitigated by encouraging recycling</i> <i>schemes.</i>
ET4 To reduce the effects of traffic upon the environment	IP146: -	Long term Direct Reversible High	Localised	The proposals create employment areas at IP146 and may lead to an increase in traffic volumes and emissions. It is likely that these negative effects will be in combination with the negative effects of other employment areas nearby. There are bus services running through the Ransomes Europark, with connection close to the Makro store. A transport assessment and travel plan will be required and will help to reduce any

Objective	Performance	Temporal Scale	Geographical	Commentary/Mitigation
	of Policy	Nature of Impact (Direct/Indirect) Reversibility Certainty	Extent	
				potential negative effects on traffic.
				Sustainable modes of transport should be encouraged.
ET5 To improve access to key services for all sectors of the population	IP146: 0	Long term Indirect Reversible Low	Localised	Creation of employment areas at IP146 will not provide any key services to the local area.
ET6 To limit and adapt to climate change	IP146: -	Long term Indirect Irreversible Low	Ipswich – South East	New employment areas at IP146 will be constructed on previously undeveloped land. To mitigate this, the proposals should include energy and water efficiency measures in line with the BREEAM requirements.
ET7 To protect and enhance the quality of water features and resources and reduce the risk of flooding	IP146: -	Long term Direct Reversible High	Sites	None of the areas designated for employment use are within the floodplain. It is likely that the removal of existing fields and grassland and replacement with hard standing or buildings will increase run off and decrease ground absorption. <i>SuDS should be included within any</i> <i>proposals along with landscaped areas</i>
ET8 To conserve and enhance biodiversity and geodiversity, including favourable conditions on SSSIs, SPAs and SACs	IP146: -	Long term Direct Irreversible High	Localised	where possible. IP146 is presently located close to a small local nature reserve. It would be beneficial for any development to include open space and habitat enhancements to increase the benefits for biodiversity and wildlife associated with the LNR. The three development sites are located on previously undeveloped land and are
				close to the periphery of Ipswich and the countryside; therefore any proposed development should be sensitive to ecological receptors. <i>Surveys for protected</i> <i>species should be conducted prior to any</i> <i>proposals being approved.</i>
ET9 To conserve and enhance the historic environment, heritage assets and their settings	IP146: 0	Long term Indirect Reversible Low	Localised	There are no known heritage assets within close proximity of IP146.

Objective	Performance	Temporal Scale	Geographical	Commentary/Mitigation
	of Policy	Nature of Impact (Direct/Indirect) Reversibility Certainty	Extent	
ET10 To conserve and enhance the quality and local distinctiveness of landscapes and townscapes	IP146: 0	Long term Indirect Reversible Low	Site	Development at IP146 will be upon greenfield land. Negative effects on landscape are likely to occur. Cumulative loss of the urban fringe and development of the site for employment is reduced by the existing industrial backdrop. <i>Design of</i> <i>this development should be sensitive and</i> <i>encourage a continuation of the features</i> <i>contained within the surrounding</i> <i>countryside.</i>
HW1 To improve the health of those most in need	IP146: +	Long term Indirect Reversible Low	Site	The creation of employment areas at IP146 should attract investment into the area, resulting in an increase in wages and standard of living for the local population. This may have an indirect positive impact on mental health and well-being.
HW2 To improve the quality of life where people live and encourage community participation	IP146: 0	Long term Indirect Reversible Low	Site	The creation of employment areas at IP146 would create some jobs. However, this is unlikely to have a significant impact in isolation.
ER1 To reduce poverty and social exclusion	IP146: +	Long term Indirect Reversible Low	Localised	Creation of employment areas at IP146 will attract investment and create some jobs. Nearby residents in the Ravenswood and Gainsborough areas should benefit from close proximity to employment areas. The location of the site at the edge of the borough and close to a large road results in a reduced effect upon the local population. The Index of Multiple Deprivation highlights that the area containing these development sites is toward the least deprived. <i>The development should</i> <i>include some improvements for public</i> <i>transport and local amenity to benefit local</i> <i>residents.</i>
ER2 To offer everybody the opportunity for rewarding and satisfying employment	IP146: +	Long term Direct Reversible High	Localised	There will be permanent employment opportunities during operation and temporary employment during the construction period.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ER3 To help meet the housing requirements for the whole community	IP146: 0	Long term Indirect Reversible Low	Localised	There are no housing proposals associated with IP146.
ER4 To achieve sustainable levels of prosperity and economic growth throughout the plan area	IP146: +	Long term Direct Reversible High	Site	Creation of employment areas at IP146 will attract investment and create jobs. Nearby residents in Ravenswood and Gainsborough should benefit from close proximity to employment areas. Linkages between businesses should be sought with the existing employment area and those to the west of Nacton Road. This would extend the scope of investment in the area. The cumulative effect of these sites is likely to be positive in combination with the existing employment areas.
ER5 To support vital and viable town, district and local centres	IP146: +	Long term Direct Reversible High	Localised	Creation of employment areas at IP146 will attract investment in the long term. The proposal should support existing district and local centres and alongside appropriate investment in public transport and connectivity should increase usage of the employment area and the surrounding Europark. The proposals are likely to have little impact upon town centres when considered in isolation, however, the effect on the larger area will be positive.
ER6 To encourage efficient patterns of movement in support of economic growth	IP146: +	Long term Direct Reversible High	Ipswich – South East	Creation of employment areas at IP146 should increase economic growth to the area. <i>Improvement of public transport links and</i> <i>green space connectivity should be</i> <i>encouraged to provide efficient movement</i> <i>of a workforce.</i> Access from the local residential area around Ravenswood should be encouraged to reduce traffic, commuting distances and enable business investment.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ER7 To encourage and accommodate both indigenous and inward investment	IP146: +	Long term Direct Reversible High	Localised	The primary aim of site allocation IP146 is to encourage and accommodate investment. It would be beneficial to improve access and connectivity between the existing employment areas and IP146 to encourage existing businesses within the nearby employment area and district centre 47 (Ravenswood) to invest locally. Cumulative effects on the larger employment area are likely be positive.
CL1 To maintain and improve access to education and skills for both young people and adults	IP146: 0	Long term Indirect Reversible Low	Localised	There will be no impact upon educational assets as a result of IP146. There could be the provision of training to the workforce, but is assessed to be of negligible impact.
CD1 To minimise potential opportunities for crime and anti-social activity	IP146: 0	Long term Indirect Reversible Low	Site	Creation of employment areas at IP146 should attract investment and job creation. Secured by design principles should be considered during the design stage.

Ipswich South West

Gipping

IP147 - Land between railway junction and Hadleigh Road; creation of an employment area
IP059a - Elton Park Industrial Estate; creation of 105 new dwellings
IP059b - Arclion House, Hadleigh Road; creation of 18 new dwellings
IP061 - School Site, Lavenham Road; creation of 30 new dwellings and open space
IP168 - Stoke Park Drive - creation of 11 new dwellings

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility	Geographical Extent	Commentary/Mitigation
ET1 To improve air quality	IP147: - IP059a: - IP059b: - IP061: - IP168: -	Certainty Long term Indirect Reversible High	Localised	 Provision of housing at IP059a, IP059b, IP061, IP168 and the creation of an employment area at IP147 could result in extra traffic and related emissions through an influx of residents and workers. The area is not within an AQMA. There is public transport provision along nearby London Road and Dickens Road. The effect of additional traffic within the area could be offset by the improvement of public transport services to increase the frequency and geographical scope. Sustainable modes of transport should be encouraged.
ET2 To conserve soil resources and quality	IP147: + IP059a: + IP059b: + IP061: - IP168: +	Long term Direct Irreversible High	Site	IP061 proposes to use previously undeveloped grassland/open space to build 30 new homes. This will create disturbance to the existing soil resource and reduce the quality. Creation of housing developments at IP059a, IP059b, IP168 and an employment area at IP147 is on previously developed land. <i>Remediation works</i> <i>should be undertaken if contaminated land</i> <i>is found.</i>
ET3 To reduce waste	IP147: - IP059a: - IP059b: - IP061: - IP168: -	Long term Direct Reversible High	Site	Sites IP059a, IP059b, IP061, IP168 provide additional residential units and IP147 an employment area. This may result in an increase in household and commercial waste in the medium and long term. <i>Recycling schemes will reduce the</i> <i>impact of additional waste.</i>

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET4 To reduce the effects of traffic upon the environment	IP147: - IP059a: - IP059b: - IP061:- IP168: -	Long term Direct Reversible High	Localised	The provision of housing at IP059a, IP059b, IP061, and IP168 may lead to an increase in traffic as a result of residents increase in numbers into the area.
				The creation of designated employment area at IP147 will potentially lead to an increase in traffic movements in peak hours of the day.
				There are existing bus services along Dickens Road to the south and east of the sites and London Road to the south. Improvements in these services, such as bus stop provision and passenger information may offset any impacts from increased traffic as a result of the developments.
				A transport assessment and travel plan will be required for IP059a and IP147 which will help to reduce negative effects on traffic.
				Sustainable modes of transport should be encouraged.
ET5 To improve access to key services for all sectors of the population	IP147: 0 IP059a: 0 IP059b: 0 IP061: - IP168: 0	Long term Indirect Reversible Low	Localised	Housing developments at IP059a, IP059b, IP061, and IP168 will not provide key services directly, however, increases in population would encourage key service provision around Local centre 8.
				IP061 is developing an area of open space/playing fields. Some will be retained for use by the existing and new residents. Improvements to the amenity of this space should be considered to offset impacts of the development.
ET6 To limit and adapt to climate change	IP147: - IP059a: - IP059b: - IP061: - IP168: -	Long term Indirect Irreversible Low	Ipswich – West	New housing development and employment areas at IP147, IP059a, IP059b, IP168 and IP061 should be constructed using energy efficient techniques and materials. The increase in traffic associated with these developments could have an adverse impact, which could be offset by the improvement to public transport services.The retention of some area of open space
				at IP061 would also offset some of the impacts.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET7 To protect and enhance the quality of water features and resources and reduce the risk of flooding	IP147: - IP059a: 0 IP059b: 0 IP061: - IP168: -	Long term Direct Reversible High	Sites	Housing developments at IP168 and IP061 would replace areas of open space with housing, ancillary structures and necessary road infrastructure. This could increase the surface run-off and the risk of flooding. Drainage issues should be taken into
				account through the detailed planning of the development at planning application stage.
				A small section to the north of IP059a is adjacent to Flood zone 2 and 3. This may increase the risk of flooding unless mitigated.
				Surface water flooding local to site – will be considered at planning application stage for IP147. See Appendix 1 of the Ipswich SFRA.
				Any development within this area should be sensitive to the flood risk issues and not increase downstream residual issues.
ET8 To conserve and enhance biodiversity and geodiversity, including favourable conditions on SSSIs, SPAs and SACs	IP147: 0 IP059a: - IP059b: - IP061: 0 IP168: 0	Long term Direct Irreversible High	Localised	Housing developments at IP061, IP059a, IP059b, and IP168 could benefit biodiversity if adequate planting and landscaping is included within any proposals. There are tree preservation orders (TPOs) in place at IP059a and IP059b; any proposals should be sensitive to these natural assets.
				The creation of an employment area at IP147 is adjacent to the River Gipping, which is a County Wildlife Site. <i>Mitigation</i> <i>measures should be put in place to ensure</i> <i>any potential impacts upon the site are</i> <i>avoided or minimised. Development</i> <i>should be sensitive to the natural</i> <i>environment of the river and protect</i> <i>biodiversity resources.</i>
ET9 To conserve and enhance the historic	IP147: 0 IP059a: 0 IP059b: 0	Long term Indirect Reversible	Localised	There are no known heritage assets within close proximity to IP147, IP059a, IP059b, and IP168.
environment, heritage assets and their settings	IP061: - IP168: 0	Low		Crane Hall, a Grade II Listed Building, is adjacent to the proposed housing development site at IP061. The proposal should not have a direct impact upon the listed building or the setting, but <i>should be</i>

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
				use suitable construction techniques to avoid any visual impact.
ET10 To conserve and enhance the quality and local distinctiveness of landscapes and townscapes	IP147: 0 IP059a: 0 IP059b: 0 IP061: - IP168: 0	Long term Indirect Reversible Low	Site	New housing developments and employment areas at IP147, IP059a, IP059b and IP061, and IP168 should be designed to be sensitive and in-keeping with local character. IP061 is to be constructed on open space therefore may have a negative impact on townscape only in localised area.
				 There is a listed building in close proximity to IP061, therefore <i>construction and setting considerations should be taken into account prior to development.</i> Where possible linkages between the green spaces should be encouraged to improve the landscape characteristics.
HW1 To improve the health of those most in need	IP147: + IP059a: + IP059b: + IP061: + IP168: +	Long term Indirect Reversible Low	Site	New housing development at IP059a, IP059b, IP061 and IP168 will provide high quality housing to the area. Associated recreational areas and gardens would benefit mental and physical health of the local residents. IP061 will include some open space for recreational use of residents. The creation of an employment area at IP147 should attract investment and result in creation of job opportunities. This may benefit mental health and well- being in the long term.
HW2 To improve the quality of life where people live and encourage community participation	IP147: + IP059a: + IP059b: + IP061:+ IP168: +	Long term Indirect Reversible Low	Site	New housing development at IP059a, IP059b, IP061, and IP168 will improve the quality of the housing stock within the area. Associated recreational areas should improve health of local residents and increase community participation. <i>Any</i> <i>proposals at IP059a, IP059b, and IP168</i> <i>should include provision of open spaces</i> <i>for community use.</i> These are already provided for within the development at IP061. The creation of an employment area at IP147 should attract investment and

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
				employment to the area, therefore raising the standard of living within the area. <i>Linkages should be improved to ensure</i> <i>that the residents have easy access to the</i> <i>employment opportunities.</i>
ER1 To reduce poverty and social exclusion	IP147: + IP059a: + IP059b: + IP061:+ IP168: +	Long term Indirect Reversible Low	Localised	The area containing all sites is ranked at 5828 on the national deprivation scale (1=most deprived, 32482=least deprived). Therefore the creation of employment at IP147 could have a positive impact on the quality of life and standard of living within the area.
				New housing provision will improve the quality of the housing stock and improve the quality of life.
				The creation of an employment area at IP147 will attract investment and should result in more job opportunities for the local residents.
ER2 To offer everybody the opportunity for rewarding and satisfying employment	IP147: ++ IP059a: 0 IP059b: 0 IP061: 0 IP168: 0	Long term Direct Reversible High	Localised	Short term employment will be created with the construction of new housing facilities at IP059a, IP059b, IP168 and IP061. It is unlikely that these provisions would provide long term opportunities for employment. Development at IP147 has the potential to offer long term employment opportunities.
ER3 To help meet the housing requirements for the whole community	IP147: 0 IP059a: + IP059b: + IP061: + IP168: +	Long term Indirect Reversible Low	Localised	IP059a will create 105 new homes, IP059b will create 18, IP168 will create 11, and IP061 will create 30. There is no provision for new housing within the employment area at IP147.
				The provision of affordable housing is addressed in Policy CS12 in the Core Strategy.
ER4 To achieve sustainable levels of prosperity and economic growth throughout the plan area	IP147: ++ IP059a: + IP059b: + IP061: + IP168: +	Long term Direct Reversible High	Site	Creation of an employment area at IP147 will attract investment and create jobs. Nearby residents along London Road, Kelly Road and the surrounding estate would benefit from close proximity to the employment areas. Linkages should be sought with the
				existing employment area adjacent IP147 to extend the scope of investment in the area and increase cumulative positive impacts.

Objective	Performance	Temporal Scale	Geographical	Commentary/Mitigation
	of Policy	Nature of Impact (Direct/Indirect)	Extent	
		Reversibility		
		Certainty		
				Improved housing and community facilities at IP059a, IP059b, IP168 and IP061 will assist in regeneration of the area and potentially attract further investment when in combination with each other. The District Centre (8) at the eastern end of Kelly Road and the one at Stoke Park Drive (11) should also benefit from the influx of new residents and employees.
ER5 To support vital and	IP147: + IP059a: +	Long term Direct	Localised	Creation of an employment area at IP147 will bring investment to the area.
viable town, district and local centres	IP059b: + IP061: + IP168: +	Reversible High		The proposals should support the existing employment areas to the east of IP147; the Hadleigh Road industrial estate. Improved housing at IP059a, IP059b, IP168 and IP061 will assist in regeneration of the area and potentially attract further investment. This will benefit the District Centre (8) along Kelly Road and Dickens Road and District Centre (11) along Stoke Park Drive.
ER6 To encourage efficient patterns of movement in support of economic growth	IP147: + IP059a: + IP059b: + IP061: + IP168: 0	Long term Direct Reversible High	Ipswich – West	Creation of employment area at IP147 should provide economic growth to the area. New housing at IP059a, IP059b, IP168 and IP061 will provide temporary employment during the construction phase. The location of the housing developments and employment areas in close proximity will have a positive impact.
				IP168 is located at a distance from existing employment areas and may have a negative impact but it is considered to be negligible due to the size of the site. Improvement of public transport links and green space connectivity should encourage efficient movement of a workforce. Sustainable modes of transport should be encouraged.
ER7 To encourage and accommodate both indigenous and inward investment	IP147: + IP059a: + IP059b: + IP061: + IP168: +	Long term Direct Reversible High	Localised	An increase in high quality housing and an improvement of townscape quality at IP059a, IP059b, IP061, and IP168 could make the area more attractive to investors.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
CL1 To maintain and improve access to education and skills for both young people and adults	IP147: + IP059a: 0 IP059b: 0 IP061: 0 IP168: 0	Long term Indirect Reversible Low	Localised	The proposals at IP059a, IP059b, IP061, IP147, and IP168 will not directly affect the educational assets within the area. Creation of an employment area at IP147 may include provision of employee training and offer benefits in the long term.
CD1 To minimise potential opportunities for crime and anti-social activity	IP147: + IP059a: + IP059b: + IP061: + IP168: +	Long term Indirect Reversible Low	Site	Creation of an employment area at IP147 should attract investment and job creation. The development of IP059a, IP059b, IP061, and IP168 should improve the housing stock within the area and result in economic regeneration, which should raise living standards.

IP – One Area

Central – St. Margaret's, Alexandra

IP172: 15-19 St. Margaret's Green, 9 homes IP176: 7-9 Woodbridge Road, 14 homes IP214: 300 Old Foundry Road, 11 homes IP260: The Former Odeon Cinema, leisure uses

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET1 To improve air quality	IP172 0 IP176 0 IP214 0 IP260 -	Medium and long term Direct and indirect Irreversible Medium	Central Ipswich	IP172, IP176, and IP214 are already within AQMA, and IP260 is in close proximity to AQMA. The development of the sites may potentially increase air pollution during construction in the short term. The main use of the land (IP172, IP176, and IP214) will be residential, which may lead to additional use of private cars of any new residents. However, the sites are relatively small in size and the overall effect on traffic would be negligible due to the central location near most community facilities. IP260 is allocated for leisure uses/community facility and the increase of traffic to the site may be determined by the available car parking spaces and frequency of the local bus service that runs through the area. <i>Opportunities should be sought to encourage sustainable modes of transport through enhancement of the pedestrian infrastructure and creation of cycling routes.</i>
ET2 To conserve soil resources and quality	IP172 + IP176 + IP214 + IP260 +	Medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	The sites are located in urban Ipswich, on brownfield land and are relatively small in size. The sites may potentially be contaminated due to previous use of the land and opportunities <i>could be sought to</i> <i>remediate the land before use</i> . Therefore it is considered that the effect in the medium and long term is likely to be beneficial.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET3 To reduce waste	IP172 0 IP176 0 IP214 0 IP260 -	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	Waste is likely to be generated in the short term during construction and in the long term during operation (household waste). The capacity of sites IP172, IP176, and IP214 is low in number of residential units, therefore on their own the development of the sites is not anticipated to result in any significant negative effects. IP260 is allocated for leisure uses and waste is likely to be generated in the long term as a result of increased number of visitors. Cumulative effects are likely to occur when the site allocations are considered along with other sites in IP-one area. <i>Opportunities should be sought to</i> <i>encourage recycling and reuse of</i> <i>materials.</i>
ET4 To reduce the effects of traffic upon the environment	IP172 + IP176 + IP214 + IP260 - / ?	Short, medium and long term Direct Irreversible Low	Central Ipswich	The sites are located in close proximity to the primary and secondary shopping areas (walking distance) and other town centre jobs and facilities, which may reduce the need to travel by private car. Users of the leisure facility (IP260) may increase traffic in the area but the level of certainty of prediction of potential effects is low. A transport assessment and travel plan will be required and will help to reduce any potential negative effects on traffic. The use of sustainable modes of transport should be encouraged through improvements of the pedestrian and cycling infrastructure in the area.
ET5 To improve access to key services for all sectors of the population	IP172 + IP176 + IP214 + IP260 +	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	All the sites are located in the central urban area of Ipswich and no major access constraints are identified with IP172, IP176, IP214, and IP260.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET6 To limit and adapt to climate change	IP172 0 IP176 0 IP214 0 IP260 0	Short, medium and long term Direct Irreversible Medium	The whole borough	The sites are not located in Flood Zones 2 and 3 and the risk of flooding of development is considered to be low. The traffic generated by IP172, IP176, and IP214 is likely to be low due to the size and the central location of the sites. Cumulatively, the effects on climate change can be reduced through encouraging people to use more sustainable modes of transport. The use of sustainable modes of transport should be encouraged through improvements of the pedestrian and cycling infrastructure in the area.
ET7 To protect and enhance the quality of water features and resources and reduce the risk of flooding	IP172 + IP176 + IP214 + IP260 +	Short, medium and long term Direct Irreversible Medium	Central Ipswich	The sites are not located in Flood Zones 2 and 3 and the risk of flooding of development is considered to be low. Benefits for water can be maximised if there are opportunities to remediate some historical areas of contamination. Surface water flooding local to site – will be considered at planning application stage for IP214 and IP260. See Appendix 1 of the Ipswich SFRA.
ET8 To conserve and enhance biodiversity and geodiversity, including favourable conditions on SSSIs, SPAs and SACs	IP172 0 IP176 0 IP214 0 IP260 0	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	There are no wildlife sites/TPOs near or adjacent to the site allocations. Therefore it is considered that the effects on biodiversity will be negligible.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET9 To conserve and enhance the historic environment, heritage assets and their settings	IP172 +/- IP176 0 IP214 +/- IP260 +/-	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	The development of IP176 will have neutral effect on areas of historical importance due to no heritage assets located in the vicinity of the site. Potential for any negative effects on heritage assets are identified with the development of sites IP172, IP214 and IP260 as there listed buildings located opposite or directly adjacent to the sites: 4 Soane Street (Grade II), Freemasons Hall (Grade II), Clarence House (Grade II), Milestone (Grade II), 35, St Margaret's Street (Grade II [*]), Olde Tudor Café (Grade II [*]) <i>Appropriate design of buildings should be considered (through the use of traditional or sympathetic building materials and techniques which respect those found on the surrounding buildings) to maintain the local distinctiveness of the area in order to avoid any negative effects on the listing buildings opposite or adjacent to new development. There is potential for archaeological remains at IP214 and IP172 as they are located in Area of Archaeological importance and conservation area.</i>
ET10 To conserve and enhance the quality and local distinctiveness of landscapes and townscapes	IP172 +/- IP176 0 IP214 +/- IP260 +/-	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	Potential negative effects on listed buildings are identified with the development of IP172, IP214 and IP260. The local townscape distinctiveness should be conserved through <i>the use of</i> <i>traditional or sympathetic building</i> <i>materials and techniques which respect</i> <i>those found on the listed building</i> .

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
HW1 To improve the health of those most in need	IP172 + IP176 + IP214 + IP260 +	Short, medium and long term Indirect Irreversible Medium	Localised area in Central Ipswich	Whilst the provision of decent housing associated with the development of sites IP172, IP176, and IP214 may have some positive indirect effects on health, it is considered that the overall effect will be negligible. However, positive indirect effects on human health are likely to occur due to the close proximity of Christchurch park which may encourage people to walk and participate in sport events. IP260 is allocated for leisure uses and during the operation of the site, potential indirect benefits could result from the use of indoor sport or leisure facilities that may be available.
HW2 To improve the quality of life where people live and encourage community participation	IP172 + IP176 + IP214 + IP260 +	Short, medium and long term Indirect Irreversible Medium	Central Ipswich	Community participation will be encouraged through the allocation of land for leisure at IP260. Sites IP172, IP176, and IP214 will contribute to the achievement of the SA objective through the provision of decent housing and thus improving the quality of life.
ER1 To reduce poverty and social exclusion	IP172 0 IP176 0 IP214 0 IP260 0	Short, medium and long term Indirect Irreversible Medium	N/A	There is no clear relationship between the allocation of the sites and the achievement of the SA objective and/or the effect will be negligible.
ER2 To offer everybody the opportunity for rewarding and satisfying employment	IP172 0 IP176 0 IP214 0 IP260 0	Short Direct Reversible Medium	Localised area in Central Ipswich	Some employment opportunities are likely to arise with the development of leisure facilities at IP260 and temporary construction jobs with each new development. However, on the whole, the predicted positive effects are unlikely to be significant.
ER3 To help meet the housing requirements for the whole community	IP172 + IP176 + IP214 + IP260 0	Short, medium and long term Direct Irreversible Medium	Central Ipswich	Site allocations IP172, IP176, and IP214 are likely to contribute to the achievement of the SA objective through the delivery of residential dwellings.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ER4 To achieve sustainable levels of prosperity and economic growth throughout the plan area	IP172 0 IP176 0 IP214 0 IP260 0	Short, medium and long term Indirect Irreversible Medium	N/A	Whilst new residential developments and a leisure facility will contribute to the achievement of the SA objective through improving the overall quality of life, it is considered that on its own each site will have a negligible effect.
ER5 To support vital and viable town, district and local centres	IP172 + IP176 + IP214 + IP260 +	Short, medium and long term Indirect Irreversible Medium	Central Ipswich	The development of the site will potentially provide support to the town centre shopping areas.
ER6 To encourage efficient patterns of movement in support of economic growth	IP172 + IP176 + IP214 + IP260 +	Short, medium and long term Indirect Irreversible Medium	Central Ipswich	The provision of residential dwellings (IP172 and IP176, IP214) and a leisure facility (IP260) close to existing employment and shopping areas will encourage efficient patterns of movement and in the long term this could help to support economic growth.
ER7 To encourage and accommodate both indigenous and inward investment	IP172 0 IP176 0 IP214 0 IP260 +	Short, medium and long term Indirect Irreversible Medium	Localised area in Central Ipswich	On its own, each site is unlikely to contribute to the achievement of the SA objective. Some local economic benefits are associated with the development of IP260.
CL1 To maintain and improve access to education and skills for both young people and adults	IP172 0 IP176 0 IP214 0 IP260 0	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	Whilst the close proximity of the sites to the University Campus Suffolk and Suffolk New College may result in educational benefits for the residents, it is considered that due to the small size of each site the effect will be negligible.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
CD1 To minimise potential opportunities for crime and anti-social activity	IP172 0 IP176 0 IP214 0 IP260 0	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	Whilst the size of the sites does not suggest any significant effect on the baseline conditions, it should be noted that the sites are located in an area ranked as most deprived. On the whole, crime levels are higher in Ipswich than national average. Therefore, <i>it would be beneficial</i> <i>to incorporate secured by design principles</i> <i>within new developments regardless of</i> <i>their size.</i> http://opendatacommunities.org/deprivatio n/map

Central – Alexandra

IP055 Crown Street – short stay car park IP253: Electric House, Lloyds Avenue, 13 homes IP245: 12-12a Arcade Street, 14 homes IP040: Civic Centre Area / Civic Drive, 80% retail, 20% residential (29 homes)

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET1 To improve air quality	IP253 0 IP245 0 IP040 - IP055 -	Medium and long term Direct and indirect Irreversible Medium	Central Ipswich	The sites (IP253, IP245, and IP040) are located in or near AQMA (along Crown Street and St. Margaret's Street), and potential negative effects are associated with dust and air pollution during construction The main use of the land (IP253, IP245) will be residential which may lead to additional use of private cars of any new residents. However, the sites are relatively small in size and the overall effect on traffic may be negligible due to the central location near most community facilities. The sites are well served by public transport therefore it is considered that effects will be neutral in the long term. IP040 has the potential to generate traffic from users of the retail services. Therefore, <i>the use of public transport should be encouraged through appropriate location of bus stops in the area.</i> IP055 is allocated for a short stay shopper car park. It will attract cars but it is north of the AQMA so could help to prevent cars from entering the AQMA to find parking elsewhere.
ET2 To conserve soil resources and quality	IP253 0 IP245 + IP040 + IP055 0	Medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	The sites are located in urban Ipswich, on brownfield land and are relatively small in size. IP245, IP040 may potentially be contaminated due to previous use of the land and opportunities <i>could be sought</i> <i>to remediate the land before use.</i> Therefore it is considered that the effect in the medium and long term is likely to be beneficial.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET3 To reduce waste	IP253 0 IP245 0 IP040 - IP055 0	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	Waste is likely to be generated in the short term during construction and in the long term during operation (household waste and waste from retail operations). The indicative capacity of sites IP245 and IP253 is low in number of residential units, therefore on their own the development of the sites is not anticipated to result in any significant negative effects. Cumulative effects are likely to occur when the site allocations are considered along with other sites in IP-one area. <i>Opportunities should be sought to</i> <i>encourage recycling and reuse of</i> <i>materials.</i> It is considered that IP055 will have a negligible effect on waste as the future uses of the site will be the same as the
ET4 To reduce the effects of traffic upon the environment	IP253 + IP245 + IP040 +/- IP055 -	Short, medium and long term Direct Irreversible Medium	Central Ipswich	existing use. IP253 is located within a speciality shopping area where community facilities and employment hubs are available at walking distance which may result in reduced need to travel by private car. The area is well served by public transport. The use of sustainable modes of transport should also be encouraged through improvements of the pedestrian and cycling infrastructure in the area. The availability of additional car parking spaces at IP055 may encourage people to travel by private car, particularly those travelling from adjacent districts into Ipswich town centre.
ET5 To improve access to key services for all sectors of the population	IP253 + IP245 + IP040 + IP055 +	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	The sites are located in a central urban area of Ipswich and will improve access to shopping areas, education facilities, etc. Access will be improved mainly for those who have access to a car and people travelling from outside Ipswich.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET6 To limit and adapt to climate change	IP253 0 IP245 0 IP040 0 IP055 -	Short, medium and long term Direct Irreversible Medium	The whole borough	None of the sites is located in Flood Zones 2 and 3 and the risk of flooding of development is considered to be rather low. The central location of IP253, IP245, and IP040 may encourage the use of more sustainable modes of transport, particularly due to the fact that community facilities are within walking distance. Therefore, cumulative positive effects may occur in the long term. Negative effects are likely to occur as a result of the allocation of site IP055 for car parking uses which may increase the use of private cars. <i>Sustainable modes of transport should be encouraged through improved pedestrian infrastructure and/or where appropriate the creation of cycle routes/lanes.</i>
ET7 To protect and enhance the quality of water features and resources and reduce the risk of flooding	IP253 + IP245 + IP040 + IP055 0	Short, medium and long term Direct Irreversible Medium	Central Ipswich	The sites are not located in Flood Zones 2 and 3 and the risk of flooding of development is considered to be low. Benefits for water quality improvements are available if opportunities to remediate some historical areas of contamination are used (IP245 and IP040). Surface water flooding local to site – will be considered at planning application stage for IP040 and IP245. See Appendix 1 of the Ipswich SFRA.
ET8 To conserve and enhance biodiversity and geodiversity, including favourable conditions on SSSIs, SPAs and SACs	IP253 - IP245 0 IP040 0 IP055 ?	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	There are no designated sites near or adjacent to any of the site allocations. Therefore it is considered that the effects on biodiversity will be negligible. TPOs are identified at IP055 and IP253 has TPO nearby and an application for Tree Works may be required (dependent on the design of the new car parking at IP055).

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET9 To conserve and enhance the historic environment, heritage assets and their settings	IP253 +/- IP245 +/- IP040 +/- IP055 +/-	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	Potential for any negative effects on heritage assets are identified with the development of site IP245 as there is a listed building located adjacent to the site (14&16 Arcade Street, Grade II). Appropriate design of building should be considered in order to avoid any negative effects on the setting of the designated site. There is potential for archaeological
				remains at the sites as they are located in an Area of Archaeological importance and three of them are near a conservation area (IP245, IP055 and IP040).
				There are no Scheduled monuments in the vicinity of the sites.
				Early evaluation would be advised so that decisions can be taken on preservation and/or appropriate investigation strategies designed.
				Mitigation measures may be required if any archaeological remains are discovered to avoid damage of heritage assets (where appropriate the completion of a licensed excavation and recording of remains before development commences).
ET10 To conserve and enhance the quality and local distinctiveness of landscapes and townscapes	IP253 0 IP245 - IP040 0 IP055 -	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	Potential negative effects on listed buildings are identified only with the development of IP245. The site is also located in a Conservation area. IP055 is adjacent to a Conservation area. The local townscape distinctiveness should be conserved through <i>the use of</i> <i>traditional or sympathetic building</i> <i>materials and techniques which respect</i> <i>those found on the listed building and</i> <i>maintain the local distinctiveness of the</i> <i>area.</i>

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
HW1 To improve the health of those most in need	IP253 0 IP245 0 IP040 0 IP055 0	Short, medium and long term Indirect Irreversible Medium	Localised area in Central Ipswich	Whilst the provision of decent housing associated with the sites may have some positive indirect effects on health, it is considered that the overall effect will be negligible. No significant impacts on health are identified with the allocation of IP055 for car parking.
HW2 To improve the quality of life where people live and encourage community participation	IP253 + IP245 + IP040 + IP055 +	Short, medium and long term Indirect Irreversible Medium	Central Ipswich	Community participation will be encouraged through the allocation of sites in the central urban area near a wide range of facilities. The sites will also contribute to the achievement of the SA objective through the provision of decent housing thus improving the quality of life.
ER1 To reduce poverty and social exclusion	IP253 0 IP245 0 IP040 0 IP055 0	Short, medium and long term Indirect Irreversible Medium	N/A	There is no clear relationship between the allocation of the housing sites and the achievement of the SA objective and/or the effect will be negligible. No significant impacts on poverty are identified with the allocation of IP055 for car parking.
ER2 To offer everybody the opportunity for rewarding and satisfying employment	IP253 0 IP245 0 IP040 + IP055 0	Short, medium Direct Reversible Medium	Localised area in Central Ipswich	Some employment opportunities are likely to arise with the retail development of IP040 and temporary construction jobs with each new development. However, on the whole, the predicted positive effects are unlikely to be significant.
ER3 To help meet the housing requirements for the whole community	IP253 + IP245 + IP040 + IP055 0	Short, medium and long term Direct Irreversible Medium	Central Ipswich	IP253, IP245, and IP040 are likely to contribute to the achievement of the SA objective through the delivery of residential dwellings. No effects are recorded with the allocation of IP055 as the site will be used for car parking.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ER4 To achieve sustainable levels of prosperity and economic growth throughout the plan area	IP253 0 IP245 0 IP040 0 IP055 +	Short, medium and long term Indirect Irreversible Medium	N/A	Whilst new residential developments and a retail centre will contribute to the achievement of the SA objective through improving the overall quality of life, it is considered that on its own each of the housing sites will have a negligible effect. IP055 is allocated for a short stay car park for shoppers. Its purpose is to support town centre shops and leisure facilities, recognising that not everyone who may want to use town centre facilities has access to regular public transport, especially in rural areas outside lpswich. Therefore it is considered that IP055 may have some indirect positive effects on economic growth through improved access to town centre facilities.
ER5 To support vital and viable town, district and local centres	IP253 0 IP245 0 IP040 + IP055 +	Short, medium and long term Indirect Irreversible Medium	Localised area in Central Ipswich	The development of IP040 will contribute to the achievement of the SA objective through opportunity to offer additional retail services in the central urban Ipswich. IP055 may offer direct benefits through improved access to town centre facilities. There is no clear relationship between the allocation of the rest of the sites and the achievement of the SA objective and/or the effect will be negligible.
ER6 To encourage efficient patterns of movement in support of economic growth	IP253 + IP245 + IP040 + IP055 +	Short, medium and long term Indirect/Direct Irreversible Medium	Central Ipswich	The provision of residential dwellings close to existing employment and shopping areas will encourage efficient patterns of movement and in the long term this could help to support economic growth. IP040is at the edge of the existing central shopping area thus adding to an existing attraction in a highly accessible location rather than creating a rival attraction out of town. IP055 may offer direct benefits through improved access to town centre facilities, especially to people visiting from rural areas outside Ipswich.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ER7 To encourage and accommodate both indigenous and inward investment	IP253 0 IP245 0 IP040 + IP055 0	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	On its own, each site is unlikely to contribute to the achievement of the SA objective. Some local economic benefits are associated with the development of IP040 for retail use.
CL1 To maintain and improve access to education and skills for both young people and adults	IP253 0 IP245 0 IP040 0 IP055 0	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	Whilst the close proximity of the sites to the University Campus Suffolk and Suffolk New College may result in educational benefits for the residents, it is considered that due to the small size of each site the effect will be negligible. There is no relationship between the allocation of IP055 for car parking and educational attainment.
CD1 To minimise potential opportunities for crime and anti-social activity	IP253 0 IP245 0 IP040 0 IP055 0	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	Whilst the size of the sites does not suggest any significant effect on the baseline conditions, it should be noted that the sites are located in an area ranked as most deprived. On the whole, crime levels are higher in Ipswich than national average. Therefore, <i>it would be</i> <i>beneficial to incorporate secured by</i> <i>design principles within new</i> <i>developments regardless of their size.</i> <u>http://opendatacommunities.org/deprivati</u> <u>on/map</u>

Group 17 – Central – Alexandra

IP048: Mint Quarter / Cox Lane – open space, short stay parking, 72 homes IP074: Church and land at Upper Orwell Street, 12 homes IP264: 28-32 Tacket Street, 16 homes

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET1 To improve air quality	IP048 - IP074 - IP264 -	Medium and long term Direct and indirect Irreversible Medium	Central Ipswich	IP048 and IP074 are located in or close to AQMA (in the area of St. Helen's Street), and potential negative effects are associated with dust and air pollution during construction. IP264 is unlikely to have any impact on AQMA due to its distance and size.
				The main use of the land (IP048, IP074, and IP264) will be residential which may lead to additional use of private cars by new residents. Although the sites are well served by public transport and near two bus stations, the number of new residents suggests potential for increased traffic in the town centre.
				Therefore, sustainable modes of transport should be encouraged through improved pedestrian routes in the area.
ET2 To conserve soil resources and quality	IP048 + IP074 + IP264 +	Medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	The sites are located in urban Ipswich, on brownfield land and are likely to be contaminated due to previous use of the land and opportunities <i>could be sought</i> <i>to remediate the land before use.</i> Therefore it is considered that the effect in the medium and long term is likely to be beneficial.
Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
---	--	---	--------------------------------------	---
ET3 To reduce waste	IP048 - IP074 0 IP264 0	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	Waste is likely to be generated in the short term during construction and in the long term during operation (household waste). The indicative capacity of IP048 suggests potential for negative effects if mitigation measures are not in place. Cumulatively, negative effects are likely to occur when the site allocations are considered along with other sites in IP- one area. <i>Opportunities should be sought to</i> <i>encourage recycling and reuse of</i> <i>materials.</i>
ET4 To reduce the effects of traffic upon the environment	IP048 + IP074 + IP264 +	Short, medium and long term Direct Irreversible Medium	Central Ipswich	The central location of the sites may result in reduced need to travel by private car. The area is well served by public transport. Tower Ramparts Bus Station and Old Cattle Market Bus Station and Ipswich Railway Station are located at a walking distance from the sites. Therefore, it is considered that the site allocation is likely to contribute to the achievement of the SA objective.
ET5 To improve access to key services for all sectors of the population	IP048 + IP074 + IP264 +	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	All the sites are located in the central urban area of Ipswich near shopping and community facilities and no major access constraints are identified.
ET6 To limit and adapt to climate change	IP048 + IP074 + IP264 +	Short, medium and long term Direct Irreversible Medium	The whole borough	None of the sites are located in Flood Zones 2 and 3 and the risk of flooding of development is considered to be rather low. The central location of the sites may encourage the use of more sustainable modes of transport, particularly due to the fact that community facilities are within walking distance. Cumulatively, the effects from climate change can be reduced through encouraging people to use more sustainable modes of transport, e.g. walking at short distances.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET7 To protect and enhance the quality of water features and resources and reduce the risk of flooding	IP048 + IP074 + IP264 +	Short, medium and long term Direct Irreversible Medium	Central Ipswich	The sites are not located in Flood Zones 2 and 3 and the risk of flooding of development is considered to be low. Benefits for water quality can be maximised if opportunities are sought to remediate the contaminated land where appropriate. Surface water flooding local to site – will be considered at planning application stage for IP048. See Appendix 1 of the Ipswich SFRA.
ET8 To conserve and enhance biodiversity and geodiversity, including favourable conditions on SSSIs, SPAs and SACs	IP048 - IP074 - IP264 0	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	There are no biodiversity/geodiversity designated sites within or near the site allocations. TPOs are identified at IP048 and IP074 and an <i>application for Tree</i> <i>Works may be required</i> .

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET9 To conserve and enhance the historic environment, heritage assets and their settings	IP048 - IP074 - IP264 -	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	 IP264 is located in a conservation areas and IP048 is adjacent to it. Potential for any negative effects on sites of historical importance are identified with the development of these three sites as there are listed buildings located adjacent to the sites: Church of St. Pancras (Grade II) T0-74 Upper Orwell Street (Grade II) T0-74 Upper Orwell Street (Grade II) County Hall (Grade II) Toury Hall (Grade II) County Hall (Grade II) County Hall (Grade II) The locally listed façade to Carr Street will be retained. In addition, IP048 is also adjacent to Central Conservation Area and Grade II listed Christ Church is located to the south. Appropriate design of building should be considered in order to avoid any negative effects on the setting of designated site. There is also potential for any archaeological remains at the sites as they are located in Area of Archaeological importance. The Area between Upper Brook Street and Upper Orwell Street is a Scheduled Monument - middle and late Saxon town. Any proposals would need to consider the impact of development upon designated and non-designated heritage assets and their setting, including any resulting benefit, harm or loss to their significance. Due to the high potential for archaeological remains of national significance, detailed early preapplication discussions with Suffolk County Council Archaeological Service and Historic England will be required in order to agree the principle of development and inform designs (e.g. to allow for preservation in-situ of deposits

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
				or appropriate programmes of work) this will help to avoid potential negative impacts. Mitigation measures may be required if any archaeological remains are discovered to avoid damage of heritage assets. Where appropriate the mitigation measures can include completion of a licensed excavation and recording of remains before development commences.
ET10 To conserve and enhance the quality and local distinctiveness of landscapes and townscapes	IP048 - IP074 - IP264 -	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	Potential negative effects on listed buildings are identified with the development of the sites. Sites IP048 and IP264 are located respectively next to and in a Conservation area and the local townscape distinctiveness should be conserved through <i>the use of</i> <i>traditional or sympathetic building</i> <i>materials and techniques which respect</i> <i>those found on the surrounding buildings</i> <i>to maintain the local distinctiveness of</i> <i>the area.</i>

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
HW1 To improve the health of those most in need	IP048 + IP074 + IP264 +	Short, medium and long term Indirect Irreversible Medium	Localised area in Central Ipswich	Whilst the provision of decent housing associated with the development of the sites may have some positive indirect effects on health, it is considered that the overall effect will be negligible. However, positive indirect effects are likely to occur with regards to the potential use of Christchurch park and Alexandra Park located near the sites. The creation of new public open space at IP048 will also have an indirect positive effect on health in a localised area.
HW2 To improve the quality of life where people live and encourage community participation	IP048 + IP074 + IP264 +	Short, medium and long term Indirect Reversible Medium	Central Ipswich	Community participation will be encouraged through the allocation of sites in the central urban area near a wide range of facilities. The sites will also contribute to the achievement of the SA objective through the provision of decent housing and open space and thus improving the quality of life.
ER1 To reduce poverty and social exclusion	IP048 + IP074 + IP264 +	Short, medium and long term Indirect Reversible Medium	Central Ipswich	Social exclusion will be reduced through the development of residential dwellings near a wide range of community facilities easily accessible by different means of transport.
ER2 To offer everybody the opportunity for rewarding and satisfying employment	IP048 0 IP074 0 IP264 0	Short, medium Direct Reversible Medium	Localised area in Central Ipswich	The allocation of sites detracts from the SA objective as the land use is primarily residential and any jobs during construction will only be temporary.
ER3 To help meet the housing requirements for the whole community	IP048 + + IP074 + IP264 +	Short, medium and long term Direct Irreversible Medium	Central Ipswich	IP048, IP074, and IP264 are likely to contribute to the achievement of the SA objective through the delivery of residential dwellings.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ER4 To achieve sustainable levels of prosperity and economic growth throughout the plan area	IP048 0 IP074 0 IP264 0	Short, medium and long term Indirect Irreversible Medium	N/A	Whilst new residential developments will contribute to the achievement of the SA objective through improved quality of life, it is considered that on its own each site will have a negligible effect on the economic growth of the borough.
ER5 To support vital and viable town, district and local centres	IP048 + IP074 + IP264 +	Short, medium and long term Indirect/Direct Irreversible Low	Localised area in Central Ipswich	Positive effects are identified with regard to the shopping areas located in the town centre, which may benefit from the increase of new residents in the area. Short stay car parking for shoppers (element of IP048) will replace and add to existing surface car parking and directly support the SA objective.
ER6 To encourage efficient patterns of movement in support of economic growth	IP048 + IP074 + IP264 +	Short, medium and long term Indirect/Direct Irreversible Medium	Central Ipswich	The provision of residential dwellings close to existing employment and shopping areas will encourage efficient patterns of movement and in the long term this could help to support economic growth. Short stay car parking for shoppers (element of IP048) will replace and add to existing surface car parking and directly support the SA objective.
ER7 To encourage and accommodate both indigenous and inward investment	IP048 + IP074 + IP264 +	Short, medium and long term Direct Irreversible Low	Localised area in Central Ipswich	On its own, each site is unlikely to contribute to the achievement of the SA objective. Cumulatively, the sites will contribute to the regeneration / redevelopment of central town areas, with some infrastructure improvements (car parking) at IP048, the provision of new public open space which may increase the attractiveness of the area.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
CL1 To maintain and improve access to education and skills for both young people and adults	IP048 0 IP074 0 IP264 0	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	Whilst the close proximity of the sites to the University Campus Suffolk and Suffolk New College may result in educational benefits for the residents, it is considered that due to the relatively small size of each site the effect will be negligible.
CD1 To minimise potential opportunities for crime and anti-social activity	IP048 0 IP074 0 IP264 0	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	Whilst the size of the sites does not suggest any significant effect on the baseline conditions, it should be noted that the sites are located in an area ranked as most deprived. On the whole, crime levels are higher in Ipswich than national average. Therefore, <i>it would be</i> <i>beneficial to incorporate secured by</i> <i>design principles within new</i> <i>developments regardless of their size</i> . http://opendatacommunities.org/deprivatii on/map

1.

Central – Alexandra

IP052: Land between Lower Orwell Street, 29 homes + employment use IP011a: Smart Street, Foundation Street (former Gym and Trim), 14 homes IP011b: Smart Street / Foundation Street, 50 homes + employment use IP054: Land between Old Cattle Market and Star Lane, 28 homes + 70% for office/leisure use, retail use, short stay car parking

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET1 To improve air quality	IP052 - IP011a 0 IP011b - IP054 -	Medium and long term Direct and indirect Irreversible Medium	Central Ipswich	The sites are located in or close to AQMA Sites IP052 and IP054 are allocated for mixed use developments and may generate traffic on Star Lane and key town centre roads. Although the sites are well served by public transport and near two bus stations, the number of new residents suggests potential for congestion at key junctions in the town centre. Therefore, sustainable modes of transport should be encouraged through improved pedestrian routes in the area.
ET2 To conserve soil resources and quality	IP052 + IP011a + IP011b + IP054 +	Medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	The sites are located in urban Ipswich, on brownfield land and are likely to be contaminated due to previous use of the land and opportunities <i>could be sought</i> <i>to remediate the land before use.</i> Therefore it is considered that the effect in the medium and long term is likely to be beneficial.
ET3 To reduce waste	IP052 0 IP011a 0 IP011b - IP054 -	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	Waste is likely to be generated in the short term during construction and in the long term during operation (household waste). The indicative capacity of sites IP011b and IP054 suggest potential for negative effects if mitigation measures are not in place. Cumulatively, negative effects are likely to occur when the site allocations are considered along with other sites in IP- one area. <i>Opportunities should be sought to</i> <i>encourage recycling and reuse of</i>

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET4 To reduce the effects of traffic upon the environment	IP052 + IP011a + IP011b + IP054 +	Short, medium and long term Direct Irreversible Medium	Central Ipswich	The central location of the sites may result in reduced need to travel by private car. The area is well served by public transport. Tower Ramparts Bus Station and Old Cattle Market Bus Station and Ipswich Railway Station are located at a walking distance from the sites. In addition, the main shopping area is adjacent to IP011a, IP052 and IP054 and land is allocated to the south of the sites (IP035) for employment use. For all these reasons it is considered that the sites will contribute to the achievement of the SA objective. A transport assessment and travel plan will be required for IP011b and IP054. This will help to avoid potential negative impacts on traffic.
ET5 To improve access to key services for all sectors of the population	IP052 + IP011a + IP011b + IP054 +	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	All the sites are located in the central urban area of Ipswich near shopping areas and community facilities and no major access constraints are identified.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET6 To limit and adapt to climate change	IP052 +/- IP011a +/- IP011b +/- IP054 +/-	Short, medium and long term Direct Irreversible Medium	The whole borough	Parts of the sites are located in Flood Zones 2 and 3 with a risk of flooding of the southern parts of new development. <i>Flood risk assessment may be required</i> <i>at project assessment level to identify</i> <i>appropriate mitigation measures for the</i> <i>parts of the development that fall within</i> <i>Flood Zone 2 and 3 (e.g. through</i> <i>design). New developments should be</i> <i>encouraged to use SuDS to manage</i> <i>runoff, reduce further flood risk and</i> <i>protect water quality.</i> The central location of the sites may encourage the use of more sustainable modes of transport, particularly due to the fact that community facilities are within walking distance. Cumulatively, the effects from climate change can be reduced through encouraging people to use more sustainable modes of transport.
ET7 To protect and enhance the quality of water features and resources and reduce the risk of flooding	IP052 +/- IP011a +/- IP011b +/- IP054 +/-	Short, medium and long term Direct Irreversible Medium	Central Ipswich	Parts of the sites are located in Flood Zones 2 and 3 and there is a risk of flooding of the southern parts of new development. In the medium and long term it is anticipated that the risk will be alleviated through the delivery of Ipswich Flood Defence. <i>However, flood risk</i> <i>assessment may be required at project</i> <i>assessment level to identify appropriate</i> <i>mitigation measures for the parts of the</i> <i>development that fall within Flood Zone 2</i> <i>and 3 (e.g. through design).</i> Water quality could be enhanced if opportunities are sought to remediate the contaminated land where appropriate. Surface water flooding local to site – will be considered at planning application stage for IP011b and IP054. See Appendix 1 of the Ipswich SFRA.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET8 To conserve and enhance biodiversity and geodiversity, including favourable conditions on SSSIs, SPAs and SACs	IP052 0 IP011a - IP011b - IP054 -	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	There are no biodiversity/geodiversity designated sites within or near the site allocations. TPO nearby is identified in IP011a, IP011b and IP054 and an <i>application for Tree Works may be</i> <i>required.</i>
ET9 To conserve and enhance the historic environment, heritage assets and their settings	IP052 - IP011a - IP011b - IP054 -	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	All the sites are adjacent to a conservation area and part of IP054 is located within a Conservation area. Potential for any negative effects on sites of historical importance are identified with the development of IP052 and IP054 as there are listed buildings located adjacent to the sites (32 Lower Brook Street – Grade II, 24 Fore Street – Grade II*, 26-28 Fore Street Grade II, Conservative Office – Grade II). Also IP011a and b because they adjoin the listed Tooley's Court. IP054 is also located partly within Central Conservation Area and contains two Grade II listed buildings (18-20 Lower Brook Street). The site is flanked by the Central conservation area and several listed buildings, with two Grade II* churches to the south. Scheduled monuments in the vicinity of the sites include Dominican Friary (remains of); area of middle and late Saxon town, off Foundation Street; area of middle and late Saxon and medieval defences, off Shire Hall Yard; area of middle and late Saxon town, off Star Lane. <i>Appropriate design of building should be considered in order to avoid any negative effects on the setting of designated site.</i> There is also potential for any archaeological remains at all four sites as they are located in an Area of Archaeological importance. Therefore mitigation measures may be required if any archaeological remains

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
				are discovered to avoid damage of heritage assets (<i>where appropriate the</i> <i>completion of a licensed excavation and</i> <i>recording of remains before development</i> <i>commences</i>). Detailed pre-application discussions with Suffolk County Council Archaeological Service and Historic England will be required in order to agree the principle of development and inform design (e.g. to allow preservation in-situ of deposits or appropriate programmes of work) this will help to avoid potential negative effects.
ET10 To conserve and enhance the quality and local distinctiveness of landscapes and townscapes	IP052 - IP011a - IP011b - IP054 -	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	Potential negative effects on listed buildings are identified with the development of IP011a and b, IP052 and IP054. IP052 and IP011a are adjacent to a Conservation area and IP054 is located part within a Conservation area. The townscape distinctiveness should be conserved through <i>the use of traditional</i> <i>or sympathetic building materials and</i> <i>techniques which respect those found on</i> <i>the surrounding buildings to maintain the</i> <i>local distinctiveness of the area.</i>

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
HW1 To improve the health of those most in need	IP052 + IP011a + IP011b + IP054 +	Short, medium and long term Indirect Irreversible Medium	Localised area in Central Ipswich	Whilst the provision of decent housing associated with the development of sites all four sites may have some positive indirect effects on health, it is considered that the overall effect will be negligible. However, positive indirect effects are likely to occur with regards to the potential use of Christchurch park Alexandra Park located near the sites. The sites are also near designated river paths and close to community facilities (restaurants, cinemas, churches) which may encourage more people to choose walking as the main mode of transport.
HW2 To improve the quality of life where people live and encourage community participation	IP052 + + IP011a + IP011b + IP054 + +	Short, medium and long term Direct / Indirect Reversible Medium	Central Ipswich	Community participation will be encouraged through the allocation of sites in the central urban area near a wide range of facilities (e.g. football club, university campus, etc.). The sites will also contribute to the achievement of the SA objective through the provision of decent housing and some employment opportunities (mixed use developments IP052 and IP054) thus improving the quality of life.
ER1 To reduce poverty and social exclusion	IP052 + IP011a + IP011b + IP054 +	Short, medium and long term Direct / Indirect Reversible Medium	Central Ipswich	Social exclusion will be reduced through the allocation of sites for residential/mixed use near a wide range of community facilities easily accessible by different means of transport.
ER2 To offer everybody the opportunity for rewarding and satisfying employment	IP052 + IP011a 0 IP011b + IP054 + +	Short, medium Direct Reversible Medium	Localised area in Central Ipswich	The allocation of IP052, IP011b and IP054 will contribute to the SA objective as part of the sites will be for business/employment use.
ER3 To help meet the housing requirements for the whole community	IP052 + IP011a + IP011b ++ IP054 +	Short, medium and long term Direct Irreversible Medium	Central Ipswich	IP052, IP011a and b and IP054 will contribute directly to the achievement of the SA objective through the delivery of residential dwellings.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ER4 To achieve sustainable levels of prosperity and economic growth throughout the plan area	IP052 + IP011a 0 IP011b + IP054 +	Short, medium and long term Direct/Indirect Irreversible Medium	Central Ipswich	The SA objective will be achieved through the delivery of decent housing, employment opportunities and opportunities for inward investment with the development of IP052, IP011b, and IP054.
ER5 To support vital and viable town, district and local centres	IP052 + IP011a 0 IP011b + IP054 +	Short, medium and long term Direct/Indirect Irreversible Medium	Central Ipswich	IP052, IP011b, and IP054 would support new business formation through the allocation of land for business/employment use and thus increasing the number of new jobs. IP054 also supports the SA Objective though the delivery of car parking which may improve access to essential services and facilities for both existing and new residents.
ER6 To encourage efficient patterns of movement in support of economic growth	IP052 + IP011a + IP011b + IP054 +	Short, medium and long term Direct/Indirect Irreversible Medium	Central Ipswich	The provision of residential/mixed use dwellings and student accommodation close to existing/future employment and shopping areas will encourage efficient patterns of movement and in the long term this could help to support economic growth.
ER7 To encourage and accommodate both indigenous and inward investment	IP052 0 IP011a 0 IP011b 0 IP054 ++	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	IP054 strongly supports the achievement of the SA objective through the allocation of land for various uses including office, leisure and retail. On its own, IP052, IP011a and IP011b are unlikely to contribute to the achievement of the SA objective due to their size and purpose. Cumulatively, the sites will contribute to the regeneration / redevelopment of central town areas, with some infrastructure improvements (car parking) at IP0054, the replacement of an electricity substation which may increase the business attractiveness of the area.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
CL1 To maintain and improve access to education and skills for both young people and adults	IP052 0 IP011a 0 IP011b 0 IP054 0	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	Whilst the close proximity of the sites to the University Campus Suffolk and Suffolk New College may result in educational benefits for the residents, it is considered that due to the relatively low indicative capacity of each site the effect will be negligible.
CD1 To minimise potential opportunities for crime and anti-social activity	IP052 0 IP011a 0 IP011b 0 IP054 0	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	Whilst the size of the sites does not suggest any significant effect on the baseline conditions, it should be noted that the sites are located in an area ranked as most deprived. On the whole, crime levels are higher in Ipswich than national average. Therefore, <i>it would be</i> <i>beneficial to incorporate secured by</i> <i>design principles within new</i> <i>developments regardless of their size</i> . <u>http://opendatacommunities.org/deprivati</u> <u>on/map</u>

2.

Central – Alexandra

IP089: Waterworks Street, 23 homes

IP012: Peter's Ice Cream, 29 homes IP258: Land at University Campus Suffolk as part of the Education Quarter, new primary school IP043: Commercial Buildings and Jewish Burial Ground, Star Lane, 50 homes + employment use

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET1 To improve air quality	IP089 - IP012 - IP258 - IP043 -	Medium and long term Direct and indirect Irreversible Medium	Central Ipswich	IP089, IP012, IP258 and IP043 are located near AQMA. As a result of the increase of new residents traffic may be generated in the vicinity of Fore St, Star Lane and Grimwade Street. Although the sites are well served by public transport and near two bus stations, the number of new residents suggests potential for congestion at key junctions in the town centre. Therefore, sustainable modes of transport should be encouraged through improved pedestrian routes in the area.
ET2 To conserve soil resources and quality	IP089 + IP012 + IP258 + IP043 +	Medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	The sites are located in urban Ipswich, on brownfield land and are likely to be contaminated due to previous use of the land and opportunities <i>could be sought</i> <i>to remediate the land before use.</i> Therefore it is considered that the effect in the medium and long term is likely to be beneficial.
ET3 To reduce waste	IP089 0 IP012 0 IP258 0 IP043 -	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	Waste is likely to be generated in the short term during construction and in the long term during operation (household waste). The indicative capacity of IP043 suggests potential for negative effects if mitigation measures are not in place. Cumulatively, negative effects are likely to occur when the site allocations are considered along with other sites in IP- one area. <i>Opportunities should be sought to</i> <i>encourage recycling and reuse of</i> <i>materials.</i>

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET4 To reduce the effects of traffic upon the environment	IP089 + IP012 + IP258 + IP043 +	Short, medium and long term Direct Irreversible Medium	Central Ipswich	The central location of the sites may result in reduced need to travel by private car. The area is well served by public transport. Tower Ramparts Bus Station and Old Cattle Market Bus Station and Ipswich Railway Station are located at a walking distance from the sites. In addition, the IP089 and IP043 are adjacent to speciality shopping area and IP012 is near existing Local Centre. The provision of school at IP258 would provide an opportunity for children living around the Waterfront to walk to school. For all these reasons, it is considered that the sites are likely to contribute to the achievement of the SA objective. A transport assessment and travel plan will be required for IP043. This will help to avoid potential negative effects on
ET5 To improve access to key services for all sectors of the population	IP089 + IP012 +/- IP258 + IP043 +/-	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	traffic. All the sites are located in the central urban area of Ipswich near shopping areas and community facilities. Access constraints are identified at IP012 and IP043 associated with potential negative effects on the local highway network (A1156 and Star Lane). Sustainable modes of transport should be encouraged through improved pedestrian infrastructure and/or where appropriate the creation of cycle routes/lanes particularly around the vicinity of the university buildings.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET6 To limit and adapt to climate change	IP089 0 IP012 0 IP258 0 IP043 -	Short, medium and long term Direct Irreversible Medium	Central Ipswich	IP043 is located in Flood Zones 2 and 3 with a risk of flooding of the southern parts of new development. Flood risk assessment may be required at project assessment level to identify appropriate mitigation measures for the parts of the development that fall within Flood Zone 2 and 3 (e.g. through design). New development should be encouraged to use SuDS to manage runoff, reduce further flood risk and protect water quality. The central location of the sites may encourage the use of more sustainable modes of transport, particularly due to the fact that most community facilities are within walking distance.
ET7 To protect and enhance the quality of water features and resources and reduce the risk of flooding	IP089 0 IP012 0 IP258 0 IP043 -	Short, medium and long term Direct Irreversible Medium	Central Ipswich	 IP043 is located in Flood Zones 2 and 3 and there is a risk of flooding of the southern parts of new development. In the medium and long term it is anticipated that the risk will be alleviated through the delivery of Ipswich Flood Defence. Surface water flooding local to site – will be considered at planning application stage for IP012, IP089, IP258 and IP043. See Appendix 1 of the Ipswich SFRA. Flood risk assessment may be required at project assessment level to identify appropriate mitigation measures for the parts of the development that falls within Flood Zone 2 and 3 (e.g. through design). Water quality could be enhanced if opportunities are sought to remediate the contaminated land where appropriate.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET8 To conserve and enhance biodiversity and geodiversity, including favourable conditions on SSSIs, SPAs and SACs	IP089 - IP012 0 IP258 - IP043 0	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	There are no biodiversity/geodiversity designated sites within or near the site allocations. TPO nearby is identified in IP089 and IP258 and an <i>application for</i> <i>Tree Works may be required.</i>
ET9 To conserve and enhance the historic environment, heritage assets and their settings	IP089 - IP012 0 IP258 0 IP043 -	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	Listing buildings are adjacent or near IP089 and IP043 with potential for negative effects from new developments: 44 Fore Street (Grade II) 54-58 Fore Street (Grade II*) Walls Enclosing Jews' Burial Ground (Grade II) The Lord Nelson Inn (Grade II) 13-25 Waterworks Street (Grade II) These sites also fall within the boundary of a Conservation area. Grade II* Old Custom House is also located to the south west of IP043 and a complex of highly graded buildings at Isaac Lord to the south east. In addition, IP258 is adjacent to Central Conservation Area and Grade II listed Holy Trinity Church to the south. It is noted that IP043 presents opportunities for enhancing the Jewish burial ground which would need to be carefully respected by ant development proposal. <i>Appropriate design of building should be</i> <i>considered in order to avoid any</i> <i>negative effects on the setting of</i> <i>designated sites.</i> There is also potential for archaeological remains at the sites as they are located in an Area of Archaeological importance. Early evaluation would be advised so that decisions can be taken on preservation and/or appropriate investigation strategies designed. Mitigation measures may be required if any archaeological remains are

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
				discovered to avoid damage of heritage assets (where appropriate the completion of a licensed excavation and recording of remains before development commences). Detailed early pre-application discussions with Suffolk County Council Archaeological Service and Historic England will be required in order to agree the scope of required assessment, the principle of development and to inform design. This will help to avoid potential negative effects.
ET10 To conserve and enhance the quality and local distinctiveness of landscapes and townscapes	IP089 - IP012 0 IP258 0 IP043 -	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	Potential negative effects on listed buildings and local distinctiveness of the conservation area are identified with the development of IP089 and IP043. The townscape distinctiveness should be conserved through <i>the use of traditional</i> <i>or sympathetic building materials and</i> <i>techniques which respect those found on</i> <i>the surrounding buildings to maintain the</i> <i>local distinctiveness of the area.</i>

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
HW1 To improve the health of those most in need	IP089 + IP012 + IP258 + IP043 +	Short, medium and long term Indirect Irreversible Medium	Localised area in Central Ipswich	Whilst the provision of decent housing associated with the development of the sites may have some positive indirect effects on health, it is considered that the overall effect will be negligible. However, positive indirect effects are likely to occur with regards to the potential use of Christchurch park and Alexandra Park located near the sites. The sites are also near designated river paths and close to community facilities (restaurants, cinemas, churches) which may encourage more people to choose walking as the main mode of transport.
HW2 To improve the quality of life where people live and encourage community participation	IP089 + IP012 + IP258 + IP043 +	Short, medium and long term Direct / Indirect Reversible Medium	Central Ipswich	Community participation will be encouraged through the allocation of sites in the central urban area near a wide range of facilities (e.g. parks, public transport infrastructure, university campus, etc.). The sites will also contribute to the achievement of the SA objective through the provision of decent housing and a new primary school (IP258) thus improving the overall quality of life.
ER1 To reduce poverty and social exclusion	IP089 + IP012 + IP258 + IP043 +	Short, medium and long term Direct / Indirect Reversible Medium	Central Ipswich	Social exclusion will be reduced through the allocation of sites for residential/education uses near a wide range of community facilities easily accessible by different means of transport.
ER2 To offer everybody the opportunity for rewarding and satisfying employment	IP089 0 IP012 0 IP258 0 IP043 0	Short, medium Direct Reversible Medium	Localised area in Central Ipswich	The only site with potential to contribute to the achievement of the SA objective is IP043 where 20% of the land is allocated for B1 leisure use. However, the size of the site suggests a negligible change from the baseline.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ER3 To help meet the housing requirements for the whole community	IP089 + IP012 + IP258 0 IP043 +	Short, medium and long term Direct Irreversible Medium	Central Ipswich	IP089, IP012, IP043 will contribute directly to the achievement of the SA objective through the delivery of residential dwellings.
ER4 To achieve sustainable levels of prosperity and economic growth throughout the plan area	IP089 0 IP012 0 IP258 0 IP043 0	Short, medium and long term Direct/Indirect Irreversible Medium	N/A	Whilst new residential developments and a leisure facility will contribute to the achievement of the SA objective through improving the overall quality of life, it is considered that on its own each site will have a negligible effect.
ER5 To support vital and viable town, district and local centres	IP089 + IP012 + IP258 + IP043 +	Short, medium and long term Direct/Indirect Irreversible Medium	N/A	All the sites are within the town centre and will directly support the SA objective. IP012 is located near a Local centre (45) with a potential to attract new residents and contribute directly to the achievement of the SA objective.
ER6 To encourage efficient patterns of movement in support of economic growth	IP089 + IP012 + IP258 + IP043 +	Short, medium and long term Direct/Indirect Irreversible Medium	Central Ipswich	The provision of residential/mixed use dwellings and student accommodation close to existing/future employment and shopping areas will encourage efficient patterns of movement and in the long term this could help to support economic growth.
ER7 To encourage and accommodate both indigenous and inward investment	IP089 0 IP012 0 IP258 0 IP043 0	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	On its own, each site is unlikely to contribute to the achievement of the SA objective. Cumulatively, the sites will contribute to the regeneration / redevelopment of central town areas, through the provision of new primary school (IP258), new decent housing (IP089, IP012, and IP043) and improve the business attractiveness of the area.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
CL1 To maintain and improve access to education and skills for both young people and adults	IP089 + IP012 + IP258 ++ IP043 +	Short, medium and long term Direct Reversible Medium	Localised area in Central Ipswich	Potential positive effects are associated with the close proximity of the sites to University Campus Suffolk which may contribute to the achievement of the SA objective. In addition, IP258 is allocated for new primary school which will have a direct positive effect on education in the area.
CD1 To minimise potential opportunities for crime and anti-social activity	IP089 0 IP012 0 IP258 0 IP043 0	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	Whilst the size of the sites does not suggest any significant effect on the baseline conditions, it should be noted that the sites are located in an area ranked as most deprived. On the whole, crime levels are higher in Ipswich than national average. Therefore, <i>it would be</i> <i>beneficial to incorporate secured by</i> <i>design principles within new</i> <i>developments regardless of their size</i> . <u>http://opendatacommunities.org/deprivati</u> <u>on/map</u>

Central – Alexandra

IP035: Key Street / Star Lane / Burtons (St. Peter Port) employment use (office), hotel and retail IP132: Former St Peter's Warehouse Site 4 Bridge Street, 73 dwellings with B1a office use, leisure or small scale retail IP136: Silo, College Street, 48 homes IP205: Burton's, College Street, 125 homes IP206: Cranfields, College Street, 140 homes

IP211: Regatta Quay, Key Street, 149 homes

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET1 To improve air quality	IP035 - IP132 - IP136 - IP205 - IP206 - IP211 -	Medium and long term Direct and indirect Irreversible Medium	Central Ipswich	The area along Star Lane, College Street, and Bridge Street is designated as AQMA. Five site allocations will contribute to an increase of new residents in the area as well as an increase from employment and commercial uses from offices, leisure or small scale retail with IP047 as it is anticipated new developments will be primarily residential (IP132, IP136, IP205, IP206, and IP211). As a result of the influx of a significant number of new residents the traffic is likely to increase on key roads adjacent to the new developments. In addition, IP035 has the potential to generate traffic in peak hours from users of the office or hotel buildings. Although the sites are well served by public transport and near two bus stations, the number of new residents suggests potential for congestion issues. Therefore, sustainable modes of transport should be encouraged through improved pedestrian/cycle routes in the area.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET2 To conserve soil resources and quality	IP035 + IP132 + IP136 + IP205 + IP206 + IP211 +	Medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	The sites are located in urban Ipswich, on brownfield land and are likely to be contaminated due to previous use of the land and opportunities <i>could be sought</i> <i>to remediate the land before use.</i> Therefore it is considered that the effect in the medium and long term is likely to be beneficial.
ET3 To reduce waste	IP035 - IP132 0 IP136 0 IP205 - IP206 - IP211 -	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	Waste is likely to be generated in the short term during construction and in the long term during operation (household waste as well as commercial waste – IP132). The size of the sites and the indicative capacity of IP205, IP206, and IP211 suggest significant increase of new residents and potential for negative effects if mitigation measures are not in place. Cumulatively, negative effects are likely to occur when the site allocations are considered along with other sites in IP- one area and the whole borough. <i>Opportunities should be sought to</i> <i>encourage recycling and reuse of</i> <i>materials.</i>
ET4 To reduce the effects of traffic upon the environment	IP035 - IP132 - IP136 - IP205 IP206 IP211	Short, medium and long term Direct Irreversible Medium	Central Ipswich	Whilst there are two bus stations and Ipswich Railway Station are located near the sites, the area is not currently served by local public transport. In addition, the main shopping area is located north of the sites and existing employment areas are further south and west which may result in increase of the use of private cars and congestion issues. For these reasons it is considered that negative effects are likely to occur unless appropriate mitigation measures are put in place. A transport assessment and travel plan will be required. Pedestrian capacity along College St footways and the one- way traffic system will be considered. This will help to reduce potential negative effects on traffic.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
				Mitigation measures at project level can be included in Section 106 Agreement with regards to pedestrian infrastructure improvements or new bus stops (if any bus route is diverted to meet the local needs).
ET5 To improve access to key services for all sectors of the population	IP035 +/- IP132 +/- IP136 +/- IP205 +/- IP206 +/- IP211 +/-	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	Although the sites are located in the central urban area of Ipswich near main community facilities which would offer some benefits in terms of access, access constraints are identified with the majority of sites associated with potential negative effects on the local highway network from potential congestion issues (Star Lane). <i>Sustainable modes of transport should be encouraged through improved</i> <i>pedestrian infrastructure and/or where</i> <i>appropriate the creation of cycle</i> <i>routes/lanes.</i>

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET6 To limit and adapt to climate change	IP035 - IP132 - IP136 - IP205 - IP206 - IP211 -	Short, medium and long term Direct Irreversible Medium	Central Ipswich	All the sites are located in Flood Zones 2 and 3 with a risk of flooding from the river Orwell. <i>Flood risk assessment may be required</i> <i>at project assessment level to identify</i> <i>appropriate mitigation measures. New</i> <i>developments should be encouraged to</i> <i>use SuDS to manage runoff, reduce</i> <i>further flood risk and protect water</i> <i>quality.</i> As previously identified, <i>the use of</i> <i>sustainable modes of transport should</i> <i>be encouraged due to potential of</i> <i>increased traffic and congestion issues</i> <i>in the vicinity of the sites.</i> Cumulatively, the effects from climate change can be reduced through encouraging people to use more sustainable modes of transport. The central location of these sites offers the potential for walking and cycling.
ET7 To protect and enhance the quality of water features and resources and reduce the risk of flooding	IP035 - IP132 - IP136 - IP205 - IP206 - IP211 -	Short, medium and long term Direct Irreversible Medium	Central Ipswich	All the sites are located in Flood Zones 2 and 3 with a risk of flooding from the river Orwell. Although in the medium and long term it is anticipated that the risk will be alleviated through the delivery of Ipswich Flood Defence, <i>flood risk assessment</i> <i>may be required at project assessment</i> <i>level to identify appropriate mitigation</i> <i>measures for the parts of the</i> <i>development that falls within Flood Zone</i> <i>2 and 3 (e.g. through design).</i> Water quality could be enhanced if opportunities are sought to remediate the contaminated land where appropriate. Surface water flooding local to site – will be considered at planning application stage for IP136. See Appendix 1 of the Ipswich SFRA.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET8 To conserve and enhance biodiversity and geodiversity, including favourable conditions on SSSIs, SPAs and SACs	IP035 0 IP132 0 IP136 0 IP205 0 IP206 - IP211 -	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	IP132, IP136, IP205, IP206 and IP211 are adjacent to County Wildlife Site (the river Orwell and the Wet Dock), which is also designated as an important wildlife corridor with high conservation value. Potential negative effects are associated with noise, air pollution and disturbance of species.
ET9 To conserve and enhance the historic environment, heritage assets and their settings	IP035 - IP132 - IP136 - IP205 - IP206 - IP211 -	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	Listing buildings are adjacent or near IP035, IP132 and IP136 with potential for negative effects from new developments:

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET10 To conserve and enhance the quality and local distinctiveness of landscapes and townscapes	IP035 - IP132 - IP136 - IP205 - IP206 - IP211 -	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	Potential negative effects on listed buildings and local distinctiveness of the conservation area are identified with the development of all the sites. In addition, IP205, IP206, and IP211 are located at the Waterfront and consideration should be given to any negative impacts on the local character of Neptune Marina and the quays in the area. The townscape distinctiveness should be conserved through good design which responds to the character of surrounding buildings to maintain the local distinctiveness of the area.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
HW1 To improve the health of those most in need	IP035 0 IP132 + IP136 + IP205 + IP206 + IP211 +	Short, medium and long term Direct / Indirect Irreversible Medium	Localised area in Central Ipswich	The provision of decent housing associated with the development of IP132, IP136, IP205, IP206, and IP211 may have some positive indirect effects on health as it will result in improvement of the quality of life of residents in new developments. The sites are also near designated river paths at the Waterfront which may encourage more people to lead healthy life styles.
HW2 To improve the quality of life where people live and encourage community participation	IP035 0 IP132 + IP136 + IP205 + IP206 + IP211 +	Short, medium and long term Direct / Indirect Reversible Medium	Central Ipswich	Community participation will be encouraged through the allocation of sites in the central urban area near a wide range of facilities (e.g. parks, public transport infrastructure, university campus, etc.). The sites will also contribute to the achievement of the SA objective through the provision of decent housing and some employment opportunities (IP035) and thus improving the overall quality of life. In addition, leisure provision with IP132 may also contribute towards achieving the SA Objective.
ER1 To reduce poverty and social exclusion	IP035 0 IP132 + IP136 + IP205 + IP206 + IP211 +	Short, medium and long term Direct / Indirect Reversible Medium	Central Ipswich	Social exclusion is likely to be reduced through the allocation of sites for residential uses near a wide range of community facilities. <i>However, opportunities to improve the</i> <i>public transport provision in the vicinity of</i> <i>the new developments can be beneficial</i> <i>in terms of access.</i>

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ER2 To offer everybody the opportunity for rewarding and satisfying employment	IP035 + IP132 + IP136 + IP205 0 IP206 0 IP211 0	Short, medium Direct Reversible Medium	Localised area in Central Ipswich	The only sites with potential to contribute to the achievement of the SA objective are IP035 and IP136 through the allocation of land for business, retail and mixed use. IP132 would also partly contribute towards achieving the SA Objective with provision of B1a office use, leisure or small scale retail though it would not be on a significant scale.
ER3 To help meet the housing requirements for the whole community	IP035 0 IP132 + IP136 + IP205 ++ IP206 ++ IP211 ++	Short, medium and long term Direct Irreversible Medium	Central Ipswich	IP205, IP206, IP211 strongly support the SA objective through the delivery of significant number of residential dwellings. However, these dwelling are all anticipated to be flats and may lead to oversupply of this type of housing in central Ipswich.
ER4 To achieve sustainable levels of prosperity and economic growth throughout the plan area	IP035 + IP132 + IP136 + IP205 + IP206 + IP211 +	Short, medium and long term Direct/Indirect Irreversible Medium	Central Ipswich	Whilst the provision of decent housing and some employment opportunities will improve the overall attractiveness of the area making it a better place to live, on its own each is unlikely to contribute to achievement of the SA objective, However, cumulative positive effects are likely to occur as a result of regeneration of the Waterfront area and enhancement of the existing land use.
ER5 To support vital and viable town, district and local centres	IP035 + IP132 + IP136 + IP205 + IP206 + IP211 +	Short, medium and long term Direct/Indirect Irreversible Medium	Localised area in Central Ipswich	All the sites are located in the town centre and new residents would add to its vitality and viability. The development of IP035 and IP136 will contribute to the achievement of the SA objective through the allocation of small amount of land for retail and business uses to meet the demands of the growing number of residents in the area.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ER6 To encourage efficient patterns of movement in support of economic growth	IP035 +/- IP132 +/- IP136 +/- IP205 +/- IP206 +/- IP211 +/-	Short, medium and long term Direct/Indirect Irreversible Low	Central Ipswich	The central location of the sites may lead to more efficient patterns of movement. However, it is uncertain whether sustainable modes of transport will be used unless public transport services are provided and pedestrian routes linking the Waterfront to the central shopping area are enhanced.
ER7 To encourage and accommodate both indigenous and inward investment	IP035 + IP132 + IP136 + IP205 + IP206 + IP211 +	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	On its own, each site is unlikely to contribute to the achievement of the SA objective. Cumulatively, the sites will contribute to the regeneration / redevelopment of central town areas to enhance the image of the Waterfront, with the provision of new hotel (IP035), retail units and car parking (IP035), new decent housing (IP132, IP136, IP205, IP206 and IP211) and provision of office use, leisure or small scale retail (IP132) and improve the business attractiveness of the area.
CL1 To maintain and improve access to education and skills for both young people and adults	IP035 0 IP132 + IP136 + IP205 + IP206 + IP211 +	Short, medium and long term Indirect Reversible Medium	Localised area in Central Ipswich	Potential positive effects are associated with the close proximity of the sites to University Campus Suffolk which may contribute indirectly to the achievement of the SA objective.
CD1 To minimise potential opportunities for crime and anti-social activity	IP035 0 IP132 0 IP136 0 IP205 0 IP206 0 IP211 0	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	On the whole, crime levels are higher in Ipswich than national average. It should be noted that the sites are located in an area ranked as most deprived. Therefore, <i>it would be beneficial to</i> <i>incorporate secured by design principles</i> <i>within new developments.</i> <u>http://opendatacommunities.org/deprivati</u> <u>on/map</u>

South East – Holywells – Island site

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET1 To improve air quality	IP037 -	Medium and long term Direct and indirect Irreversible Medium	Central Ipswich	The site is close to an area designated as AQMA along Star Lane, College Street, and Bridge Street located to the north and northeast of the island. The site will contribute to an increase of new residents in the area as it is anticipated 50% of the land will be allocated for residential use with indicative capacity of 271 new homes. As a result of the influx of a significant number of new residents the traffic is likely to increase on the road that connects the island to the mainland and the impacts may affect a greater number of people than are currently affected by traffic on and around the Island site. In addition, IP037 has the potential to generate traffic in peak hours from users of the office or leisure facilities. The site is currently not served by public transport (due to its existing use) and the additional vehicular access may lead to increase in the use of private cars in the area. Potential negative effects may occur as a result of congestion at junctions Grey Friars Road/A1022 and Bridge Street/A137. Therefore, sustainable modes of transport should be encouraged through the provision of safe pedestrian/cycle bridge at an appropriate location.
ET2 To conserve soil resources and quality	IP037 +	Medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	The site is located in urban Ipswich, on brownfield land and is potentially contaminated due to existing use of the land and opportunities <i>could be sought</i> <i>to remediate the land before use.</i> Therefore it is considered that the effect in the medium and long term is likely to be beneficial.

IP037: Island site, 271 homes, open space, B1 uses, boat building

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET3 To reduce waste	IP037 -	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	Waste is likely to be generated in the short term during construction and in the long term during operation (household waste). The size of the site suggests significant increase of new residents and potential for negative effects if mitigation measures are not in place. Cumulatively, negative effects are likely to occur when the site allocations are considered along with other sites in IP- one area and the whole borough. <i>Opportunities should be sought to</i> <i>encourage recycling and reuse of</i> <i>materials.</i>
ET4 To reduce the effects of traffic upon the environment	IP037 -	Short, medium and long term Direct Irreversible Medium	Central Ipswich	Whilst there are two bus stations and Ipswich Railway Station are located near the sites, the area is not currently served by local public transport. In addition, the main shopping area is located north of the site and existing/future employment areas are further west and east from the river banks which may result in increase of the use of private cars and congestion issues on the approach roads to and from the island. For these reasons it is considered that negative effects are likely to occur unless appropriate mitigation measures are put in place. A transport assessment and travel plan will be required for the site. Traffic impact especially on Bridge Street will be considered. This will help to avoid potential negative impacts on traffic. <i>Mitigation measures at project level can be included in Section 106 Agreement with regards to pedestrian/cycle infrastructure improvements (new bridge) with extensions where appropriate along the existing river walks or along roads leading to key facilities. Due to the size of the site a separate Transport assessment may be required to identify the need to provide a new bus service in the area.</i>

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET5 To improve access to key services for all sectors of the population	IP037 +/-	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	Although the site is located in the central urban area of lpswich, access constraints are identified associated with the potential increase of the residents and the need to provide vehicular/pedestrian/cycle access via a new bridge. Potential positive effects are associated with the provision of some services on site as it anticipated that the development will be mixed use and land will be allocated for leisure and small scale retail uses. <i>Sustainable modes of transport should be encouraged through improved</i> <i>pedestrian and cycle infrastructure.</i>
ET6 To limit and adapt to climate change	IP037 -	Short, medium and long term Direct Irreversible Medium	Central Ipswich	The entire site is located in Flood Zones 2 and 3 with a risk of flooding from the river Orwell. Therefore <i>development</i> <i>should be directed to areas of lower</i> <i>flood risk through the Sequential Test</i> <i>process and highly vulnerable</i> <i>development should not be permitted</i> <i>unless there are no alternative sites</i> <i>available.</i> <i>Flood risk assessment may be required</i> <i>at project assessment level to identify</i> <i>appropriate mitigation measures if the</i> <i>site is taken forward.</i> Cumulatively, the effects from climate change can be reduced through encouraging people to use more sustainable modes of transport. However, currently limited opportunities exist for sustainable movements to and from the site.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET7 To protect and enhance the quality of water features and resources and reduce the risk of flooding	IP037 -	Short, medium and long term Direct Irreversible Medium	Central Ipswich	The entire site is located in Flood Zones 2 and 3 with a risk of flooding from the river Orwell. Although in the medium and long term it is anticipated that the risk will be alleviated through the delivery of Ipswich Flood Defence, <i>the flood risk is</i> <i>considered to be high. Therefore</i> <i>development should be directed to areas</i> <i>of lower flood risk through the Sequential</i> <i>Test process and highly vulnerable</i> <i>development should not be permitted</i> <i>unless there are no alternative sites</i> <i>available.</i> There is also potential for deterioration of water quality of the river Orwell as a result of construction activities and increased development along the river banks and on the island.
ET8 To conserve and enhance biodiversity and geodiversity, including favourable conditions on SSSIs, SPAs and SACs	IP037 -	Short, medium and long term Direct/Indirect Irreversible Medium	Central Ipswich The river Orwell	The site is surrounded by designated County Wildlife site and is close to a designated SPA site located south of the island along the river Orwell. Potential negative effects are associated with deterioration of air quality and water quality, disturbance of protected birds (from noise and dust). Therefore Habitats Regulation Assessment will be required to ensure that no adverse effects are likely to occur as a result of a new development.
Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
---	--------------------------	---	--------------------------------------	--
ET9 To conserve and enhance the historic environment, heritage assets and their settings	IP037 -	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	Whilst there are no Listing Buildings located on the island, the entire site is located within a Conservation area and Area of Archaeological importance. <i>Appropriate design of building should be</i> <i>considered in order to avoid any</i> <i>negative effects on the local character</i> <i>particularly with regards to the Neptune</i> <i>marina and the Neptune quay.</i> Mitigation measures may be required if any archaeological remains are discovered to avoid damage of heritage assets (<i>where appropriate the</i> <i>completion of a licensed excavation and</i> <i>recording of remains before development</i> <i>commences</i>).
ET10 To conserve and enhance the quality and local distinctiveness of landscapes and townscapes	IP037 +/-	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	IP037 is located at the Waterfront and consideration should be given to any negative effects on the local character of Neptune Marina and the quays in the area. Potential positive effects are likely to occur with the allocation of 15% of the site for open space. The townscape distinctiveness should be conserved through <i>appropriate design to</i> <i>maintain the local distinctiveness of the</i> <i>area. It would be beneficial if more land</i> <i>is allocated for open space along the</i> <i>banks of the island to enhance the</i> <i>landscape/townscape of the site.</i>

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
HW1 To improve the health of those most in need	IP037 +/-	Short, medium and long term Indirect Irreversible Medium	Localised area in Central Ipswich	Potential negative indirect effects are associated with the high level of flood risk. Negative indirect effects are also likely to occur as a result of deterioration of the air quality. Some positive effects may be associated with the allocation of open space with new development. <i>Opportunities should be sought to use the river Orwell for leisure activities where appropriate.</i>
HW2 To improve the quality of life where people live and encourage community participation	IP037 -	Short, medium and long term Direct / Indirect Reversible Medium	Central Ipswich	Although the site is located in the Central IP-One area, severance occurs due to the river Orwell playing a role of a barrier from all sides of the island. The only road connecting the site with the main land is located to the northwest. Although some transport infrastructure improvements are envisaged with regards to the redevelopment of the site, <i>careful consideration should be given on the ancillary facilities that may be required within a new mixed use development in order to ensure the quality of life of new residents is improved (e.g. post office).</i>
ER1 To reduce poverty and social exclusion	IP037 +/-	Short, medium and long term Direct / Indirect Reversible Medium	Central Ipswich	Social exclusion is likely to be reduced through the allocation of the site for mixed use and the provision of open space on-site. <i>However, improvements</i> <i>with regards to access to the site will be</i> <i>crucial to ensure residents are not</i> <i>affected by the existing movement</i> <i>barrier of the river Orwell.</i>

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ER2 To offer everybody the opportunity for rewarding and satisfying employment	IP037 +	Short, medium Direct Reversible Medium	Localised area in Central Ipswich	Some employment opportunities are identified with regards to the allocation of 30% of the land to employment/leisure use and 5% small scale retail use. In addition, temporary jobs will be created during construction of the envisaged transport infrastructure and residential dwellings.
ER3 To help meet the housing requirements for the whole community	IP037 +	Short, medium and long term Direct Irreversible Medium	Central Ipswich	The site allocation supports the SA objective through the delivery of 271 residential dwellings. However, these dwelling are all anticipated to be flats and may lead to oversupply of this type of housing in central Ipswich.
ER4 To achieve sustainable levels of prosperity and economic growth throughout the plan area	IP037 0	Short, medium and long term Direct/Indirect Irreversible Medium	Central Ipswich	Whilst the provision of decent housing and some employment opportunities will improve the overall attractiveness of the area making it a better place to live, on its own the site is unlikely to contribute to achievement of the SA objective due a number of constraints identified with its location. However, cumulative positive effects are likely to occur as a result of regeneration of the Waterfront area and enhancement of the existing land use.
ER5 To support vital and viable town, district and local centres	IP037 +	Short, medium and long term Direct/Indirect Irreversible Medium	Localised area in Central Ipswich	The development of IP037 has the potential to contribute to the achievement of the SA objective through the allocation of land for retail and leisure uses to meet the demands of the growing number of residents in the area.
ER6 To encourage efficient patterns of movement in support of economic growth	IP037 ?	Short, medium and long term Direct/Indirect Irreversible Low	Central Ipswich	Efficient patterns of movement could be encouraged if a new pedestrian/cycling bridge is built at an appropriate location to connect residents to key community facilities and some employment areas in Holywells. It is uncertain whether sustainable modes of transport will be used unless <i>public transport services are</i> <i>provided in the area of Bridge Street and</i> <i>pedestrian routes are enhanced.</i>

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ER7 To encourage and accommodate both indigenous and inward investment	IP037 +	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	On its own, the site is unlikely to contribute to the achievement of the SA objective. Cumulatively, the site will contribute to the regeneration / redevelopment of the Waterfront through the provision of new retail units and open space, new decent housing and thus improve the image and business attractiveness of the area.
CL1 To maintain and improve access to education and skills for both young people and adults	IP037 +	Short, medium and long term Indirect Reversible Medium	Localised area in Central Ipswich	Potential positive effects are associated with the close proximity of the site to University Campus Suffolk which may contribute indirectly to the achievement of the SA objective.
CD1 To minimise potential opportunities for crime and anti-social activity	IP037 -	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	The site is located in an area ranked as less deprived when compared to adjacent areas to the north and west of the site. However, in the short term, crime levels may increase as a result of influx of new residents. Therefore, <i>it</i> <i>would be beneficial to incorporate</i> <i>secured by design principles within new</i> <i>development</i> . <u>http://opendatacommunities.org/deprivati</u> <u>on/map</u>

South East – Holywells

IP142: Land at Duke Street, 26 homes + 25% open space IP098: Transco, south of Patteson Road, 51 homes IP226: Helena Road, 540 homes IP042: Land between Cliff Quay and Landseer Road, 27 homes plus offices, retail and leisure uses IP049: No. 8 Shed, Orwell Quay car parking, education

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET1 To improve air quality	IP142 - IP098 - IP226 - IP042 +/- IP049 +	Medium and long term Direct and indirect Irreversible Medium	Central Ipswich	IP142 and IP098 are located near an AQMA designated due to air pollution along Duke Street. As a result of the influx of new residents, traffic may be generated in the vicinity of Duke Street, the A1156, and Fore Street. The area is currently very well served by public transport. However, cumulatively, there is potential for congestion at key junctions of approach roads to the town centre or some employment areas. IP049 is directly adjacent to Duke Street which is an AQMA. IP049 is allocated for public car parking, however part of the existing site is now allocated for education and ancillary uses to university (e.g. GP surgery). This will result in reduced car parking capacity hence it can be assumed that less traffic will be generated to and from the site. The effects are assessed as positive in the long term but the level of certainty is low. Therefore, <i>the use of sustainable modes of transport should be encouraged</i> <i>through improved pedestrian routes in</i> <i>the area</i> .
ET2 To conserve soil resources and quality	IP142 + IP098 + IP226 + IP042 + IP049 +	Medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	The sites are located in urban Ipswich, on brownfield land and are likely to be contaminated due to previous use of the land and opportunities <i>could be sought</i> <i>to remediate the land before use.</i> Therefore it is considered that the effect in the medium and long term is likely to be beneficial.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET3 To reduce waste	IP142 0 IP098 0 IP226 - IP042 0 IP049 0	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	Waste is likely to be generated in the short term during construction and in the long term during operation (household waste). The indicative capacity of IP226 suggests potential for negative effects if mitigation measures are not in place. It is considered that IP049 will have a negligible effect on waste as the future use of the site will be the similar to the existing use. Any waste generated from the ancillary uses at IP049 is not assessed as significant. Cumulatively, negative effects are likely to occur when the site allocations are considered along with other sites in IP- one area. <i>Opportunities should be sought to</i> <i>encourage recycling and reuse of</i> <i>materials.</i>
ET4 To reduce the effects of traffic upon the environment	IP142 + IP098 + IP226 +/- IP042 + IP049 +	Short, medium and long term Direct Irreversible Medium	Central Ipswich	The relatively central location of the sites may result in reduced need to travel by private car. The area is very well served by public transport (three local buses running on Duke Street and two buses running on Bishops Hill). In addition, the sites are located near a District Centre on Duke Street. IP049 may contribute to the achievement of the SA objective as the currently existing car parking spaces will be reduced and the location of the site does not suggest that people will be using it to access the Central Shopping area. For all these reasons it is considered that the sites are likely to contribute to the achievement of the SA objective.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET5 To improve access to key services for all sectors of the population	IP142 + IP098 + IP226 - IP042 + IP049 +	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	All the sites are located in a central urban area of Ipswich near a District centre (48), parks, university buildings, etc. Potential access constraints are identified with IP226 associated with potential negative effects on the local highway network. <i>Sustainable modes of transport should be encouraged through improved pedestrian infrastructure and/or where appropriate the creation of cycle routes/lanes connecting major developments such as IP226 to the town centre, parks and gardens, etc. <i>A separate Transport assessment may be required to identify any significant negative effects from the development of IP226 due to its size and proposed use.</i></i>

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET6 To limit and adapt to climate change	IP142 - IP098 - IP226 IP042 - IP049 0	Short, medium and long term Direct Irreversible Medium	Central Ipswich	 IP049 is located in Flood Zones 2 or 3 prone to flooding from the river Orwell. However the site is already in use therefore there will be only negligible change from the existing baseline. IP098, and IP226 are located in Flood Zones 2 and 3 with a risk of flooding across the entire site. Western parts of IP142 and IP042 also fall within a Flood zone 2 and 3. Although the implementation of the tidal barrier and raised defences would raise the level of protection, the area will still be within Flood Zone 3 and there is still a residual risk of flooding by either failure of the new defences, or overtopping in extreme events. Therefore development should be directed to areas of lower flood risk through the Sequential Test process and highly vulnerable development should not be permitted unless there are no alternative sites available. Flood risk assessment may be required at project level to identify appropriate mitigation measures for the parts of the development (IP042 and IP142) that fall within Flood Zone 2 and 3 (e.g. through design). Cumulatively, the effects from climate change can be reduced through encouraging people to use more sustainable modes of transport.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET7 To protect and enhance the quality of water features and resources and reduce the risk of flooding	IP142 - IP098 - IP226 IP042 - IP049 0	Short, medium and long term Direct Irreversible Medium	Central Ipswich	IP098, and IP226 are located in Flood Zones 2 and 3 with a risk of flooding across the entire site. Western parts of IP142 and IP042 also fall within a Flood zone 2 and 3. Surface water flooding local to site – will be considered at planning application stage for IP098 and IP142. See Appendix 1 of the Ipswich SFRA. <i>New developments should be</i> <i>encouraged to use SuDS to manage</i> <i>runoff, reduce further flood risk and</i> <i>protect water quality. Flood risk</i> <i>assessment may be required at project</i> <i>assessment level to identify appropriate</i> <i>mitigation measures for the parts of the</i> <i>development that falls within Flood Zone</i> <i>2 and 3 (e.g. through design).</i> <i>Water quality could be enhanced if</i> <i>opportunities are sought to remediate the</i> <i>contaminated land where appropriate.</i>
ET8 To conserve and enhance biodiversity and geodiversity, including favourable conditions on SSSIs, SPAs and SACs	IP142 - IP098 0 IP226 - IP042 - IP049: 0	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	IP042 is adjacent to a CWS located to the northeast (Holywells Park) but the relatively low capacity of the site is likely to result in only negligible effects on the designated site. The close proximity of IP226 to the river Orwell to the west may result in some significant negative effects associated with pollution, noise and disturbance of species. There are no designated sites located near IP049. No significant impacts are likely to occur as the site is already in use and no major constraints in terms of biodiversity have been identified. TPO nearby is identified within the northern parts of IP142 and IP042 and application for Tree Works may be required. Potential indirect positive effects are associated with remediation of previously contaminated land and improved air quality due to reduced need to travel.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET9 To conserve and enhance the historic environment, heritage assets and their settings	IP142 0 IP098 0 IP226 0 IP042 + IP049 -	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	Listing buildings are adjacent or near IP042 (Cliff Cottage – Grade II and Tolly Cobbold Brewery – Grade II). It should be noted that the redevelopment which has planning permission at IP042 has been allowed as enabling development to bring the listed brewery back into use and thereby secure its future. Therefore effects are assessed as positive. IP226 and IP049 are adjacent to a Conservation area. <i>Appropriate design of</i> <i>building should be considered in order to</i> <i>avoid any negative effects on the setting</i> <i>of designated sites.</i> There is also potential for any archaeological remains at IP049 as it is located in an Area of Archaeological importance. Therefore mitigation measures may be required if any archaeological remains are discovered to avoid damage of heritage assets (where appropriate the completion of a licensed excavation and recording of remains before development
ET10 To conserve and enhance the quality and local distinctiveness of landscapes and townscapes	IP142 0 IP098 0 IP226 - IP042 + IP049 -	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	<i>commences</i>). Potential positive effects on listed buildings and local distinctiveness are identified with the development of IP042. The development of IP049 and IP226 (adjacent to a Conservation area) may have negative impacts. <i>Mitigation would</i> <i>include appropriate design to maintain</i> <i>the local distinctiveness of the area.</i>

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
HW1 To improve the health of those most in need	IP142 + IP098 + IP226 + IP042 + IP049 0	Short, medium and long term Indirect Irreversible Medium	Localised area in Central Ipswich	Positive indirect effects are likely to occur with regards to the close proximity of the sites to Holywells and Alexandra Park to the east and north east. The site allocations are also near designated river paths and close to community facilities which may encourage residents to lead a healthier lifestyle. There is no relationship between the allocation of IP049 and the SA objective.
HW2 To improve the quality of life where people live and encourage community participation	IP142 + IP098 + IP226 + IP042 + IP049 +	Short, medium and long term Direct / Indirect Reversible Medium	Central Ipswich	Community participation will be encouraged through the allocation of sites in a central urban area near a wide range of facilities (e.g. parks, public transport infrastructure, university campus, etc.). The sites will also contribute to the achievement of the SA objective through the provision of decent housing, car parking and ancillary building to the University (IP049) and open space (IP142) thus improving the overall quality of life.
ER1 To reduce poverty and social exclusion	IP142 + IP098 + IP226 + IP042 + IP049 0	Short, medium and long term Direct / Indirect Reversible Medium	Central Ipswich	Social exclusion will be reduced through the allocation of sites for residential/education uses near a wide range of community facilities easily accessible by different means of transport. No significant impacts on poverty are identified with the allocation of IP049 for car parking.
ER2 To offer everybody the opportunity for rewarding and satisfying employment	IP142 0 IP098 0 IP226 0 IP042 + IP049 0	Short Direct Reversible Medium	Localised area in Central Ipswich	Some employment opportunities are identified with IP042 (industrial and commercial use, hotel, food store included in the planning application along with residential units). The only employment opportunities likely to arise from the development of the rest of the sites are associated with temporary construction jobs. Therefore it is considered that the effects will be negligible.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ER3 To help meet the housing requirements for the whole community	IP142 + IP098 + IP226 ++ IP042 + IP049 0	Short, medium and long term Direct Irreversible Medium	Central Ipswich	IP142, IP098, IP226, and IP042 will contribute directly to the achievement of the SA objective through the delivery of new residential dwellings.
ER4 To achieve sustainable levels of prosperity and economic growth throughout the plan area	IP142 0 IP098 0 IP226 0 IP042 + IP049 0	Short, medium and long term Direct/Indirect Irreversible Medium	N/A	Some benefits are recorded with the allocation of land for industrial and commercial use, hotel, and food store along with residential units at IP042. It should be noted that cumulatively the allocation of homes, employment and food stores within on site allocation is considered sustainable. Whilst new residential developments will contribute to the achievement of the SA objective through improving the overall quality of life, it is considered that on its own the rest of the sites will have a negligible effect.
ER5 To support vital and viable town, district and local centres	IP142 + IP098 + IP226 + IP042 + IP049 0	Short, medium and long term Direct/Indirect Irreversible Medium	Localised area in Central Ipswich	The vitality of the nearby District Centre will be supported through the increase of the potential users and the easy access via Duke Street from all new developments. Car parking provision will be slightly reduced at IP049, however the effects on access to town, district, or local centres is assessed as negligible.
ER6 To encourage efficient patterns of movement in support of economic growth	IP142 + IP098 + IP226 + IP042 + IP049 0	Short, medium and long term Direct/Indirect Irreversible Medium	Central Ipswich	The provision of residential dwellings and student accommodation close to existing/future employment area and a wide range of community facilities will encourage efficient patterns of movement and in the long term this could help to support economic growth. Car parking provision will be slightly reduced at IP049, however the effects on patterns of movement is assessed as negligible.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ER7 To encourage and accommodate both indigenous and inward investment	IP142 0 IP098 0 IP226 0 IP042 + IP049 0	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	Some benefits are recorded with the allocation of land for industrial and commercial use, hotel, and food store along with residential units, at IP042. On their own, the rest of the sites are unlikely to contribute to the achievement of the SA objective as the focus is primarily on residential developments. Cumulatively, the sites will contribute to the regeneration / redevelopment of a central town area, with the provision of new decent housing (IP142, IP098, IP226) and new commercial buildings (IP042), car parking and university buildings (IP049) and improve the business attractiveness of the area.
CL1 To maintain and improve access to education and skills for both young people and adults	IP142 + IP098 + IP226 + IP042 + IP049 +	Short, medium and long term Direct/Indirect Reversible Medium	Localised area in Central Ipswich	Potential positive effects are associated with the close proximity of the sites to University Campus Suffolk which may contribute to the achievement of the SA objective. IP049 will contribute indirectly to the achievement of the SA objecting through land allocated for ancillary uses to University.
CD1 To minimise potential opportunities for crime and anti-social activity	IP142 0 IP098 0 IP226 - IP042 0 IP049 0	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	The sites are located in an area ranked as less deprived when compared to adjacent areas to the north and west of the site. However, in the short term, crime levels may increase as a result of influx of new residents, particularly with regards to major developments such as IP226. Therefore, <i>it would be beneficial to</i> <i>incorporate secured by design principles</i> <i>within new development</i> . <u>http://opendatacommunities.org/deprivati</u> <u>on/map</u>

South West – Bridge

IP188: Websters Saleyard site, Dock Street 9 dwellings IP039a: Land between Gower Street & Gt Whip Street 43 dwellings IP133: South of Felaw Street 33 dwellings IP200: Griffin Wharf, Bath Street 187 dwellings IP080: 240 Wherstead Road 27 dwellings

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET1 To improve air quality	IP188 +/- IP039a +/- IP133 +/- IP200 +/- IP080 +/-	Medium and long term Direct and indirect Irreversible Medium	Central Ipswich	IP188 and IP039a are located near an AQMA designated due to air pollution along Vernon Street and Bridge Street. Currently the sites are located near existing employment areas to the southeast which may result in reduced need to travel by private car. However, the increase of new residents may cause congestion and deteriorate air pollution to the north of the sites in the vicinity of Bridge Street, Star Lane and Commercial Road. IP133, IP200, and IP080 are located at a significant distance from the AQMA; however it is likely that residents from the new housing dwellings within these sites generate traffic northbound to access shopping areas and community services in the town centre area. The area is currently very well served by public transport. However, <i>sustainable</i> <i>modes of transport should be</i> <i>encouraged through improved</i> <i>pedestrian routes and where appropriate</i> <i>the creation of cycling infrastructure in</i> <i>the area</i> .
ET2 To conserve soil resources and quality	IP188 + IP039a + IP133 + IP200 + IP080 +	Medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	The sites are located in urban lpswich, on brownfield land and are likely to be contaminated due to previous use of the land and opportunities <i>should be sought</i> <i>to remediate the land before use.</i> Therefore it is considered that the effect in the medium and long term is likely to be beneficial.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET3 To reduce waste	IP188 0 IP039a 0 IP133 0 IP200 - IP080 0	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	Waste is likely to be generated in the short term during construction and in the long term during operation (household waste). The indicative capacity of IP200 suggests potential for negative effects if mitigation measures are not in place. Cumulatively, negative effects are likely to occur when the site allocations are considered along with other sites in IP- one area. <i>Opportunities should be sought to</i> <i>encourage recycling and reuse of</i> <i>materials.</i>
ET4 To reduce the effects of traffic upon the environment	IP188 + IP039a + IP133 + IP200 + IP080 +	Short, medium and long term Direct Irreversible Medium	Central Ipswich	The relatively central location of the sites may result in reduced need to travel by private car. The area is very well served by public transport (local buses running on Vernon Street, Hawes Street, and Burrell Road). In addition, the sites are located near major employment areas to the south and northwest. For all these reasons it is considered that the sites are likely to contribute to the achievement of the SA objective <i>if mitigation measures</i> <i>to encourage the use sustainable modes</i> <i>of transport are in place.</i>

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET5 To improve access to key services for all sectors of the population	IP188 + IP039a + IP133 - IP200 - IP080 +	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	All the sites are located in a central urban area of Ipswich near parks, schools, river paths, with good public transport access to the main shopping area to the north. Potential access constraints are identified with IP133 and IP200 associated with potential negative effects on the local highway network. <i>Sustainable modes of transport should be encouraged through improved</i> <i>pedestrian infrastructure and/or where</i> <i>appropriate the creation of cycle</i> <i>routes/lanes connecting major</i> <i>developments such as IP200 to the town</i> <i>centre, parks and gardens, etc.</i> <i>A separate Transport Impact</i> <i>assessment may be required to identify</i> <i>any significant negative effects from the</i> <i>development of IP200 due to its size and</i> <i>proposed use.</i>
ET6 To limit and adapt to climate change	IP188 - IP039a - IP133 - IP200 - IP080 0	Short, medium and long term Direct Irreversible Medium	Central Ipswich	IP188, IP039a, IP133, IP200 are located in Flood Zones 2 or 3 and are prone to flooding from the river Orwell. IP080 is adjacent to Flood zone 2 or 3. <i>Although the implementation of the tidal</i> <i>barrier and raised defences would raise</i> <i>the level of protection, there is still a</i> <i>residual risk of flooding by either failure</i> <i>of the new defences, or overtopping in</i> <i>extreme events.</i> <i>Therefore flood risk assessment may be</i> <i>required at project assessment level to</i> <i>identify appropriate mitigation measures</i> <i>for the parts of the development that fall</i> <i>within Flood Zone 2 or 3 (e.g. through</i> <i>design).</i> <i>Cumulatively, the effects from climate</i> <i>change can be reduced through</i> <i>encouraging people to use more</i> <i>sustainable modes of transport.</i>

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET7 To protect and enhance the quality of water features and resources and reduce the risk of flooding	IP188 - IP039a - IP133 - IP200 - IP080 0	Short, medium and long term Direct Irreversible Medium	Central Ipswich	The majority of sites are located in Flood Zones 2 or 3 with a risk of flooding form the river Orwell. Surface water flooding local to site – will be considered at planning application stage for IP039a, IP133, IP188 and IP080. See Appendix 1 of the Ipswich SFRA. <i>New developments should be</i> <i>encouraged to use SuDS to manage</i> <i>runoff, reduce further flood risk and</i> <i>protect water quality. Flood risk</i> <i>assessment may be required at project</i> <i>assessment level to identify appropriate</i> <i>mitigation measures for the parts of the</i> <i>development that falls within Flood Zone</i> <i>2 and 3 (e.g. through design).</i> Water quality could be enhanced if opportunities are sought to remediate the contaminated land where appropriate.
ET8 To conserve and enhance biodiversity and geodiversity, including favourable conditions on SSSIs, SPAs and SACs	IP188 +/- IP039a 0 IP133 +/- IP200 +/- IP080 0	Short, medium and long term Direct/Indirect Irreversible Medium	Localised area in Central Ipswich	IP200, IP133 and IP188 are adjacent to a County Wildlife Site (CWS) located to the east (river Orwell) and the indicative capacity of the sites suggests potential negative effects on the designated site associated with pollution, noise and disturbance of species. <i>Potential indirect</i> <i>positive effects are associated with</i> <i>remediation of previously contaminated</i> <i>land and improved air quality due to</i> <i>reduced need to travel.</i>

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET9 To conserve and enhance the historic environment, heritage assets and their settings	IP188 - IP039a - IP133 - IP200 0 IP080 0	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	Listing buildings are adjacent IP188 and IP133 with potential for negative effects from new development (Stoke Bridge Maltings and Felaw Maltings). The only site to fall entirely within the boundary of a Conservation area is IP188. Part of IP133 and IP200 are adjacent to Wet Dock Conservation area. IP133 is also adjacent to the Grade II listed Felaw Maltings. <i>Appropriate design of building should be considered in order to avoid any negative effects on the setting of designated sites.</i> There is also potential for any archaeological remains at IP188, IP039a, and IP133 as they are located in an Area of Archaeological importance. Therefore mitigation measures may be required if any archaeological remains are discovered to avoid damage of heritage assets (<i>where appropriate the completion of a licensed excavation and recording of remains before development commences</i>).
ET10 To conserve and enhance the quality and local distinctiveness of landscapes and townscapes	IP188 - IP039a - IP133 - IP200 0 IP080 0	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	Potential negative effects on listed buildings and local distinctiveness of the conservation area are identified with the development of IP188, IP039a, and IP133. The townscape distinctiveness should be conserved through <i>the use of traditional</i> <i>or sympathetic building materials and</i> <i>techniques which respect those found on</i> <i>the surrounding buildings to maintain the</i> <i>local distinctiveness of the area.</i>
HW1 To improve the health of those most in need	IP188 + IP039a + IP133 + IP200 + IP080 +	Short, medium and long term Indirect Irreversible Medium	Localised area in Central Ipswich	Positive indirect effects are likely to occur with regards to the proximity of all the sites to Gippeswyk Park and Stoke Park Wood and The People's Community Garden located to the west and south west of the site allocations. The site allocations are also near designated river paths and close to community facilities which may

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
				encourage residents to lead a healthier lifestyle.
HW2 To improve the quality of life where people live and encourage community participation	IP188 + IP039a + IP133 + IP200 + IP080 +	Short, medium and long term Direct / Indirect Reversible Medium	Central Ipswich	Community participation will be encouraged through the allocation of sites in a central urban area near a wide range of facilities (e.g. parks, public transport infrastructure, etc.). The sites will also contribute to the achievement of the SA objective through the provision of decent housing thus improving the overall quality of life.
ER1 To reduce poverty and social exclusion	IP188 + IP039a + IP133 + IP200 + IP080 +	Short, medium and long term Direct / Indirect Reversible Medium	Central Ipswich	Social exclusion will be reduced through the allocation of sites for residential uses near a wide range of community facilities easily accessible by different means of transport.
ER2 To offer everybody the opportunity for rewarding and satisfying employment	IP188 0 IP039a 0 IP133 0 IP200 0 IP080 0	Short Direct Reversible Medium	Localised area in Central Ipswich	The only employment opportunities likely to arise from the development of the sites are associated with temporary construction jobs. Therefore it is considered that the effects will be negligible.
ER3 To help meet the housing requirements for the whole community	IP188 + IP039a + IP133 + IP200 ++ IP080 +	Short, medium and long term Direct Irreversible Medium	Central Ipswich	IP200 will contribute directly to the achievement of the SA objective through the delivery of new residential dwellings. IP188, IP039a, IP133, and IP080 are also allocated for residential use with lower indicative capacity.
ER4 To achieve sustainable levels of prosperity and economic growth throughout the plan area	IP188 0 IP039a 0 IP133 0 IP200 0 IP080 0	Short, medium and long term Direct/Indirect Irreversible Medium	N/A	Whilst new residential developments and will contribute to the achievement of the SA objective through improving the overall quality of life, it is considered that on its own each site will have a negligible effect.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ER5 To support vital and viable town, district and local centres	IP188 + IP039a + IP133 + IP200 + IP080 0	Short, medium and long term Direct/Indirect Irreversible Medium	Localised area in Central Ipswich	The vitality of the nearby District Centre (located to the west of the site allocations) will be supported through the increase of the potential users and the easy access via public transport services from most new developments.
ER6 To encourage efficient patterns of movement in support of economic growth	IP188 + IP039a + IP133 + IP200 + IP080 +	Short, medium and long term Direct/Indirect Irreversible Medium	Central Ipswich	The provision of residential dwellings close to existing employment area and a wide range of community facilities will encourage efficient patterns of movement and in the long term this could help to support economic growth.
ER7 To encourage and accommodate both indigenous and inward investment	IP188 0 IP039a 0 IP133 0 IP200 0 IP080 0	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	On its own, each site is unlikely to contribute to the achievement of the SA objective as the focus is primarily on residential developments. Cumulatively, the sites will contribute to the regeneration / redevelopment of a central town area, with the provision of new decent housing and improve the business attractiveness of the borough.
CL1 To maintain and improve access to education and skills for both young people and adults	IP188 + IP039a + IP133 + IP200 + IP080 +	Short, medium and long term Direct Reversible Medium	Localised area in Central Ipswich	Potential positive effects are associated with the close proximity of the sites to St. Joseph's College and University Campus Suffolk which may contribute to the achievement of the SA objective.
CD1 To minimise potential opportunities for crime and anti-social activity	IP188 0 IP039a 0 IP133 0	Short, medium and long term Direct Irreversible	Localised area in Central Ipswich	On the whole, crime levels are higher in Ipswich than national average. It should also be noted that the sites are located in an area ranked as most deprived.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
	IP200 - IP080 0	Medium		In the short term, crime levels may increase as a result of influx of new residents, particularly with regards to major developments (IP200).
				Therefore, it would be beneficial to incorporate secured by design principles within new development.
				http://opendatacommunities.org/deprivati on/map

South West – Bridge

IP031: Burrell Road, 20 homes IP169: 23-25 Burrell Road, 14 homes IP047: Land at Commercial Road, 103 dwellings, office, leisure, hotel and open space

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET1 To improve air quality	IP031 0 IP169 0 IP047-	Medium and long term Direct and indirect Irreversible Medium	Central Ipswich	The sites are located near an AQMA designated due to air pollution along Vernon Street and Bridge Street. IP047 will contribute to an increase of new residents in the area as it is anticipated that 103 new homes will be built. Along with new office, leisure and hotel development, traffic is likely to increase on key roads adjacent to the site. It is considered likely that residents from the new housing dwellings at IP031 and IP169 could generate additional traffic eastbound along Burrell road to access shopping areas and community services in the town centre area via Bridge Street. However, the number of dwellings does not suggest a significant change from the baseline. IP047 does however incorporate public open space which may contribute partially to the achievement of the SA objective through the provision of 0.17ha open space at the Waterfront. However, cumulative negative effects on air quality are likely to occur if the mitigation measures identified below are not put in place. The area is currently very well served by public transport and the use of sus <i>tainable modes of transport should be further encouraged through improved pedestrian routes and where appropriate the creation of cycling infrastructure.</i>

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET2 To conserve soil resources and quality	IP031 + IP169 0 IP047 +	Medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	The sites are located in urban Ipswich on previously developed land. IP031 and IP047 are potentially contaminated due to previous use of the land and opportunities <i>should be sought to</i> <i>remediate the land before use</i> . Therefore it is considered that the effect in the medium and long term is likely to be beneficial.
ET3 To reduce waste	IP031 0 IP169 0 IP047 -	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	Waste is likely to be generated in the short term during construction and in the long term during operation (household and commercial waste). The capacity of IP031 and IP169 is low in number of residential units, therefore on their own the development of the sites is not anticipated to result in any significant negative effects. The size and indicative capacity of IP047 suggest significant increase of new residents and commercial use and potential for negative effects if mitigation measures are not in place. Cumulatively, negative effects are likely to occur when the site allocations are considered along with other sites in IP- one area. <i>Opportunities should be sought to encourage recycling and reuse of materials.</i>
ET4 To reduce the effects of traffic upon the environment	IP031 + IP169 + IP047 +	Short, medium and long term Direct Irreversible Medium	Central Ipswich	The relatively central location of the sites may result in reduced need to travel by private car. The area is very well served by public transport (one local bus running on Burrell Road connecting the area with facilities on the other side of the river). Ipswich railway station is at a walking distance from IP031, IP169, and 047. In addition, the sites are located near major employment areas to the southeast and northwest. For all these reasons it is considered that the sites are likely to contribute to the achievement of the SA objective.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET5 To improve access to key services for all sectors of the population	IP031 + IP169 + IP047 +	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	All the sites are located in a central urban area of Ipswich near parks, schools, river paths, Ipswich railway Station, with a good public transport access to the main shopping area to the north. <i>Sustainable modes of transport should be encouraged through improved pedestrian infrastructure and/or where appropriate the creation of cycle routes/lanes.</i>
ET6 To limit and adapt to climate change	IP031 - IP169 - IP047 -	Short, medium and long term Direct Irreversible Medium	Central Ipswich	IP031 and IP047 are located in Flood Zones 2 or 3 being prone to flooding from the river Orwell. IP169 is adjacent to Flood Zone 2 or 3. Although the implementation of the tidal barrier and raised defences would raise the level of protection, there is still a residual risk of flooding by either failure of the new defences, or overtopping in extreme events. Therefore flood risk assessment may be required at project assessment level to identify appropriate mitigation measures for the parts of the development that fall within Flood Zone 2 or 3 (e.g. through design). Cumulatively, positive effects are likely to occur as a result of the allocation of sites near community facilities with a reduced need to travel by car.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET7 To protect and enhance the quality of water features and resources and reduce the risk of flooding	IP031 +/- IP169 – IP047 -	Short, medium and long term Direct Irreversible Medium	Central Ipswich	IP031 and IP047 are located in Flood Zones 2 or 3 being prone to flooding from the river Orwell. IP169 is adjacent to Flood Zone 2 or 3. Water quality could be enhanced if opportunities are sought to remediate the contaminated land (IP031) before further use of the land. New developments should be encouraged to use SuDS to manage runoff, reduce further flood risk and protect water quality. Flood risk assessment may be required at project assessment level to identify appropriate mitigation measures for the parts of the development that falls within Flood Zone 2 and 3 (e.g. through design).
ET8 To conserve and enhance biodiversity and geodiversity, including favourable conditions on SSSIs, SPAs and SACs	IP031 0 IP169 0 IP047 +	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	There are no TPOs located near IP031 and IP169. All the sites are adjacent to a County Wildlife site. Potential indirect negative effects are likely to occur with regards to air pollution and temporary noise disturbance during construction. However, due to the small size of IP031 and IP169, it is considered that the effects will be negligible. IP047 includes a site allocated for open space which may offer biodiversity benefits in a localised town centre area. <i>Mitigation opportunities should be sought</i> to enhance the area allocated for an open space (IP047). (Note, this is not a new mitigation measure but carried over from the assessment of IP263 which has been merged with IP047)
ET9 To conserve and enhance the historic environment, heritage assets and their settings	IP031 0 IP169 0 IP047 0	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	There are no listing buildings located in the vicinity of IP031 and IP169. Part of IP031 falls within an Area of Archaeological Importance and adjacent to a Conservation Area in its most eastern parts. There are no Scheduled Monuments within or adjacent to the sites. Appropriate design of buildings should be considered in order to avoid any negative effects on the setting of the designated site.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
				IP047 falls within an Area of Archaeological Importance. Previous planning permissions have had a condition attached requiring archaeological investigation, which could include archaeological monitoring and recording of initial groundworks with contingency for fuller archaeological recording if deemed necessary; recording the remaining railway features; checking whether any trace of the dock tramway survives; and a palaeoenvironmental sampling strategy.
ET10 To conserve and enhance the quality and local distinctiveness of landscapes and townscapes	IP031 - IP169 - IP047 +	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	Potential negative effects on local distinctiveness and the conservation area at IP031 and IP169 are associated with deterioration of the waterfront townscape character. The townscape distinctiveness should be conserved through <i>the use of traditional</i> <i>or sympathetic building materials and</i> <i>techniques which respect those found on</i> <i>the surrounding buildings to maintain the</i> <i>local distinctiveness of the area.</i> Potential beneficial opportunities are identified at IP047 where at present the area is occupied by commercial / industrial buildings with low townscape value. New development may contribute to the achievement of the SA objective with appropriate design of buildings.
HW1 To improve the health of those most in need	IP031 + IP169 + IP047 +	Short, medium and long term Indirect Irreversible Medium	Localised area in Central Ipswich	Positive indirect effects are likely to occur with regards to the proximity of all the sites to Gippeswyk Park located to the west of the site allocations. The site allocations are also near designated river paths and close to community facilities which may encourage residents to lead a healthy lifestyle. Positive indirect effects are also likely to occur with regards to the potential use of open space and leisure provision (IP047).

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
HW2 To improve the quality of life where people live and encourage community participation	IP031 + IP169 + IP047 +	Short, medium and long term Direct / Indirect Reversible Medium	Central Ipswich	Community participation will be encouraged through the allocation of sites in a central urban area near a wide range of facilities (e.g. parks, public transport infrastructure, playing fields, etc.). The sites will also contribute to the achievement of the SA objective through the provision of decent housing thus improving the overall quality of life. In addition, leisure provision with IP047 may also contribute towards achieving the SA Objective.
ER1 To reduce poverty and social exclusion	IP031 + IP169 + IP047 +	Short, medium and long term Direct / Indirect Reversible Medium	Central Ipswich	Social exclusion will be reduced through the allocation of sites for residential uses near a wide range of community facilities easily accessible by different means of transport.
ER2 To offer everybody the opportunity for rewarding and satisfying employment	IP031 0 IP169 0 IP047 +	Short Direct Reversible Medium	Localised area in Central Ipswich	IP047 includes office, leisure and hotel development. These developments will create employment opportunities and will contribute towards supporting the SA Objective. The only employment opportunities likely to arise from the development of IP031 and IP169 are associated with temporary construction jobs. Therefore it is considered that the effects will be negligible.
ER3 To help meet the housing requirements for the whole community	IP031 + IP169 + IP047 + +	Short, medium and long term Direct Irreversible Medium	Central Ipswich	IP031, IP169, IP047 will contribute directly to the achievement of the SA objective through the delivery of new residential dwellings.
ER4 To achieve sustainable levels of prosperity and economic growth throughout the plan area	IP031 0 IP169 0 IP047 +	Short, medium and long term Direct/Indirect Irreversible Medium	N/A	The provision of office, leisure and hotel development with IP047 will help to provide employment opportunities which could contribute towards economic growth. Whilst new residential developments and will contribute to the achievement of the SA objective through improving the overall quality of life, it is considered that

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
				on its own each site will have a negligible effect.
ER5 To support vital and viable town, district and local centres	IP031 0 IP169 0 IP047 +	Short, medium and long term Direct/Indirect Irreversible Medium	Central Ipswich	The capacity of sites IP031 and IP169 do not suggest any significant change from the baseline; therefore it is considered that the impact on the vitality of the town centre will be negligible. There are no District or Local centres near IP047; however new residents and occupants at the site may contribute to the achievement of the SA by supporting town centre shopping areas.
ER6 To encourage efficient patterns of movement in support of economic growth	IP031 + IP169 + IP047 +	Short, medium and long term Direct/Indirect Irreversible Medium	Central Ipswich	The provision of residential dwellings, new offices, leisure provision and a hotel close to an existing employment area and a wide range of community facilities will encourage efficient patterns of movement and in the long term this could help to support economic growth.
ER7 To encourage and accommodate both indigenous and inward investment	IP031 0 IP169 0 IP047 +	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	On its own, IP031 is unlikely to contribute to the achievement of the SA objective as the focus is only on residential developments. IP047 could help to attract inward investment through the provision of offices and cumulatively, the sites will contribute to the regeneration / redevelopment of a central town area, with the provision of new decent housing and improve the business attractiveness of the borough.
CL1 To maintain and improve access to education and skills for both young people and adults	IP031 0 IP169 0 IP047 0	Short, medium and long term Direct Reversible Medium	Localised area in Central Ipswich	Whilst the close proximity of the sites to the University Campus Suffolk and Suffolk New College may result in educational benefits for the residents, it is considered that due to the small size of each site the effect will be negligible.
CD1 To minimise potential	IP031 0	Short, medium and long term	Localised area in Central Ipswich	On the whole, crime levels are higher in Ipswich than national average. It should

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
opportunities for crime and anti-social activity	IP169 0 IP047 -	Direct Irreversible Medium		also be noted that the sites are located in an area ranked as most deprived. In the short term, crime levels may increase as a result of the increase in the number of new residents. Therefore, <i>it would be beneficial to</i> <i>incorporate secured by design principles</i> <i>within new development</i> . <u>http://opendatacommunities.org/deprivati</u> <u>on/map</u>

South West – Gipping

IP083: Banks of the river upriver from Princess Street, open space IP015: West End Road Surface Car Park, 22 homes + employment use IP094: Land to rear of Grafton House, employment use

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET1 To improve air quality	IP083: + IP015: - IP094: -	Medium and long term Direct and indirect Irreversible Medium	Central Ipswich	IP083, IP015, and IP094 are near an AQMA designated due to air pollution along Vernon Street and Star Lane. IP094 is allocated for employment use and could potentially generate traffic in peak hours. Only 20% of sites IP083 and IP015 will be allocated for residential uses; therefore it is not considered that new residents will contribute significantly to increase in the traffic on the key roads in the area. However, a large part of IP015 is allocated for car parking which may result in additional traffic and deterioration of the air quality on the main roads. Positive effects on air quality are identified with the allocation of open space on the north bank of the river where opportunities could be sought to enhance the existing habitat. (IP083). The area is currently served by public transport and the use of sustainable modes of transport should be further encouraged through improved pedestrian routes and where appropriate the creation of cycling infrastructure.
ET2 To conserve soil resources and quality	IP083: + IP015: + IP094: +	Medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	The sites are located in urban Ipswich on previously developed land. Possible contamination is identified at the sites due to previous use of the land. Opportunities <i>should be sought to</i> <i>remediate the land before use</i> . Therefore it is considered that the effect in the medium and long term is likely to be beneficial.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET3 To reduce waste	IP083: 0 IP015: 0 IP094: -	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	Waste is likely to be generated in the short term during construction and in the long term during operation. Only 20% of IP083 and IP015 is allocated for residential use, therefore on their own the development of the sites is not anticipated to result in any significant negative effects. IP094 may have negative effects on waste if no mitigation measures are put in place. Cumulatively, negative effects are likely to occur when the site allocations are considered along with other sites in IP- one area. <i>Opportunities should be sought to</i> <i>encourage recycling and reuse of</i> <i>materials.</i>
ET4 To reduce the effects of traffic upon the environment	IP083: 0 IP015: - IP094: -	Short, medium and long term Direct Irreversible Medium	Central Ipswich	The relatively central location of the sites may result in reduced need to travel by private car. The sites are located near major employment areas to the north. The area is served by public transport. However, the purpose of IP015 (car parking) may encourage the use of private cars. For all these reasons it is considered that site allocations IP015 and IP094 are likely to detract from the achievement of the SA objective. A transport assessment and travel plan will be required for IP094. This will help to reduce potential negative effects on traffic.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET5 To improve access to key services for all sectors of the population	IP083: + IP015: + IP094: +/-	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	The sites are located in a central urban area of Ipswich near parks, sport facilities, river paths, Ipswich railway Station, with a good public transport access to the main shopping area to the north. Access constraints are identified only with IP094 due to potential congestion issues on A137. <i>Sustainable modes of transport should be encouraged through improved pedestrian infrastructure and/or where appropriate the creation of cycle routes/lanes.</i>
ET6 To limit and adapt to climate change	IP083: + IP015: - IP094: -	Short, medium and long term Direct Irreversible Medium	Central Ipswich	IP083, IP094 and IP015 are located in Flood Zones 2 or 3 prone to flooding from the river Orwell. Retaining IP083 for an open space will contribute to the achievement of the SA objective as less impervious surface area in the floodplain reduces runoff, which can decrease peak discharge for rainfall events and lower flood heights. <i>Flood risk assessment may be required</i> <i>at project assessment level to identify</i> <i>appropriate mitigation measures for the</i> <i>parts of the site allocations where</i> <i>housing may be built (e.g. through</i> <i>design).</i> Negative effects are likely to occur as a result of the allocation of site IP015 for car parking uses which may increase the use of private cars. <i>Sustainable modes of transport should</i> <i>be encouraged through improved</i> <i>pedestrian infrastructure and/or where</i> <i>appropriate the creation of cycle</i> <i>routes/lanes.</i>

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET7 To protect and enhance the quality of water features and resources and reduce the risk of flooding	IP083: + IP015: +/- IP094: +/-	Short, medium and long term Direct Irreversible Medium	Central Ipswich	Retaining IP083 for an open space will contribute to the achievement of the SA objective as less impervious surface area in the floodplain reduces runoff, which can decrease peak discharge for rainfall events and lower flood heights. Water quality could be enhanced if opportunities are sought to remediate the contaminated land before future use of the land and enhancement of the open space (IP083). Surface water flooding local to site – will be considered at planning application stage for IP094. See Appendix 1 of the Ipswich SFRA. <i>New developments should be</i>
				encouraged to use SuDS to manage runoff, reduce further flood risk and protect water quality.
ET8 To conserve and enhance biodiversity and geodiversity, including favourable conditions on SSSIs, SPAs and SACs	IP083: + IP015: 0 IP094: -	Short, medium and long term Direct/Indirect Irreversible Medium	Localised area in Central Ipswich	. IP015 and IP083 are adjacent to a County Wildlife Site (the river). Potential direct positive effects are likely to occur at IP083 where the majority of land will be retained as open space or vegetation supporting wildlife habitats. Consequently, indirect positive effects may occur with regards to air quality, water quality, and soil resources. Potential negative effects at IP015 are associated with noise disturbance; however these effects will be temporary and on the whole negligible. One TPO is identified at IP094 and <i>application for</i> <i>Tree Works may be required</i> .
ET9 To conserve and enhance the historic environment, heritage assets and their settings	IP083: 0 IP015: - IP094: 0	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	None of the sites fall within an Area of Archaeological Importance or Conservation Area. There is one listed building adjacent to IP015 (Paul's Maltings - Grade II) and negative effects may occur if housing or multi storey car parking is developed in the eastern part of the land. Appropriate design of residential buildings should be considered in order to avoid any negative effects on the setting of the designated site.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET10 To conserve and enhance the quality and local distinctiveness of landscapes and townscapes	IP083: + IP015: 0 IP094: +	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	Potential positive effects on local distinctiveness are associated with improved waterfront landscape /townscape through the retention of open space and enhancement of the river path and the development of a vacant site bounded by hoardings at IP094.
HW1 To improve the health of those most in need	IP083: + IP015: - IP094: 0	Short, medium and long term Indirect Irreversible Medium	Localised area in Central Ipswich	Positive indirect effects are likely to occur as a result of the retention of 80% of IP083 as an open space and the enhancement of the river path in this section. The provision of a cycle path through the southern part of IP083 as part of a continuous path along the River Gipping would also encourage healthy lifestyles. Residents (IP083/IP015) may also benefit from the close location of sport facilities at the Ipswich Town Football Club Stadium and playing fields located to the north of the site allocations. The provision of a new car park may result in indirect negative effects on health if people are discouraged to choose walking or cycling as a mode of transport.
HW2 To improve the quality of life where people live and encourage community participation	IP083: + IP015: + IP094: +	Short, medium and long term Direct / Indirect Reversible Medium	Central Ipswich	Community participation will be encouraged through the allocation of sites in a central urban area near a wide range of facilities (e.g. parks, public transport infrastructure, Ipswich Town Football Club, etc.). The sites will also contribute to the achievement of the SA objective through the provision of decent housing and open space thus improving the overall quality of life.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ER1 To reduce poverty and social exclusion	IP083: + IP015: + IP094: +	Short, medium and long term Direct / Indirect Reversible Medium	Central Ipswich	Social exclusion will be reduced through the allocation of sites for residential uses near a wide range of community facilities easily accessible by different means of transport. Some positive effects may occur with the allocation of land for employment use and thus improving the employment opportunities in the borough.
ER2 To offer everybody the opportunity for rewarding and satisfying employment	IP083: 0 IP015: 0 IP094: +	Short Direct Reversible Medium	Localised area in Central Ipswich	IP094 will contribute to the achievement of the SA objective through the allocation of land for employment use (B1 Class). It is anticipated that IP083 and IP015 will have a negligible effect due to only temporary short term employment opportunity during construction of residential units.
ER3 To help meet the housing requirements for the whole community	IP083: + IP015: + IP094: 0	Short, medium and long term Direct Irreversible Medium	Central Ipswich	IP083 and IP015 will contribute to the achievement of the SA objective through the delivery of residential dwellings.
ER4 To achieve sustainable levels of prosperity and economic growth throughout the plan area	IP083: + IP015: + IP094: +	Short, medium and long term Direct/Indirect Irreversible Medium	N/A	New residential development (IP083 and IP015) and new employment opportunities (IP094) will contribute to the achievement of the SA objective.
ER5 To support vital and viable town, district and local centres	IP083: 0 IP015: 0 IP094: 0	Short, medium and long term Direct/Indirect Irreversible Medium	N/A	There is no clear relationship between the allocation of the sites and the achievement of the SA objective and/or the effect will be negligible.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ER6 To encourage efficient patterns of movement in support of economic growth	IP083: + IP015: + IP094: +	Short, medium and long term Direct/Indirect Irreversible Medium	Central Ipswich	The provision of residential dwellings close to existing employment areas and a wide range of community facilities will encourage efficient patterns of movement and in the long term this could help to support economic growth.
ER7 To encourage and accommodate both indigenous and inward investment	IP083: + IP015: + IP094: +	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	The sites will contribute to the regeneration / redevelopment of a central town area, with the provision of new decent housing, office buildings, open space and car parking facilities and thus improving the business attractiveness of the town centre. IP083 and IP015 in particular could improve a key gateway into the town centre from the railway station.
CL1 To maintain and improve access to education and skills for both young people and adults	IP083: 0 IP015: 0 IP094: 0	Short, medium and long term Direct Reversible Medium	Localised area in Central Ipswich	Whilst the close proximity of the sites to the University Campus Suffolk and Suffolk New College may result in educational benefits for the residents, it is considered that due to the small proportion of housing development of IP083 and IP015 the effect will be negligible.
CD1 To minimise potential opportunities for crime and anti-social activity	IP083: 0 IP015: 0 IP094: 0	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	On the whole, crime levels are higher in Ipswich than national average. It should also be noted that the sites are located in an area ranked as relatively deprived. Crime levels may increase in the vicinity of the new multi-storey car park. Therefore, <i>it would be beneficial to incorporate secured by design principles within new development</i> . <u>http://opendatacommunities.org/deprivati</u> on/map
South West – Gipping

IP004: Bus depot Sir Alf Ramsey Way, 48homes + 50% employment use IP051: Old Cattle Market Portman Road, 80% B1 development + 20% leisure IP096: Car park Handford Road East, 20 homes IP006: Coop Warehouse, Pauls Road, 28 homes

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET1 To improve air quality	IP004 + IP051 + IP096 + IP006 +	Medium and long term Direct and indirect Irreversible Medium	Central Ipswich	None of the site allocations are located near or in an AQMA. The indicative capacity of the IP004, IP051 and IP096 suggest that traffic may be generated in the vicinity of Sir Alf Ramsey Way particularly during peak hours. The area is served by public transport (two bus routes running on Handford Road) which may reduce the need to travel by private car. In addition the close location of a number of employment sites provides an opportunity to encourage the implementation of car-share schemes. <i>Sustainable modes of transport should be further encouraged through improved pedestrian routes in the area and/or through provision of cycling facilities at major employment hubs.</i>
ET2 To conserve soil resources and quality	IP004 + IP051 + IP096 + IP006 +	Medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	The sites are located in urban Ipswich, on brownfield land and are likely to be contaminated due to previous use of the land and opportunities <i>could be sought</i> <i>to remediate the land before use.</i> Therefore it is considered that the effect in the medium and long term is likely to be beneficial.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET3 To reduce waste	IP004 - IP051 - IP096 0 IP006 0	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	Waste is likely to be generated in the short term during construction and in the long term during operation (household waste + office waste). The indicative capacity of IP004 and IP051 suggests potential for negative effects if mitigation measures are not in place. Cumulatively, negative effects are likely to occur when the site allocations are considered along with other sites in IP- one area. <i>Opportunities should be sought to</i> <i>encourage recycling and reuse of</i> <i>materials.</i>
ET4 To reduce the effects of traffic upon the environment	IP004 + IP051 - / ? IP096 + IP006 +	Short, medium and long term Direct Irreversible Medium	Central Ipswich	The central location of the sites may result in reduced need to travel by private car. The area is well served by public transport: Tower Ramparts Bus Station and Old Cattle Market Bus Station are located to the east, Ipswich Railway Station is located to the south, and two local bus lines connect the north bank with residential areas and facilities on the south bank. For all these reasons it is considered that the sites are likely to contribute to the achievement of the SA objective. <i>A separate transport assessment may be required for IP051</i> due its size and close proximity to Ipswich Town Football Club stadium which may result in potential negative effects on traffic particularly during major sport events. A transport assessment and travel plan will be required for IP004. This will help to reduce potential negative effects on traffic.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET5 To improve access to key services for all sectors of the population	IP004 + IP051 + IP096 + IP006 +	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	All the sites are located in the central urban area of Ipswich near shopping areas and community facilities. No major access constraints are identified. Sustainable modes of transport should be encouraged through improved pedestrian infrastructure and/or where appropriate the creation of cycle routes/lanes particularly around the vicinity of the office buildings.
ET6 To limit and adapt to climate change	IP004 - IP051 - IP096 0 /- IP006 0	Short, medium and long term Direct Irreversible Medium	Central Ipswich	The central location of the sites may encourage the use of more sustainable modes of transport, particularly due to the fact that most community facilities are within walking distance. IP004 and IP051 are entirely located in Flood Zones 2 and 3 with a risk of flooding from the river Orwell. Part of IP096 is also located in a zone prone to flooding. <i>New developments should be</i> <i>encouraged to use SuDS to manage</i> <i>runoff, reduce further flood risk and</i> <i>protect water quality. Flood risk</i> <i>assessment may be required at project</i> <i>assessment level to identify appropriate</i> <i>mitigation measures for the parts of the</i> <i>development that fall within Flood Zone 2</i> <i>and 3 (e.g. through design).</i> <i>Cumulatively, the effects from climate</i> <i>change can be reduced through</i> <i>encouraging people to use more</i> <i>sustainable modes of transport.</i>

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET7 To protect and enhance the quality of water features and resources and reduce the risk of flooding	IP004 - IP051 - IP096 0 /- IP006 0	Short, medium and long term Direct Irreversible Medium	Central Ipswich	IP004 and IP051 are entirely located in Flood Zones 2 and 3 with a risk of flooding from the river Orwell. Part of IP096 is also located in a zone prone to flooding. Water quality could be enhanced if opportunities are sought to remediate the contaminated land where appropriate. Surface water flooding local to site – will be considered at planning application stage for IP004, IP006 and IP006. See Appendix 1 of the Ipswich SFRA. <i>Flood risk assessment may be required</i> <i>at project assessment level to identify</i> <i>appropriate mitigation measures for the</i> <i>parts of the development that falls within</i> <i>Flood Zone 2 and 3 (e.g. through</i> <i>design).</i>
ET8 To conserve and enhance biodiversity and geodiversity, including favourable conditions on SSSIs, SPAs and SACs	IP004 - IP051 0 IP096 - IP006 0	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	IP096 and IP004 are located near a local nature reserve. IP004 is also close to CWS. Negative effects are likely to occur during construction with regards to dust and air pollution, noise and disturbance of species. <i>Mitigation measures should</i> <i>be put in place to ensure any potential</i> <i>impacts upon the local nature reserve</i> <i>are avoided or minimised.</i>
ET9 To conserve and enhance the historic environment, heritage assets and their settings	IP004 + IP051 0 IP096 0 IP006 0	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	There are no designated sites located in or near the site allocations and none of the sites fall within an Area of Archaeological Importance/Conservation area. However the allocation of IP004 helps to retain and conserve the locally listed tram shed on the site.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ET10 To conserve and enhance the quality and local distinctiveness of landscapes and townscapes	IP004 0 IP051 0 IP096 0 IP006 0	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	Whilst none of the site allocations is anticipated to have any negative effects on the local distinctiveness of townscape, opportunities should be sought to enhance the quality of the townscape through appropriate design of office buildings ((IP051) and residential dwellings (IP004, IP096, and IP006).
HW1 To improve the health of those most in need	IP004 + IP051 + IP096 + IP006 -	Short, medium and long term Indirect Irreversible Medium	Localised area in Central Ipswich	Whilst the provision of decent housing and employment opportunities may have some positive indirect effects on health, it is considered that the effect from the development of each site will be negligible. Some negative effects are likely to occur with the development of IP006 as result of the location of rail lines adjacent to the site and potential noise disturbance. Positive indirect positive effects are likely to occur with regards to the use of the playing grounds and sport facilities near IP004, IP051, and IP096. The sites are also near designated river paths and close to community facilities (restaurants, cinemas, churches) which may encourage more people to choose walking as the main mode of transport.
HW2 To improve the quality of life where people live and encourage community participation	IP004 + IP051 + IP096 + IP006 +	Short, medium and long term Direct / Indirect Reversible Medium	Central Ipswich	Community participation will be encouraged through the allocation of sites in the central urban area near a wide range of facilities (e.g. parks, public transport infrastructure, playing grounds, etc.). The sites will also contribute to the achievement of the SA objective through the provision of decent housing and employment opportunities and thus improving the overall quality of life.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ER1 To reduce poverty and social exclusion	IP004 + IP051 + IP096 + IP006 +	Short, medium and long term Direct / Indirect Reversible Medium	Central Ipswich	Social exclusion will be reduced through the allocation of sites for residential/business uses near a wide range of community facilities easily accessible by different means of transport. The main benefits are associated with the existing open space at Alderman Park and playing fields at Ipswich Town Football Club Stadium.
ER2 To offer everybody the opportunity for rewarding and satisfying employment	IP004 + IP051 + IP096 0 IP006 0	Short, medium, and long term Direct Reversible Medium	Localised area in Central Ipswich	The only sites with potential to contribute to the achievement of the SA objective is IP051 allocated for B1 business use and IP004 allocated for mixed use. IP096 and IP006 are allocated for residential use and employment opportunities will be temporary during construction of new development.
ER3 To help meet the housing requirements for the whole community	IP004 + IP051 0 IP096 + IP006 +	Short, medium and long term Direct Irreversible Medium	Central Ipswich	IP004, IP096, IP006 will contribute directly to the achievement of the SA objective through the delivery of residential dwellings.
ER4 To achieve sustainable levels of prosperity and economic growth throughout the plan area	IP004 + IP051 + IP096 0 IP006 0	Short, medium and long term Direct/Indirect Irreversible Medium	N/A	Whilst new residential developments and a new office building (IP051 and IP004) will contribute to the achievement of the SA objective through improving the overall quality of life, it is considered that on its own each site will have a negligible effect.
ER5 To support vital and viable town, district and local centres	IP004 + IP051 + IP096 + IP006 +	Short, medium and long term Direct/Indirect Irreversible Medium	Localised area in Central Ipswich	There are no District or Local centres in the vicinity of the sites. However, the site allocations will contribute to the achievement of the SA objective through regeneration of a central town area and influx of new residents/employees.

Objective	Performance of Policy	Temporal Scale Nature of Impact (Direct/Indirect) Reversibility Certainty	Geographical Extent	Commentary/Mitigation
ER6 To encourage efficient patterns of movement in support of economic growth	IP004 + IP051 + IP096 + IP006 +	Short, medium and long term Direct/Indirect Irreversible Medium	Central Ipswich	The provision of residential and office buildings close to existing/future employment hubs, shopping areas, and community facilities will encourage efficient patterns of movement and in the long term this could help to support economic growth.
ER7 To encourage and accommodate both indigenous and inward investment	IP004 + IP051 + IP096 0 IP006 0	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	Cumulatively, the sites will contribute to the regeneration / redevelopment of central town areas. IP051 will contribute directly to the achievement of the SA objective through the provision of office/hotel/leisure space in a central part of Ipswich next to existing employment sites. IP004 includes business uses as part of a mixed use allocation.
CL1 To maintain and improve access to education and skills for both young people and adults	IP004 0 IP051 0 IP096 0 IP006 0	Short, medium and long term Direct Reversible Medium	Localised area in Central Ipswich	Whilst the close proximity of the sites to the University Campus Suffolk and Suffolk New College may result in educational benefits for the residents, it is considered that due to the relatively small size of each site the effect will be negligible.
CD1 To minimise potential opportunities for crime and anti-social activity	IP004 - IP051 - IP096 0 IP006 0	Short, medium and long term Direct Irreversible Medium	Localised area in Central Ipswich	Whilst the size of the sites does not suggest any significant effect on the baseline conditions, it should be noted that the sites are located in an area ranked as most deprived. On the whole, crime levels are higher in Ipswich than national average and the close proximity of Ipswich Town Football Club has potential to result in anti-social activities during sport events. Therefore, <i>it would be beneficial to incorporate secured by design principles within new developments.</i> <u>http://opendatacommunities.org/deprivati</u> on/map

Appendix B

Pre-Submission Additional Modifications Table

Policy / paragraph	Additional Modification	Reason	Objector(s)	Significance to the SA
Throughout	Change references to English Heritage to Historic England.	To reflect change in title.	Editorial update.	Reference to English Heritage will be replaced with Historic England throughout the SA.
Throughout	Changes references to the Highways Agency to Highways England.	To reflect change in title.	Editorial update.	The SA has been reviewed and does not contain reference to the Highways Agency therefore no further action is required.
CHAPTER 3		1	1	
New paragraph 3.6 to follow paragraph 3.5	Add new paragraph as follows: 'In July 2015 a non-statutory document entitled 'The Vision for Ipswich: East Anglia's Waterfront Town' was published by partners – University Campus Suffolk, New Anglia Local Enterprise Partnership, Suffolk County Council, Ipswich Central, Ipswich Borough Council, Ben Gummer MP and the Ipswich Chamber of Commerce. This Vision brings together the aspirations of the partners on a range of issues and identifies a series of actions for the next few years. Some of these are relevant to the Local Plan and others are not, because they relate to matters beyond the remit of the planning system (e.g. starting works on the I-Am Project around the Museum on High Street). The two documents (the statutory 'Local Plan' and the 'Vision for Ipswich') are considered to complement each other	For completeness	Editorial update.	The update provides completeness within the supporting text and does not change the SA.
CHAPTER 4	in a helpful way.'			
-	CATED FOR HOUSING / MIXED USE			
SP2 / para 4.8	Split the final sentence from paragraph 4.8 and create a new paragraph as follows:	To add weight to the site sheets and ensure the constraints they identify are taken into account. To	Historic England	The additional text updates ensures constraints in the site sheets are taken into account and updates the situation in relation to the Urban Archaeological

Policy /	Additional Modification	Reason	Objector(s)	Significance to the SA
paragraph				Detabage This days not sharped the OA
	'Appendix 3 provides additional	update the situation in relation to the Urban		Database. This does not change the SA.
	information about the sites allocated	Archaeological Database.		
	through this policy. Information on	Archaeological Dalabase.		
	development constraints contained in			
	the site sheets must be taken into			
	account in development proposals, in			
	accordance with the Ipswich Local			
	Plan's Development Management			
	policies. The Council is working with			
	Historic England and Suffolk County			
	Council to establish an Urban			
	Archaeological Database for Ipswich,			
	to make available desk-based			
	information on known heritage			
	<u>assets.'</u>			
SP2 / para 4.9	Amend paragraph 4.9 to read:	For clarity and	Environment Agency	The amended text provides clarity to the
		completeness.		supporting text regarding flooding and
	<u>'In allocating sites for development</u>			does not change the SA of the policy.
	the Council has followed the			
	sequential approach, to ensure that			
	sites are not allocated in areas with a			
	greater probability of flooding if sites			
	in lower risk areas are available. It			
	has also applied the exception test to			
	ensure that the benefits to the			
	community of development outweigh			
	flood risk, and ensure that			
	development will be safe. Planning			
	applications Risk Assessment.			
	The Council's supplementary			
	planning document on Development			
	and Flood Risk provides more			
	guidance.'			

Policy /	Additional Modification	Reason	Objector(s)	Significance to the SA
paragraph SP2/para 4.10	Amend reference to Lifetime Homes and Lifetime Neighbourhoods: 'Housing is a key issue for older people. There is a need to ensure a suitable mix of housing for older people, whether market housing or types of supported housing. The Council encourages new housing to be built to <u>be accessible and</u> <u>adaptable</u> the Lifetime Homes standard, which makes it easier for people to remain in their own homes as their mobility needs change. In order to create an environment which enables older people to have a good quality of life, the Council supports the implementation of the Government's 'Lifetime Neighbourhoods' principle in new development. This concept covers the built environment, access to services and resident empowerment in addition to housing standards mix <u>and design</u> .	To reflect the fact that new, optional standards have been introduced for accessible and adaptable dwellings and wheelchair user dwellings, which have led to amendments to policy DM5 of the Core Strategy.	Editorial update	The amendment text reflects the fact that new, optional standards have been introduced for accessible and adaptable dwellings and wheelchair user dwellings, which have led to amendments to policy DM5 of the Core Strategy. This does not change the SA on the policy.
Various sites IP052, IP133, IP140, IP258	Amendments to site sheets: IP052 – Amend first sentence: 'Within an Air Quality Management Area and an area of archaeological importance and adjacent to a conservation area, anda Grade II* listed building to the north (24 Fore	For clarity.	Historic England	The corresponding SA assessments against ET9 have been updated to reflect to these additions, although the overall result has not changed. Reference to IP258s proximity to an AQMA is already included within the assessment.

Policy / paragraph	Additional Modification	Reason	Objector(s)	Significance to the SA
	Street) <u>and a Grade II building (26-28</u> Fore St).'			
	IP133 – Add to the end of the first paragraph of the Development constraints section the following: ' <u>Site is adjacent to the Wet Dock</u> <u>conservation area and to the Grade II</u> <u>listed Felaw Maltings.</u> '			
	IP140 – Add to the beginning of the second paragraph the following: ' <u>The site is adjacent to the Whitton</u> <u>conservation area and any</u> <u>development should have regard to</u> <u>the setting of the conservation area</u> <u>and conserve its significance.</u> '			
	IP258 – Add to the end of the second sentence of the first paragraph the following: 'Adjacent to an Air Quality Management Area <u>, and</u> the Central conservation area <u>and the Grade II</u> <u>listed Holy Trinity Church to the</u> <u>south.'</u>			
Various sites IP10a, IP029, IP033, IP058,	Amend site sheets to add reference to need for ecological and/or reptile survey to site sheet where necessary.	For clarity.	Suffolk Wildlife Trust	The modification provides clarity to the site sheet on the need for ecological and reptile survey and does not change the

Policy /	Additional Modification	Reason	Objector(s)	Significance to the SA
paragraph IP067, IP083, IP140a, IP0150b and c	IP10a, IP067: ' <u>An ecological and</u> reptile survey will be needed prior to any vegetation clearance and mitigation where appropriate.' IP029, IP033, IP058, IP083, IP140 and IP150b & c: 'an ecological and			SA of the sites.
Various sites IP005, IP029, IP032, IP033, IP059a, IP061, IP105, IP140, IP165, IP175, IP221, IP265,	reptile survey will be needed' Add note to site sheet of relevant allocations: 'This site falls within the 91.4m height consultation zones surrounding Wattisham airfield, therefore any proposed structures which may exceed this height would need to be	For clarity	MoD	The modification provides clarity of the sites that fall within the 91.4m height consultation zones surrounding Wattisham airfield and does not change the SA of the sites.
IP261 Various sites IP005 IP010a&b IP011b IP029 IP032 IP040 IP043 IP048 IP054 IP059a IP165 IP004 and IP033	reviewed by the Ministry of Defence.'Add Suffolk County Councilinformation to site sheets aboutpotential transport requirements:A transport assessment and travelplan will be required.A transport assessment will berequired.	For completeness	Suffolk County Council	The modification is for completeness in relation to the requirement for transport assessments and travel plans. Reference has been added within the assessment of the sites against SA Objective ET4. The overall result of the SA has not changed.
IP032	A transport assessment and travel plan will be required. The traffic impact of access from Bury Road will need to be considered.			
IP033	A transport assessment and travel plan will be required. Access visibility and junction spacing along Bramford Road will need to be considered.			
IP037	A transport assessment and travel plan will be required. Traffic impact especially on Bridge Street will need			

Policy / paragraph	Additional Modification	Reason	Objector(s)	Significance to the SA
	to be considered.			
IP136	A transport assessment and travel plan will be required . Pedestrian capacity along College St footways, and the one-way traffic system will need to be considered.			
Various sites IP004 IP005 IP006 IP009 IP011b IP012 IP032 IP033 IP039a IP040 IP043 IP048 IP054 IP059a IP066 IP080 IP089 IP096 IP088 IP131 IP133 IP136 IP142 IP165 IP188 IP214 IP245 IP256	Add Suffolk County Council information to site sheets on surface water flooding: <u>'Surface water flooding local to site -</u> will need to be considered at planning application stage. See Appendix 1 of the Ipswich SFRA.' Also add to site sheets for IP136 and IP245: <u>'There is a Flood Incident Report for this site.'</u>	For completeness.	Suffolk County Council	The modification is for completeness providing reference to considering surface flood water local to the sites and has been referenced within the assessment of each site against SA Objective ET7. It does not change the overall scores.
IP004 Bus depot	The site sheet should mention proximity to a permitted minerals/waste use as a constraint, through the inclusion of the following text: ' <u>The site is close to uses which are safeguarded through the Suffolk</u> <u>Waste Core Strategy, and is also in close proximity to existing minerals</u> uses.'	For accuracy.	Suffolk County Council	The modification provides accuracy regarding proximity to minerals uses. It does not change the SA of the site.
IP006 Co-Op Warehouse Pauls Road	Flag up Co-op Juniors issue on the site sheet: ' <u>The Council would wish to see the</u> <u>Co-Op Juniors accommodated</u> <u>through refurbishment of the building,</u> <u>or relocated to suitable premises</u> <u>elsewhere in Ipswich</u> .'	For clarity.	Paul Lofts (Co-op Juniors)	The modification provides clarity regarding Co-Op Juniors and does not change the SA of the site.

Policy / paragraph	Additional Modification	Reason	Objector(s)	Significance to the SA
IP012 Peter's Ice Cream	Amend penultimate paragraph to read: 'In terms of archaeology, This site lies in the historic core of Anglo-Saxon and medieval Ipswich, and could involve potentially high excavation costs. <u>Necessary measures for</u> <u>archaeology should be addressed at</u> <u>an appropriate stage in the planning</u> <u>process.</u> There is no objection in principle to development but any <u>permission will require a condition</u> <u>relating to archaeological</u> <u>investigation.</u> <u>Early evaluation is</u> <u>advised so that decisions can be</u> <u>taken on preservation and/or</u> <u>appropriate investigation strategies</u> <u>designed.</u> Archaeological costs have the potential to be relatively high.'	For clarity.	Suffolk County Council	The modification provides clarity regarding measures for archaeology and has been referenced within the assessment of the site against SA Objective ET9. It does not change the overall score.
IP037 Island Site	Site IP037 needs to recognise the mineral handling facilities at the Port which are protected through the Minerals Plan and DM26 and are part of the delivery of a wider marine plan, through the inclusion of the following text on the site sheet: <u>'The site is close to uses which are safeguarded through the Suffolk</u> Minerals Core Strategy.'	For clarity and compliance with the Suffolk Minerals Core Strategy	Suffolk County Council	The modification provides clarity regarding proximity to minerals uses. It does not change the SA of the site.
IP040 Civic Centre Area / Civic Drive	Amend final paragraph of site sheet to read: 'This site lies in the historic core of Anglo-Saxon and medieval Ipswich. <u>Measures for archaeology should be</u> <u>addressed at an appropriate stage in</u> <u>the planning process. There is no</u>	For clarity	Suffolk County Council	The modification provides clarity and has been referenced within the assessment of the site against SA Objective ET9. However, the overall result has not changed.

Policy /	Additional Modification	Reason	Objector(s)	Significance to the SA
paragraph				
	objection in principle to development,			
	but any permission will require a			
	condition relating to archaeological			
	investigation. Early evaluation is			
	advised so that decisions can be			
	taken on preservation and/or			
	appropriate investigation strategies			
	designed. Archaeological costs have			
	the potential to be relatively high.'			
IP043 Commercial	Amend sections of site sheet to read:	For clarity.	Historic England &	The modification provides clarity and
Buildings and			Suffolk County	has been referenced within the
Jewish Burial	'This is a sensitive site in heritage		Council	assessment of the site against SA
Ground, Star Lane	terms, located partly in the Central			Objective ET9. However, the overall
	conservation area and containing			result has not changed.
	Grade II listed buildings and adjoining			
	others, including the Grade II* 54-58			
	Fore Street to the north-east, the			
	Grade II* Old Custom House to the			
	south-west and the complex of highly			
	graded buildings at Isaac Lord to the			
	south-east. Archaeological issues			
	include the Jewish Burial Ground.			
	The site presents opportunities for			
	enhancing the Jewish burial ground			
	which needs to be carefully respected			
	by any development proposal.			
	This site is within the urban core (IPS			
	413) and previous archaeological			
	evaluation and limited excavation has			
	revealed Anglo-Saxon and Medieval			
	remains in particular (IPS 639,			
	371,372, 358). There is outstanding			
	post-excavation work under			
	IP/11/00267) and further work would			
	be needed across the site.			
	Development of the site could involve			
	potentially high excavation costs.			
	The site presents opportunities for			

Policy /	Additional Modification	Reason	Objector(s)	Significance to the SA
paragraph				
	enhancing the Jewish Burial Ground.			
	There is no objection in principle to			
	development but any permission will			
	require a condition relating to			
	archaeological investigation. Detailed			
	early pre-application discussions with			
	Suffolk County Council			
	Archaeological Service and Historic			
	England would be required in order to			
	agree the scope of required			
	assessment, the principle of			
	development and to inform design.			
	Archaeological costs have the			
	potential to be relatively high.'			
IP048 Mint Quarter	Amend second paragraph of	For clarity.	Historic England &	The modification provides clarity and
Cox Lane	Constraints section as follows:	i or oranty:	Suffolk County	has been referenced within the
Cox Lano	Close to an Air Quality Management		Council	assessment of the site against SA
	Area and in an area of archaeological			Objective ET9. However, the overall
	importance. <u>The site is</u> adjacent to			result has not changed.
	the Central conservation area and			result has not shanged.
	two Grade II listed churches (Christ			
	Church and St Pancras) and other			
	listed buildings to the south. Any			
	proposals would need to consider the			
	impact of development upon			
	designated and non-designated			
	heritage assets and their setting,			
	including any resulting benefit, harm			
	or loss to their significance.' Possible			
	contamination and TPOS'			
	Amond fourth norganization of follows:			
	Amend fourth paragraph as follows:			
	'Much of this site is a Scheduled			
	Monument (List entry No 1005983),			
	therefore separate Scheduled			
	Monument Consent will need to be			
	obtained in addition to planning			

Policy /	Additional Modification	Reason	Objector(s)	Significance to the SA
paragraph				
	permission from Historic England and			
	they should be consulted at the			
	earliest opportunity. The scheduled			
	area represents a large portion of the			
	Anglo-Saxon and medieval town,			
	preserved under current car parks.			
	There would be extremely high			
	archaeological costs associated with			
	development. Due to the high			
	potential for archaeological remains			
	of national significance, detailed early			
	pre-application discussions with			
	Suffolk County Council			
	Archaeological Service and Historic			
	England would be required in order to			
	agree the principle of development			
	and inform designs (e.g. to allow for			
	preservation in-situ of deposits or			
	appropriate programmes of work).			
	Excavations and interventions have			
	taken place in parts of the site and			
	revealed evidence for occupation and			
	activity from the Middle Saxon period			
	onwards. The rest of the site is			
	undisturbed from modern			
	development under car parks and is			
	anticipated to contain rich and well			
	preserved archaeological remains.			
	Development would require full			
	assessment prior to the granting of			
	<u>consent to any proposals – DBAdesk</u>			
	based assessment, building survey			
	and field evaluation. As noted in			
	Policy CS4, English Heritage consent			
	would be needed for any			
	development on scheduled sites.			
	Should the principle of any			
	development be acceptable,			
	archaeological remains will be			

Policy /	Additional Modification	Reason	Objector(s)	Significance to the SA
paragraph	complex, important and involve significant costs. Archaeological management would be a major consideration and should to involve upfront assessment to inform design (e.g. to allow for preservation in situ of deposits or appropriate programmes of work).'			
IP049 No. 8 Shed, Orwell Quay	Delete Archaeology will need to be considered and add paragraph dealing with archaeology: 'This site has potential for evidence relating to exploitation of the foreshore from the Middle Saxon period onwards, as well as potential for remains relating to medieval and post-medieval shipyards. Buildings on the site should be assessed, and it will require a condition relating to archaeological investigation attached to any planning consent.'	For clarity	Suffolk County Council	The modification provides clarity regarding archaeological potential and does not change the SA of the site.
IP054 Land between Old Cattle Market and Star Ln	Add this wording to the end of the fourth para of Development constraints/issues: ' <u>The site is partly within the Central</u> <u>Conservation Area and contains two</u> <u>Grade II listed buildings (18-20 Lower</u> <u>Brook Street). The site is flanked by</u> <u>the Central conservation area and</u> <u>several listed buildings, with two</u> <u>Grade II* churches to the south.</u> ' Amend final paragraph as follows: 'This site includes two scheduled areas of the Anglo-Saxon and	For clarity.	Historic England & Suffolk County Council	The modification provides clarity and has been referenced within the assessment of the site against SA Objective ET9. However, the overall result has not changed.

Policy / paragraph	Additional Modification	Reason	Objector(s)	Significance to the SA
pulugiupii	medieval town of Ipswich (SF 192 a			
	and b), which are statutorily protected			
	as of National Importance. Parts of			
	the area have been investigated (IPS			
	214, IPS 369, IPS 574). ThisThe			
	latter found a wood-lined well with an			
	assemblage of boar tusks,			
	demonstrating good potential for the			
	survival of wet and well preserved			
	organic deposits. As noted in Policy			
	CS4, English Heritage Scheduled			
	Monument consent wouldwill be			
	needed to be obtained from Historic			
	England for any development			
	onworks within scheduled			
	sitesmonuments and they should be			
	consulted at the earliest opportunity.			
	Detailed pre-application discussions			
	with Suffolk County Council			
	Archaeological Service and Historic			
	England would be required in order to			
	agree the principle of development			
	and inform design (e.g. to allow			
	preservation in-situ of deposits or			
	appropriate programmes of work).			
	Where development is accepted in			
	principle, archaeological remains will			
	be complex, important and involve			
	significant costs. Archaeological			
	management will be a major			
	consideration and should involve			
	early assessment to inform design (e.g. to allow for preservation in situ			
	of deposits or appropriate			
	programmes of work). The route of			
	Turret lane should be protected in			
	development proposals. Outside the			
	scheduled areas, there is no objection			
	in principle to development but any			

Policy / paragraph	Additional Modification	Reason	Objector(s)	Significance to the SA
Paragrap	permission will require a condition relating to archaeological investigation. Archaeological costs have the potential to be relatively high.'			
IP058 Former Volvo Site, Raeburn Road	Amend penultimate paragraph as follows: 'This site overlooks the Orwell and in relation to archaeology is in a topographically favourable location in the vicinity of prehistoric sites, particularly of Palaeolithic date. There is no objection in principle to development but any permission will require a condition relating to archaeological investigation. Measures for archaeology should be addressed at an appropriate stage in the planning process. A desk-based assessment is recommended in the first instance.'	For clarity	Suffolk County Council	The modification provides clarity regarding measures for archaeology and does not change the SA of the site.
IP080 Wherstead Road	Site IP080 needs to recognise the mineral handling facilities at the Port which are protected through the Minerals Plan and DM26 and are part of the delivery of a wider marine plan, through the inclusion of the following text in the site sheet: <u>'The site is close to uses which are safeguarded through the Suffolk</u> Minerals Core Strategy.'	For clarity.	Suffolk County Council	The modification provides clarity regarding proximity to minerals uses. It does not change the SA of the site.
IP083 Banks of river upriver from Princes Street	Add archaeological information to site sheet: <u>'Depending on the nature of</u> groundworks, there may be a requirement for a condition relating to	For clarity and to safeguard biodiversity	Suffolk County Council Suffolk Wildlife Trust	The modification provides clarity regarding the potential requirement for a condition relating to archaeological works and avoiding light spillage into the river. It does not change the SA of the site.

Policy / paragraph	Additional Modification	Reason	Objector(s)	Significance to the SA
	archaeological works on this land on the edge of the river.'			
	Also add to site sheet: ' <u>The design will need to avoid light</u> spillage in the river corridor.'			
IP089 Waterworks Street	Need to refer on site sheet to the Old Ragged School building. Add text: ' <u>Contains fragment of the Ipswich</u> <u>Ragged School, founded 1849.</u> ' Also, the boundary of site needs amending slightly to follow building lines.	For clarity.	Editorial update (following telephone call from member of the public).	The modification adds reference to the Old Ragged School building and does not change the SA of the site.
IP096 Car Park, Handford Road East	Amend final paragraph of site sheet to read: 'This site is close to Roman, Anglo- Saxon and prehistoric excavated sites. It has high potential for archaeological remains. There is no objection in principle to development but any permission will require a condition relating to archaeological investigation. Measures for archaeology should be addressed at an appropriate stage in the planning process. Early evaluation is advised so that decisions can be taken on preservation in situ and/or appropriate investigation strategies designed. Archaeological costs have the potential to be relatively high. '	For clarity	Suffolk County Council	The modification provides clarity and has been referenced within the assessment of the site against SA Objective ET9. However, the overall result has not changed.
IP098 Transco South of Patterson Road	Site IP098 needs to recognise the mineral handling facilities at the Port which are protected through the Minerals Plan and DM26 and are part	For clarity	Suffolk County Council	The modification provides clarity regarding proximity to minerals uses. It does not change the SA of the site.

Policy / paragraph	Additional Modification	Reason	Objector(s)	Significance to the SA
	of the delivery of a wider marine plan, through the inclusion of the following text: ' <u>The site is close to uses which are</u> <u>safeguarded through the Suffolk</u> Minerals Core Strategy.'			
Site IP105 – Site Sheet	Identify need for cycle lane to run along southern part of site to enhance Gipping cycle path as follows: 'A cycle path should be provided along the south west boundary of the site, linking with the existing riverside cycle paths to the north and south of the site.'	To enhance Gipping cycle path and provide a segregated route as it currently narrows and merges with the footpath as it passes the depot.	Editorial update	The modification has been referenced within the assessment of the site against SA Objectives ET1, ET5 and HW1. This has not resulted in any changes to the assessment scores.
IP133 South of Felaw Street	Site IP133 needs to recognise the mineral handling facilities at the Port which are protected through the Minerals Plan and DM26 and are part of the delivery of a wider marine plan, through the addition of the following text to the site sheet: <u>'The site is close to uses which are safeguarded through the Suffolk</u> <u>Minerals Core Strategy.'</u>	For clarity	Suffolk County Council	The modification provides clarity regarding proximity to minerals uses. It does not change the SA of the site.
IP133 South of Felaw Street	Ensure that on-site open space provision is included in any scheme - highlight the need on the site sheet. Point out the proximity to Grade II listed Felaw Maltings. Add information on archaeology. Add to the end of the constraints section of the site sheet: <u>'On-site open space provision will be</u> required in accordance with policy DM29.	For clarity.	Private individuals Suffolk County Council	The modification provides clarity and has been referenced within the assessment of the site against SA Objective ET9. However, the overall result has not changed.

Policy / paragraph	Additional Modification	Reason	Objector(s)	Significance to the SA
	The site is adjacent to the Wet Dock Conservation Area and to the Grade II listed Felaw Maltings.			
	This site lies close to Anglo-Saxon remains (IPS 683, IPS 230). No objection in principle to development but it will require a condition relating to archaeological investigation attached to any planning consent.'			
IP136 Silo, College Street	Amend final paragraph of site sheet as follows:'This site lies in an area of international archaeological importance, on the Anglo-Saxon and medieval waterfront of Ipswich (Historic Environment Record IPS 413). Ground works could involve surviving sections of 'early' waterfront. There is no objection in principle to development but any permission will require a condition relating to archaeological investigation. Measures for archaeology should be addressed at an appropriate stage in the planning process. Early assessment would be advisable if extensive ground works are proposed so that decisions can be taken on preservation in situ and/or appropriate investigation .'	For clarity	Suffolk County Council	The modification provides clarity and has been referenced within the assessment of the site against SA Objective ET9. However, the overall result has not changed.
IP214 300 Old Foundry Road	Delete The site is also in an area of archaeological importance and add more detailed text on archaeology to site sheet:	For clarity.	Suffolk County Council	The modification provides clarity regarding the site's location within an area of archaeological importance and does not change the SA of the site.

Policy / paragraph	Additional Modification	Reason	Objector(s)	Significance to the SA
	<u>'This site lies within the historic core</u> of Ipswich and the area of archaeological importance, close to the town defences. Any consent for development will require a planning condition to secure a programme of archaeological work.'			
IP221 The Flying Horse PH, Waterford Road	Amend from short term to long term in Table 1 column 6/SP2.	To reflect discussion with IBC Land and Property	Editorial update	The modification is of an editorial nature and does not change the SA of the site.
	CTED FOR GYPSY AND TRAVELLER S			
SP4	Delete policy SP4 and combine necessary elements into policy CS11 of the Core Strategy.	To avoid confusion (arising from the existence of two policies).	National Federation of Gypsy Liaison Groups	Policy SP4 has been deleted from the SA. Policy CS11 has been reviewed and incorporates the assessment undertaken for policy SP4. The overall result has not changed.
SP5 LAND ALLOCA	TED FOR EMPLOYMENT USE / MIXED			
SP5 / para 4.29	Cross refer to the site sheets in the reasoned justification to SP5 to ensure the constraints are addressed: ' supporting their delivery. <u>The</u> <u>constraints are identified on the site</u> <u>sheets included at Appendix 3A and</u> <u>need to be taken into account by</u> <u>prospective developers.</u> '	For clarity and consistency with sites allocated through SP2.	Private individuals	The modification provides clarity and consistency with sites allocated through SP2 and does not change the SA of the policy.
IP035 Key Street / Star Lane / Burtons Site (SP5)	Amend archaeological section of site sheet to read: <u>'In terms of archaeology</u> , This site, <u>the former Cardinal Works</u> , is one of <u>the most important remaining sites in</u> <u>Ipswich with the potential to address</u> <u>major research questions about the</u> <u>origin and development of the town.</u> <u>The site</u> lies on the remains of St	For clarity	Historic England and Suffolk County Council	The modification provides clarity regarding the archaeological sensitivity of the site and has been referenced within the assessment of the site against SA Objective ET9. However, the overall result has not changed.

Policy /	Additional Modification	Reason	Objector(s)	Significance to the SA
paragraph				
	Peter's Priory, within the historic core			
	of Ipswich and close to the Anglo-			
	Saxon and medieval waterfront.			
	Evidence for the Priory and preceding			
	Anglo-Saxon occupation was			
	revealed during evaluation (IPS 455).			
	There are possible is high potential			
	for archaeological remains of possible			
	national significance, such as			
	important waterlogged remains and			
	evidence of the later Wolsey's			
	College and a Quaker burial ground.			
	There is potential for waterlogged			
	remains.			
	This site is one of the most important			
	remaining sites in Ipswich with the			
	potential to address major research			
	questions about the origin and			
	development of the town.			
	The archaeological works required			
	will comprise the total archaeological			
	excavation of the site, prior to			
	development. Detailed early pre-			
	application discussions with Suffolk			
	County Council Archaeological			
	Service and Historic England would			
	be required in order to agree the			
	scope of required assessment, the			
	principle of development and to			
	inform designs (e.g. to allow for			
	preservation in-situ of deposits or			
	appropriate programmes of work).			
	Total archaeological excavation of			
	any development footprint prior to			
	development will be required.			
	Mitigation on this site is likely to be			
	extensive and expensive.			
	1			
	Development should show			

Policy / paragraph	Additional Modification	Reason	Objector(s)	Significance to the SA
Paragraph	consideration for the adjacent Grade			
	1 medieval listed churches of St			
	Peter's and St Mary Quay. There is			
	no objection in principle to			
	development but any permission will			
	require a condition relating to			
	archaeological investigation.'			
	archaeological investigation.			
	<u>'This very sensitive site forms part of</u>			
	the transition area from the town			
	centre to the waterfront where			
	development needs to reflect this			
	transition in terms of design and scale			
	etc. the site contains a Grade II listed			
	building, 1-5 College Street, adjoins			
	the Grade I listed and scheduled Wolsey's Gateway, is located			
	between two conservation areas			
	(Central and Wet Dock) and two			
	Grade II* churches (St Peter's and St			
	Mary at the Quay). Any proposals			
	would need to consider the impact of			
	development upon designated and			
	non-designated heritage assets and			
	their setting, including any resulting			
	benefit, harm or loss to their			
	significance.			
	Proposals impacting on the			
	scheduled monument of Wolsey's			
	gate or its setting would require			
	detailed pre-application discussions			
	with Historic England.			
	The site has access constraints, is			
	within an Air Quality Management			
	Area and a flood zone.'			

Policy /	Additional Modification	Reason	Objector(s)	Significance to the SA
paragraph Site IP067 – Site Sheet	Correction to the site's western boundary, including removal of area of developed land in the south west corner. Consequential change to site area. Add wording to site sheet: It is adjacent to a County Wildlife Site, <u>contains part of a Local Wildlife Site</u> and forms part of the ecological network.	Boundary corrected to the line of ownership. Wording added to reflect Local Wildlife Site status of part of site.	EDF Energy	The site containing part of a LWS has been referenced within the assessment of the site against SA Objective ET8. However, the overall result has not changed.
Site IP152 Airport Farm Kennels	Add to the allocation and site sheet a note that significant new landscaping would be required to maintain the view from the A14 across the site.	For accuracy and to protect the adjacent AONB (the site lies part within the AONB)	Editorial update	The assessment of the site against SA Objective ET10 has been amended to reflect the requirement for landscaping at the site. This does not change the overall result.
Various sites IP067 IP094 IP099 IP140 IP147 IP150c	Add Suffolk County Council information on surface water flooding to site sheets: <u>'Surface water flooding local to site -</u> <u>will need to be considered at planning application stage. See Appendix 1 of the Ipswich SFRA.</u> '	For completeness.	Suffolk County Council	The modification provides completeness regarding the need to consider surface water flooding local to the sites and has been referenced within the assessment against SA Objective ET7. This does not change the overall scores.
Various sites IP058 IP067 IP094 IP099 IP140 IP146 IP147 IP150c IP152 IP094 IP140	Add Suffolk County Council information to site sheets about potential transport requirements: ' <u>A transport assessment and travel</u> <u>plan will be required.</u> ' ' <u>A transport assessment and travel</u> <u>plan will be required. Vehicular</u> <u>access to West End Road may not be</u> <u>acceptable.</u> '	For completeness	Suffolk County Council	The modification provides completeness regarding the need for a transport and travel assessment and does not change the overall assessment.
	Also amend site sheet to read:			

Policy /	Additional Modification	Reason	Objector(s)	Significance to the SA
paragraph				
	'Access constraints and need to			
	consider impacts on rights of way,			
<u> </u>	noise'			
	ATED AND PROTECTED AS OPEN SPA		1	
Site IP083 – Site Sheet	Identify need for cycle lane to run along southern part of site to enhance Gipping cycle path as follows: ' <u>A cycle path should be provided</u> <u>through the southern part of the site</u> <u>as part of a continuous path along the</u> <u>River Gipping</u> .'	To provide a continuous cycle path along the River Gipping	Editorial update	The modification has been referenced within the assessment of the site against SA Objective HW1. However, the overall result has not changed.
SP7 LAND ALLOC/ Various sites	ATED FOR LEISURE USES OR COMMU Add Suffolk County Council	INITY FACILITIES	Suffolk County	The modification is for completeness
IP150b IP258 IP260	information on surface water flooding to site sheets:	T of completeness	Council	within the site sheets stating the need to consider surface water flooding local to the sites and has been referenced within
	Surface water flooding local to site - will need to be considered at planning application stage. See Appendix 1 of the Ipswich SFRA.			the assessment against SA Objective ET7. The overall SA result has not changed.
	Also add to the site sheet for IP260: <u>'There is a Flood Incident Report for</u> <u>this site.</u> '			
Various sites IP150b IP258	Add Suffolk County Council information to site sheets about	For completeness	Suffolk County Council	The modification states the need for a transport assessment and travel plan
IP260	potential transport requirements:			and has been referenced within the assessment against SA Objective ET4.
	A transport assessment and travel plan will be required.			This does not change the overall SA result.
	INTRY PARK EXTENSION			
SP8 ORWELL	Add text:	For clarity.	Natural England	The effects of SP8 are assessed under
COUNTRY PARK	A visitor survey of Orwell Country			allocation IP149. The modification
EXTENSION	Park undertaken during March 2015			provides clarity to the supporting text

Policy / paragraph	Additional Modification	Reason	Objector(s)	Significance to the SA
Paragraph 4.44	concluded that the extension of the park into land at Pond Hall Carr and Farm is not likely to have harmful impacts on the Special Protection Area.			regarding an extension of the country park and does not change the SA of the policy. The addendum to the Habitats Regulations Assessment confirms that the amendment would not result in effects on the Stour and Orwell Estuaries SPA, concluding 'the text adds clarity to the existing safeguard'.
SP8 ORWELL COUNTRY PARK EXTENSION Paragraph 4.45	Add text: <u>Project level Habitats Regulations</u> <u>Assessment would be needed for any</u> <u>visitor centre proposal.</u>	For clarity.	Natural England	The modification provides clarity and does not change the SA of the Policy. The addendum to the Habitats Regulations Assessment confirms that the amendment would not result in effects on the Stour and Orwell Estuaries SPA, concluding 'the text adds clarity to the existing safeguard'.
CHAPTER 5	·		·	
Paragraph 5.8	Add to end of paragraph: <u>'Grade I and II* buildings in Ipswich</u> <u>are dealt with through the national</u> <u>Heritage at Risk register.</u>	For completeness.	Historic England.	The modification notes that Grade I and II* buildings in Ipswich are dealt with through the Heritage at Risk register and has not resulted in any overall changes to the assessment.
SP10 RETAIL SITE	ALLOCATION			
SP10 / IP040 Civic Centre Area, Civic Drive (also known as Westgate)	Add noise constraint to site sheet (in relation to residential element of the mix): 'Residential use could be located above or adjacent to retail uses. The residential element of the development should be designed and located so as not to be adversely affected by noise from the Wolsey Theatre.'	To ensure that the residential element of development is designed and located so as not to be adversely affected by noise from the Wolsey Theatre.	Theatres Trust	The addition of the noise constraint does not change the SA of the policy.
	PEDESTRIAN AND CYCLE ROUTES	1	1	
SP15 Improving Pedestrian and Cycle Routes / paragraph 5.41	Minor amendment to explanation of Travel Ipswich: 'Due for completion in 2015, it	For clarity.	Suffolk County Council	The modification provides clarity regarding Travel Ipswich and does not change the SA of the policy.

Policy /	Additional Modification	Reason	Objector(s)	Significance to the SA
paragraph				
	aimsforms part of a wider long term			
	strategy to achieve a 15% switch to			
	more sustainable modes, to enable			
	Ipswich to accommodate planned			
	growth without corresponding growth			
	in congestion. This will seeinclude			
	some further improvements to			
	walking routes from the railway			
00/00000	station'			
SP16 TRANSPOR				
SP16 Transport	Add constraints information to	For clarity.	ABP	The modification provides clarity to the
proposals in IP-	supporting text:			supporting text regarding avoiding
ONE	' <u>The Wet Dock Crossing must avoid</u>			unacceptable impact on vessel access
Paragraph 5.46	unacceptable impact on vessel			to the Wet Dock and does not change
	access to the Wet Dock, on Cliff			the SA of the policy.
	Road, on the West Bank Terminal			
	railhead, on port security and on New			
0.5.4.6	Cut navigation.'			
SP16	Update supporting text to reflect the	For accuracy	Ben Gummer MP	The modification provides accuracy of
Paragraph 5.46	fact that feasibility work is to be		(see objection to	the supporting text to reflect feasibility
	carried out on the route.		CS20)	work to be carried out and does not
	L TRE CAR PARKING			change the SA of the policy.
SP17 TOWN CEN SP17/Site IP055	Add text to site sheet on	For elevity	Suffalls Country	The medification provides clarity
		For clarity	Suffolk County Council	The modification provides clarity
Crown Car Park, Charles St	archaeological constraints: 'The site is within an area of Anglo-		Council	regarding archaeological constraints and
Chanes St	Saxon and medieval suburban			does not change the SA of the site or
	activity, beyond the early town			policy.
	defences. Geotechnical survey data			
	will inform understanding of the			
	impacts of past landscaping, but this will be heavy across most of the site.			
	In the event that archaeological			
	remains may have survived previous			
	landscaping in the southern part of it,			
	consent for development may require			
	a planning condition to secure a			
	programme of targeted			
	archaeological work.'			
	archaeological work.			

Policy / paragraph	Additional Modification	Reason	Objector(s)	Significance to the SA
SP17 Town centre car parking	Add note to site sheet for IP055 Crown Street: <u>'The redevelopment should deliver</u> <u>accessibility, legibility and</u> <u>attractiveness of the public realm.</u> '	For completeness.	Ipswich Central	The modification relates to the need to achieve accessibility, legibility and attractiveness of the public realm in the Crown Street redevelopment and is intended for completeness. It does not change the SA of the policy.
Part C, IP-One Opp	portunity Areas, CHAPTER 6			
Opportunity Areas para 6.1	A sentence should be added to the start of Part C stating that the allocation policies take precedence if there are inconsistencies. <u>'The allocation policies of the Plan</u> <u>take precedence in the event of</u> <u>inconsistencies between the</u> <u>Opportunity Area guidance and site</u> allocations.'	For clarity and to avoid confusion	Applekirk Properties Ltd	The modification provides clarity regarding the allocation policies of the plan taking precedence in the event of inconsistencies between the Opportunity Area guidance and site allocations. This does not change the SA.
Opportunity Area A Island	Clarify that the mix percentage is indicative.	For consistency with paragraph 4.8	ABP	The modification provides clarity that the mix percentage is indicative and does not change the SA.
Opportunity Area B and C	Amend development principles for areas B and C to read 'Development to take account of address heritage issues' Also amend the area analysis maps to include the scheduled monuments.	For clarity	Historic England	The modification provides clarity of the text and includes scheduled monuments on the analysis maps. This does not change the SA.
Opportunity Area B Merchant Quarter	Development Opportunities Delete reference to percentages of use: Residential (max 50%) Non-residential use (50% +)	For consistency with the site allocations in the area for which the proportions of different uses vary, e.g. IP043 and IP052 80% of the site residential. Also the percentages of mix within sites as specified through policy SP2 is indicative (see paragraph	Applekirk Properties Ltd	The removal of text does not change the SA.

Policy /	Additional Modification	Reason	Objector(s)	Significance to the SA
paragraph				
		4.8) in order to take account of viability.		
	EMENTATION, TARGETS, MONITORING			
7.3	Add reference to Ipswich Central and	For completeness.	Ipswich Central	The modification is intended for
7.0	Greater Ipswich Partnership.	Tor completeness.	ipswich Gentral	completeness and does not change the SA.
APPENDIX 2A List	of Policies		•	· ·
Policy SP4	Delete policy reference as the policy is deleted and combined with CS11.	For consistency	Editorial	The Policy has been deleted from the SA.
Policy SP9	Amend policy title to Safeguarding land on development sites for transport infrastructure	For consistency	Editorial	The policy title amendment does not change the SA of the policy.
Policy SP17	Amend policy title to Town Centre Car Parking	For consistency	Editorial	The policy title amendment does not change the SA of the policy.
All policy references	Policy SP4 is deleted, therefore all remaining policy reference numbers will need to be updated	For consistency	Editorial	The Policy has been deleted from the SA and the policy references will be updated.
APPENDIX 3A Site Allocation Deta	ails			
Appendix 3A Site Allocation Details	Please refer to modifications to the site sheets listed under policy SP2 above which allocates the sites.			The main modifications within SP2 remove sites from Policy SP3 and place them into Policy SP2 within the SA. The number of dwellings per site allocation has not changed therefore the significant findings and the overall assessment of the sites within the SA has not changed.
APPENDIX 3B Development Cons	traints for SP3 sites			
IP011a Smart Street/Foundation Street	Amend constraints data: 'This site affects an area of archaeological importance, within the area of the Anglo-Saxon and medieval town. It lies adjacent to Scheduled Monuments relating to parts of the early town. No objection in principle to development but it will require a condition relating to archaeological investigation attached	For clarity.	Suffolk County Council	The modification provides clarity regarding the archaeological importance of the area in which the site is located as well as archaeological measures that should be addressed and this has been referenced within the assessment of the site against SA Objective ET9. However, the overall result has not changed.

Policy /	Additional Modification	Reason	Objector(s)	Significance to the SA
paragraph				
	to any planning consent. Measures			
	for archaeology should be addressed			
	at an appropriate stage in the			
	planning process. Archaeological			
	costs have the potential to be			
	relatively high, and early evaluation is			
	advised so that decisions can be			
	taken on preservation and/or			
	appropriate investigation strategies			
	designed.'			
IP052 Land	Amend constraints data:	For clarity.	Suffolk County	The modification provides clarity
between Lower	'This site is within the urban core (IPS		Council	regarding archaeological constraints and
Orwell Street and	413), close to Scheduled areas of			has been referenced within the
Star Lane	Middle Saxon and medieval			assessment of the site against SA
	occupation (SF 189-191<u>Nos.</u>			Objective ET9. However, the overall
	<u>1005985, 1002973, 1005986</u>). It is			result has not changed.
	within the street pattern area close to			
	the waterfront and is likely to contain			
	complex and important archaeological			
	remains that will involve potentially			
	high excavation costs. No objection in			
	principle to development but it will			
	require a condition relating to			
	archaeological investigation attached			
	to any planning consent. Measures			
	for archaeology should be addressed			
	at an appropriate stage in the			
	planning process. Early evaluation is			
	advised so that decisions can be			
	taken on preservation in situ, and/or			
	appropriate investigation strategies			
	designed.'			
IP132 Former St	Amend as follows:	For clarity.	Suffolk County	The modification provides clarity
Peter's	'This site lies in an area of	i of Ganty.	Council	regarding the archaeological importance
Warehouse Site	international archaeological			of the area in which the site is located as
VVII ENUUSE OILE	importance, on the Anglo-Saxon and			well as archaeological measures that
	medieval waterfront of Ipswich			should be addressed and this has been
	(Historic Environment Record IPS			referenced within the assessment of the

Policy /	Additional Modification	Reason	Objector(s)	Significance to the SA
paragraph				
	413). This site potentially represents			site against SA Objective ET9. However,
	that last surviving section of 'early'			the overall result has not changed.
	waterfront, and may also encounter			
	the remains of bridges dating from at			
	least the 10th century. The site will			
	involve potentially high excavation			
	costs. No objection in principle to			
	development but it will require a			
	condition relating to archaeological			
	investigation attached to any planning			
	consent. Archaeological costs have			
	the potential to be relatively high.			
	Measures for archaeology should be			
	addressed at an appropriate stage in			
	the planning process. Early			
	consultation and evaluation is advised			
	so that decisions can be taken on			
	preservation in situ, and/or			
	appropriate investigation strategies			
	designed.'			
IP178 Island	<u>'This site, in the Ipswich waterfront</u>	For clarity.	Suffolk County	The modification provides clarity
House, Duke	area, will require a condition relating		Council	regarding archaeological investigation
Street	to archaeological investigation			and does not change the SA of the site.
	attached to any planning consent.'			
IP205 Burton's,	Amend as follows:	For clarity.	Suffolk County	The modification provides clarity
College Street	'This site lies in an area of		Council	regarding the archaeological importance
	international archaeological			of the area in which the site is located as
	importance, on the Anglo-Saxon and			well as archaeological measures that
	medieval waterfront of Ipswich			should be addressed and this has been
	(Historic Environment Record IPS			referenced within the assessment of the
	413). This site potentially represents			site against SA Objective ET9. However,
	the last surviving section of 'early'			the overall result has not changed.
	waterfront. The site will involve			
	potentially high excavation costs. No			
	objection in principle to development			
	but it will require a condition relating			
	to archaeological investigation			
l	attached to any planning consent.			
	Measures for archaeology should be			

Policy /	Additional Modification	Reason	Objector(s)	Significance to the SA
paragraph	addressed at an appropriate stage in the planning process. Early consultation and evaluation is advised so that decisions can be taken on preservation <i>in situ</i> , and/or appropriate investigation strategies designed.'			
IP264 28-32 Tacket Street	Add: <u>'This site is in the Anglo-Saxon and</u> <u>Medieval core of Ipswich, close to</u> <u>excavated remains (IPS 210). No</u> <u>objection in principle to re-</u> <u>development but it will require a</u> <u>condition relating to archaeological</u> <u>investigation attached to any planning</u> <u>consent.'</u>	For clarity.	Suffolk County Council	The modification provides clarity regarding archaeological investigation at the site and does not change the SA of the site.
NEW APPENDIX	Insert new Appendix 4 – Saved policies superseded by this plan The following saved policies are superseded by this Site Allocations Plan: (see table below)	For legal compliance.	Editorial update	This update is of an editorial nature and does not change the SA.
Policies Map	Amend the sites which have been moved from policy SP3 to policy SP2 and vice versa. Add new site added to SP3: IP272 72 Foundation Street Amend boundary to IP089 Waterworks Street. Delete IP263 West of Bridge Street, north of the River Orwell, now part of larger allocation IP047 Land at Commercial Road.	For accuracy	EDF Energy	The updates to the policies map do not change the SA.

Policy /	Additional Modification	Reason	Objector(s)	Significance to the SA	
paragraph					
	Amend western boundary to site IP067 British Energy.				
	Amend the boundary of Opportunity Area F, River and Princes Street Corridor, on the policies map. It should include the eastern end of the extended site IP047 Land at Commercial Rd as shown in Part C, Chapter 6, of the Site Allocations and Policies (incorporating IP-One Area Action Plan) development plan document. The boundary at the eastern end now follows the river, Bridge Street and Grafton Way.				

Consultation Comments on the Proposed Submission SA Report

REP ID	RESPONDENT NAME	CHAPTER	SUPPORT / OBJECT	REPRESENTATION SUMMARY	CHANGE TO PLAN REQUESTED	IBC Response
5509	Northern Fringe Protection Group (Mr Brian Samuel) [976]	4.3 Appraisal of Site Allocations	OBJECT	The viability of the allocation of the Westgate site for Retail has been questioned by Ipswich Central and the alternative options proposed by Ipswich Central for Retail sites need to be considered in the SA of the Site Allocations accordingly.	Alternative options for retail sites proposed by Ipswich Central need to be considered in the SA.	The allocation at Westgate is based upon the most up to date evidence available which is the DTZ 2013 Ipswich Town Centre Opportunity Areas Report.
5739	Save Our Country Spaces (Mrs Barbara Robinson) [978]	4.3 Appraisal of Site Allocations	OBJECT	SOCS endorse the Northern Fringe Protection Group's points. The viability of the allocation of the Westgate site for Retail has been questioned by Ipswich Central and the alternative options proposed by Ipswich Central for Retail sites need to be considered in the SA of the Site Allocations accordingly.	Alternative options for retail sites proposed by Ipswich Central need to be considered in the SA.	The allocation at Westgate is based upon the most up to date evidence available which is the DTZ 2013 Ipswich Town Centre Opportunity Areas Report.

F	REP ID	RESPONDENT NAME	CHAPTER	SUPPORT / OBJECT	REPRESENTATION SUMMARY	CHANGE TO PLAN REQUESTED	IBC Response
Ę	5621	Natural England (Mr John Jackson) [1413]	Chapter 4: APPRAISAL OF THE SITE ALLOCATIONS DPD	OBJECT	Natural England is reasonably satisfied that the SA considers the impacts of the Core Strategy and Policies [sic] on relevant aspects of the environment within our remit, including biodiversity and geology, landscape, green infrastructure and soils. We particularly welcome SA objectives to protect and enhance designated sites, including SSSIs, SACs, SPAs and Ramsar sites, in addition to locally designated and non-designated areas of biodiversity. However, we would advise that the SA should cross- reference with the findings and recommendations of the AA which identifies potential recreational disturbance effects on European sites, and measures to mitigate these.	We would advise that the SA should cross-reference with the findings and recommendations of the AA which identifies potential recreational disturbance effects on European sites, and measures to mitigate these.	Further cross references and explanation of the Habitats Regulations Assessment have been added into the addendum to the SA Report.